STATUS OF THESIS

Title of thesisINTERRELATIONSHIPS BETWEEN USERS AND SYSTEMTitle of thesisFLEXIBILITIES WITH PERCEIVED USABILITY OF ONLINEAIRLINE RESERVATION SYSTEMS

ARIF MUSHTAQ

hereby allow my thesis to be placed at the Information Resource Center (IRC) of Universiti Teknologi PETRONAS (UTP) with the following conditions:

- 1. The thesis becomes the property of UTP
- 2. The IRC of UTP may make copies of the thesis for academic purposes only.
- 3. This thesis is classified as



Ι

Confidential

 $\sqrt{}$

Non-confidential

If this thesis is confidential, please state the reason:

The contents of the thesis will remain confidential for _____ years.

Remarks on disclosure:

Alphinhlard

Signature of Author

Permanent address:

House 36A, Street 39B, Sector I-9/4

Islamabad, Pakistan

Date: 20 - 12 - 2011

Endorsed by

Signature of Supervisor

Name of Supervisor

Dr. Suziah Bt. Sulaiman

Date: 21/12/2011

UNIVERSITI TEKNOLOGI PETRONAS

DISSERTATION TITLE: INTERRELATIONSHIPS BETWEEN USERS AND

SYSTEM FLEXIBILITIES WITH PERCEIVED USABILITY OF ONLINE

AIRLINE RESERVATION SYSTEMS

by

ARIF MUSHTAQ

The undersigned certify that they have read, and recommend to the Postgraduate Studies Programme for acceptance this thesis for the fulfilment of the requirements for the degree stated.

Signature:

Main Supervisor:

Signature:

Co-Supervisor:

Signature:

Head of Department:

Dr. Suziah Bt. Sulaiman

R. P. D. D. DOMINIC Associate Professor Sciences Department Computer & Information Universit Teknologi PETRONAS

Assoc. Prof. Dr. P. Dhanapar Clorad Forman 24750 Tronoh

Dr Mohd Fadzil B Hassen Hosd

New Brit & Information Sciences Department Invarial Investige (1233-0045)

Dr. Mohd Fadzil Bin Hassan

21/12/11

Date:

DISSERTATION TITLE: INTERRELATIONSHIPS BETWEEN USERS AND SYSTEM FLEXIBILITIES WITH PERCEIVED USABILITY OF ONLINE AIRLINE RESERVATION SYSTEMS

by

ARIF MUSHTAQ

A Thesis

Submitted to the Postgraduate Studies Programme in Fulfilment of the Requirement of the Degree of

DOCTOR OF PHILOSOPHY IN INFORMATION TECHNOLOGY UNIVERSITI TEKNOLOGI PETRONAS BANDAR SERI ISKANDAR, PERAK

DECEMBER 2011

DECLARATION OF THESIS

Title of thesis

I

INTERRELATIONSHIPS BETWEEN USERS AND SYSTEM FLEXIBILITIES WITH PERCEIVED USABILITY OF ONLINE AIRLINE RESERVATION SYSTEMS

ARIF MUSHTAQ

hereby declare that the thesis is based on my original work except for quotations and citations which have been duly acknowledged. I also declare that it has not been previously or concurrently submitted for any other degree at UTP or other institutions.

Mushtar

Signature of Author

Permanent address:

House 36A, Street 39B, Sector I-9/4

Islamabad, Pakistan

Date: 20 - 12 - 2011

Witnessed by

Signature of upervisor

Name of Supervisor in rached then

Dr. Suziah Bt. Sulaiman

Date: 21/12/2011

DEDICATION

This dissertation is dedication to my father, who unconditionally loved me since I was born. I wish you could stay few more years to see all my success and amazing accomplishments. I wish you could see that I am going to be a PhD. I wish you could see that I have become a father of a loving daughter. You are truly missed. I love you!

ACKNOWLEDGEMENTS

It is a pleasure to thank those who made this thesis possible and supported me throughout the course of my studies. Foremost, my eternal gratitude goes to my loving wife, for her everlasting support and unconditional love. It would have been next to impossible to write this thesis without her help and guidance. I am forever grateful to my mom for her moral support and prayers. I would like to thank my family for all their love and encouragement. I owe my deepest gratitude to my mother in law and father in law who always remembered me in their prayers. I am thankful to my beloved daughter "Manahil" who made me smile during the hectic research days.

To my supervisor, Dr. Suziah Bt. Sulaiman, I am highly grateful for her help and guidance in the research. I am forever obliged for the encouragement and support that you never hesitated to give me. Similarly, thanks to my co-supervisor Dr. P. D. D. Dominic for his support and counseling. This work would not have been possible without their guidance.

I owe endless thankfulness to all those such as students, staff, travel agents, travelers and other anonymous evaluators who participated in the survey. Special appreciation goes to Gehan for helping me in making the interface prototype. The final study would have not been possible without them. Lastly, I am grateful to my friend for providing me numerous distractions and relaxation opportunities during the journey of this dissertation.

ABSTRACT

It is very critical for the organizations to design flexible systems that are easy to use and can accomplish all the requirements by way of offering customizability. Philosophers argue that users are good in adapting the systems; however, research shows users dissatisfaction with existing Online Airline Reservation Systems in terms of task completion. Therefore, researchers are eager to find out ways for improving online usability of the systems, how users' Perceived Usability of the system is formulated by its flexibility functions. This research therefore examines travelers' expectations, preferences and online behavior (Users' Flexibility) and aligns that with designing of flexible online airline reservation systems (System's Flexibility) and users' as evaluators of the online systems to determine its Perceived Usability through users' effectiveness, efficiency and satisfaction (Perceived Usability).

In this dissertation, both quantitative and qualitative techniques were used to analyze the data collected in the context of SF, UF and PU of the systems. A redesign solution for enhanced usability was developed based on HCI guidelines and the flexibility tactics used in online travel agencies, which led to a proposed interface with the integration of opaque mechanism. The two interfaces were used in the experiment. Participants were requested to complete the evaluation of the existing and proposed interfaces.

The findings suggested that users can be classified on the basis of their Flexible Traveling Behavior which led to the development of a Users' Flexibility measuring scale. It is further investigated that integration of opaque fares concept would increase the usability of the system. Since flexibility is referred to its ability to respond to internal or external changes, systems incorporated with opaque fares would serve the role of external change agent by way of providing flexibility in users' decision making and will also serve the role of internal change agent by way of providing the capability of accepting changed decisions.

ABSTRAK

Ini adalah sangat penting bagi organisasi untuk mereka bentuk sistem yang fleksibel dan mudah untuk digunakan serta boleh mencapai semua keperluan dengan cara menawarkan kebolehan untuk mengubahsuai. Ahli-ahli falsafah berpendapat bahawa pengguna berkebolehan untuk menyesuaikan diri menggunakan sesuatu sistem, namun kajian menunjukkan rasa tidak puas hati pengguna dengan sistem yang sedia ada dalam tempahan penerbangan secara talian dari segi menyelesaikan tugas. Oleh sebab itu, para penyelidik amat berminat untuk mengetahui cara-cara untuk meningkatkan kebolehgunaan system dalam talian, dan bagaimana persepsi pengguna terhadap menganggap kebolehgunaan sistem dapat digubal menerusi fungsi fleksibilitinya. Kajian ini meneliti jangkaan pelancong, keutamaan dan tingkah laku mereka dalam talian (fleksibiliti pengguna) dan menjajarkannya dengan reka bentuk system tempahan penerbangan dalam talian (sistem fleksibiliti) dan meletakkan pengguna sebagai penilai sistem tersebut bagi menentukan kebolehgunaan melalui kepuasan pengguna (persepsi kebolehgunaan).

Dalam disertasi ini, kedua-dua teknik kuantitatif dan kualitatif telah digunakan untuk menganalisa data yang dikumpulkan dalam konteks fleksibiliti, kelonggaran sistem pengguna dan kebolehgunaan sistem. Satu penyelesaian bagi mereka bentuk semula untuk memberi kegunaan yang lebih tinggi telah dibangunkan berdasarkan garis panduan *HCI* dan taktik yang fleksibel yang digunakan dalam agensi-agensi pelancongan dalam talian. Satu rekabentuk sistem tempahan penerbangan dalam talian yang baru telah diaplikasikan dan membawa kepada antara muka yang dicadangkan dengan integrasi mekanisme legap. Kedua-dua antara muka telah digunakan dalam eksperimen tersebut. Para peserta telah diminta untuk melengkapkan penilaian antara muka yang sedia ada dan yang dicadangkan.

Hasil penemuan mencadangkan supaya pengguna boleh dikelaskan berdasarkan tingkah laku perjalanan mereka yang fleksibel yang lantaran itu membawa kepada

pembangunan skala mengukur fleksibiliti seorang pengguna. Perkara ini disiasat dengan lebih mendalam yang mana integrasi konsep tambang legap akan meningkatkan kebolehgunaan system. Oleh sebab fleksibiliti dirujuk dengan keupayaan untuk bertindak balas terhadap perubahan dalaman atau luaran, sistem yang diperbadankan dengan tambang legap akan ber peranan sebagai agen perubahan luaran dengan menyediakan fleksibiliti supaya pengguna dapat membuat keputusan dan juga akan berperanan sebagai agen perubahan dalaman melalui penyediaan keupayaan menerima keputusan berubah.

In compliance with the terms of the Copyright Act 1987 and the IP Policy of the university, the copyright of this thesis has been reassigned by the author to the legal entity of the university,

Institute of Technology PETRONAS Sdn Bhd.

Due acknowledgement shall always be made of the use of any material contained in, or derived from, this thesis.

© Arif Mushtaq, 2011 Institute of Technology PETRONAS Sdn Bhd All rights reserved.

TABLE OF CONTENTS

STATUS OF THESIS	i
APROVAL PAGE	ii
TITLE PAGE	iii
DECLARATION	iv
DEDICATION	v
ACKNOWLEDGEMENTS	vi
ABSTRACT	vii
ABSTRAK	viii
COPYRIGHT PAGE	х
TABLE OF CONTENTS	xi
LIST OF TABLESx	cvii
LIST OF FIGURES	xix
LIST OF ABBREVIATIONS	xxi

Chapter

1.	INT	RODU	CTION	1
	1.1	Resea	rch Background	1
	1.2	Proble	em Statement	4
	1.3	Resea	rch Aims and Objectives	6
	1.4	Resea	rch Methodology	7
	1.5	Scope	of Research	9
	1.6	Resea	rch Contribution	10
	1.7	Organ	nization of the Thesis	12
2.	LITI	ERATU	JRE REVIEW	13
	2.0	Chapt	er Overview	13
	2.1	Flexit	bility and Online Airline Reservation Systems	13
		2.1.1	Flexibility Concepts	13
			2.1.1.1 System's Flexibility	14
			2.1.1.2 Users' Flexibility	16
			2.1.1.3 Conceptual Linking between SF and UF	18
		2.1.2	Airline Reservation Systems	20
			2.1.2.1 Computer Reservation Systems	20
			2.1.2.2 Self-Booking Tools vs. Online Travel Agencies	23
	2.2	Usabil	ity of Online Airline Reservation Systems	31
		2.2.1	Usability Perception	32
			2.2.1.1 Usability Perception by Performing Content Analysis	-33

			2.2.1.2 Usability Perception through User's Internet Adoptic	on 33
			2.2.1.5 Usability refeeption based on Users Preferences a	10 24
			2.2.1.4 Usability Perception based on Online Behaviour	of
			Web Users	34
		2.2.2	Conceptual Linking between PU and FTB and FOARS	35
	2.3	Usabili	ty vs. Flexibility	36
		2.3.1	Opaque Selling Mechanics in System's Flexibility	37
		2.3.2	Opaque Selling Mechanics in User's Flexibility	39
	2.4	Chapte	r Summary	40
3.	MET	HODO	LOGY	43
	3.0	Chapte	r Overview	43
	3.1	Researc	ch Methodology	43
		3.1.1	Phase I: Assessing Users' Needs	48
		3.1.2	Phase II: Classification of Users	48
		3.1.3	Phase III: Case Study	49
	3.2	Phase I	I: Systems' Flexibility & Users' Flexibility	49
		3.2.1	Study 1: Issues with Flexibility	49
			3.2.1.1 Rationale	50
			3.2.1.2 Methodology	51
			3.2.1.3 Validity	52
			3.2.1.4 Sample Size	52
			3.2.1.5 Response Rate	53
			3.2.1.6 Scale	53
			3.2.1.7 Analysis	53
		3.2.2	Study 2: Users' Flexible Behavior in Terms of Compromisin	ng
			on SQA	54
			3.2.2.1 Rationale	54
			3.2.2.2 Methodology	54
			3.2.2.3 Validity	55
			3.2.2.4 Sample Size	56
			3.2.2.5 Response Rate	57
			3.2.2.6 Scale	57
			3.2.2.7 Analysis	57
		3.2.3	Study 3: Integration of OTA Features can make SBTs mo	re
			FOARSs	58
			3.2.3.1 Rationale	58
			3.2.3.2 Methodology	58
			3.2.3.3 Analysis	59
	3.3	Phase	II: Users' Classification and Interrelationship Testing	of
		Variabl	le	60
		3.3.1	Study 4: Users' Perception on Factors Influencing FTB	60
			3.3.1.1 Rationale	60
			3.3.1.2 Methodology	60
			3.3.1.3 Validity	62
			3.3.1.4 Sample Size	62
			3.3.1.5 Response Rate	64

		3.3.1.6 Scale
		3.3.1.7 Analysis
	3.3.2	Study 5: Study to Classify Users' on the Basis of FTB
		3.3.2.1 Rationale
		3.3.2.2 Methodology
		3.3.2.3 Validity
		3.3.2.4 Sample Size
		3.3.2.5 Response Rate
		3.3.2.6 Scale
		3.3.2.7 Analysis
	3.3.3	Research Way Forward
3.4	Phase	III: Case Study to Test the Proposed Framework
5.1	341	Rationale
	342	Methodology
	343	Validity
	344	Sample Size
	345	Response Rate
	346	Paper Prototype
	347	Usability Evaluation of Prototype
	348	Scale
	3/10	Analysis
35	Statict	Analysis
5.5	2 5 1	Descriptive Analysis
	252	Beargon Coefficient Correlation
	3.5.2	Pelability Analysis
	3.3.3	ANOVA
	3.3.4	
	3.3.3	ANCOVA
	3.5.6	
	3.5.7	Multiple Regressions
	3.5.8	Post Hoc Scheffe's Test
2.0	3.5.9	Kendall's Tau
3.6	Chapte	er Summary
KES		AND ANALYSIS
4.0	Chapte	
4. I	Phase	I: User Needs Associated with System's Flexibility
	4.1.1	Descriptive and Kellability Analysis
	4.1.2	Hypotheses 1 esting H ₁
	4.1.3	Hypotheses I esting H_2
	4.1.4	Hypotheses I esting H ₃
	4.1.5	Hypotheses I esting H_4
	4.1.6	Hypotheses I esting H_5
4.2	Phase	I: Users' Flexibility in Terms of Compromising on SQAs'?
	4.2.1	Assumptions in ANOVA to 1 est H_6
		4.2.1.1 Box-and-Whisker Plot H_6
		4.2.1.2 Means Plot H ₆
		4.2.1.3 T-Test H ₆
		4.2.1.4 Error Bars H ₆

	4.2.2	Hypothesis Testing H ₆	-93
		4.2.2.1 One-way Analysis of Variance H ₆	93
		4.2.2.2 Post-hoc Scheffe Tests H ₆	94
		4.2.2.3 Effect Size for One-way ANOVA H ₆	94
	4.2.3	Assumptions in ANOVA to Test H ₇	95
		4.2.3.1 Box-and-Whisker Plot H ₇	95
		4.2.3.2 Means Plot H ₇	96
		4.2.3.3 T-Test H ₇	97
		4.2.3.4 Error Bars H ₇	97
	4.2.4	Hypothesis Testing H7	98
		4.2.4.1 One-way Analysis of Variance H ₇	99
		4.2.4.2 Post-hoc Scheffe Tests H ₇	99
		4.2.4.3 Effect Size for One-way ANOVA H ₇	100
4.3	Phase	I: Users' Satisfaction with SBTs against their rated OTA	4
	Feature	e	100
	4.3.1	Hypotheses Testing H ₈	101
		4.3.1.1 Means Plot H ₈	101
		4.3.1.2 Two-way Analysis of Variance H ₈	102
4.4	Phase	II: Users' Perception on Flexible Traveling Behavior	104
	4.4.1	Qualitative Analysis	104
4.5	Phase	II: Classification of Users on the Basis of Their FTB	109
	4.5.1	Hypotheses Testing H ₉	109
	4.5.2	Hypotheses Testing H ₁₀	111
	4.5.3	Hypotheses Testing H ₁₁	111
	4.5.4	Hypotheses Testing H ₁₂	114
		4.5.4.1 One-way Analysis of Variance H ₁₂	114
		4.5.4.2 Analysis of Covariance H ₁₂	115
4.6	Phase	III: Users' Flexibility is determined by SQAs and EVs	116
	4.6.1	Hypotheses Testing H ₁₃	116
		4.6.1.1 Scatter Plots H_{13}	116
		4.6.1.2 Pearson Correlation Coefficients H ₁₃	117
		4.6.1.3 Multiple Regression Analysis H ₁₃	118
		4.6.1.4 Analysis of Variance H_{13}	118
		4.6.1.5 Testing and Interpreting Model Coefficients H_{13}	119
4./	Phase	III: Perceived Usability with Existing and Proposed Systems	120
	4./.1	Hypotneses Testing H_{14}	120
		4.7.1.1 Descriptive Statistics on Effectiveness H_{14}	120
		4.7.1.2 Two-way Analysis of Variance on Effectiveness H_{14}	121
		4.7.1.5 Effort Bars on Effectiveness H ₁₄	121
		4.7.1.5 Descriptive Statistics on Efficiency H	122
		4.7.1.5 Descriptive Statistics on Efficiency H ₁₄	123
		4.7.1.0 Two-way Analysis of Valiance on Efficiency H	124
		4.7.1.8 Interaction Effect on Efficiency U.	124
		4.7.1.6 Interaction Effect on Efficiency Π_1	120
		4.7.1.1 Descriptive Statistics of Satisfaction II.	120
		4.7.1.10 Two-way Analysis of Variance on Satisfaction H.	127
		4 7 1 12 Interaction Effect on Satisfaction H.	127
			147

		4.7.1.13 Levene's Test on Existing Systems H ₁₄	. 129
		4.7.1.14 Post-hoc Comparisons on Existing Systems H ₁₄	. 130
		4.7.1.15 Levene's Test on Proposed Systems H ₁₄	. 130
		4.7.1.16 Post-hoc Comparisons on Proposed Systems H ₁₄	. 130
	4.8	Phase III: Effect of Users' Flexibility on Proposed Systems?	. 131
		4.8.1 Hypotheses Testing H ₁₅	131
		4 8 1 1 Scatter Plots H ₁₅	131
		4.8.1.2 Homogeneity of Covariance's H ₁₅	133
		4 8 1 3 Alpha Adjustment H ₁₅	135
		4 8 1 4 Univariate ANOVAS H.	135
		4.8.2 Hypotheses Testing Hy	137
		4.8.2 Descriptive Statistics on Perceived Usability H.	137
		4.8.2.2.1 Descriptive Statistics on referred Usability H ₆	139
		4.8.2.3 Alpha Corrections for Post box Comparisons H	130
		4.8.2.4 Post-boc Multiple Comparisons H.	139
	49	Chapter Summary	1/3
	т.2	Chapter Summary	. 1 4 5
5	DISC	USSION	144
5.	5.0	Chapter Overview	144
	5.0	System's Flexibility	147
	5.2	Users' Elevibility in terms of Compromising on SOAs of OARSs	148
	53	Integration Assessment of OTAs Features into SBTs	150
	5.5	Users' Perception on Factors Influencing Upon FTB	150
	5.5	Classification of Users on the Basis of Their FTB	159
	5.6	Role of SOAs and External Variables in Elevible Behavior of Travel	er 160
	5.7	Users' Perceived Usability with Existing and Proposed Systems	162
	5.8	Multivariate Main Effect of Users' Elexible Traveling Behavior	0n
	5.0	Proposed System's Effectiveness Efficiency and Satisfaction	164
	59	Recommendations for FOARS	164
	5.7	5.9.1 Graphical User Interface for FOARS	167
		5.9.1.1 Home Page of FOARS	167
		5.9.1.2 Booking Interface of FOARS	168
		5.9.1.3 Flight Search Window for Flexible Travelers	168
		5.9.1.4 Travelers' Details Window for Flexible Travelers	160
		5.9.1.5 Payment Window for Elevible Travelers	160
		5.9.1.6 Immeture Ticket Window for Elevible Travelers	170
		5.9.1.7 Flight Search Window for Inflexible Travelers	. 170
		5.9.1.8 Travelers' Details Window for Inflexible Travelers	171
		5.9.1.9 Payment Window for Inflexible Travelers	171
		5.9.1.10 Ticket Window for Inflexible Travelers	172
	5 10	Chapter Summary	172
	5.10		. 1/2
6.	CON	CLUSIONS	173
0.	6.1	Dissertation Summary	173
	6.2	Research Findings and Contributions	. 175
	~ • • •	6.2.1 Objective 1	. 175
		6.2.2 Objective 2	. 176
		6.2.3 Objective 3	177

6.3 Future Work	178
REFERENCES	179
LIST OF PUBLICATIONS	195

APPENDICES

A. Survey on Evaluation of Online Airline Reservation Systems	196
B. UFB in terms of Compromising on Service Quality Attributes	199
C. User's Flexibility towards Flexible Online Airlines Reservation Systems	202
D. Transformation of Evaluators Users' Flexibility Scoring	204
E. Classification of Travelers' on the Basis of their FTB	205
F. Permission to Collect Data from Travel Agents for Research Purposes	206
G. Permission to Collect Data from Travel Agents at MATTA fair 2011	207
H. Usability Evaluation Checklist	208

LIST OF TABLES

Table 2.1 : Definitions of Flexibility in the Context of System Engineering)
Table 2.2 : Comparison Chart on Offered Features by SBTs and OTAs 26	,)
Table 3.1 : Research Questions and Hypothesis 45)
Table 3.2 : Emerging Themes on Factors Influencing upon FTB and PU)
Table 3.3 : Uni-directional Scale to Measure Users' Flexibility))
Table 3.4 : Transformation Scoring Scale 69)
Table 4.1 : Coefficient of the Factors Affecting Usability of Online Systems	ŀ
Table 4.2 : Flexibility of Existing Systems 87	7
Table 4.3 : T-Test on User's Flexibility with SBTs 91	
Table 4.4 : One-Way ANOVA on Satisfaction Level with Existing SBTs	ŀ
Table 4.5 Multiple Comparisons on Satisfaction Level with Existing SBTs	ŀ
Table 4.6 : T-Test on User's Flexibility with Existing OTAs	7
Table 4.7 : ANOVA on Satisfaction Level with Existing OTAs 99)
Table 4.8 : Multiple Comparisons on Satisfaction Level with Existing OTAs10	0
Table 4.9 : Mean & SD on the Recommendation of Integrating OTA Features 10	2
Table 4.10 : Two-way ANOVA on the Satisfaction Level with Existing OARSs . 10	13
Table 4.11 : Demographics of Online Travel Forums Respondents 10	4
Table 4.12 : Results of Online Travel Forums 10	15
Table 4.13 : Demographics of in-depth Interview 10	6
Table 4.14 : Results of in-depth Interviews 10	17
Table 4.15 : Demographics of Focus Group Interviews 10	17
Table 4.16 : Results of Focus Group 10	18
Table 4.17 : Users' Flexibility Transforming Scoring Scale	19
Table 4.18 : Range for Classification of UFB on the basis of UFS11	0
Table 4.19 : Interrelationship b/w UFB & their Perceived Usability of FOARS 11	1
Table 4.20 : Pearson Correlations on Interrelationship between UFB and SF11	3
Table 4.21 : One Way Independent ANOVA with PU and System's Flexibility 11	4
Table 4.22 : ANOVA with PU and System's Flexibility11	5
Table 4.23 : ANCOVA by Introducing System's Flexibility as Covariate	.5
Table 4.24 : Correlations b/w Flexible Personality, SQA and External Variables 11	8

LIST OF FIGURES

Figure 2.1 : Two Different Perspectives of Flexibility	14
Figure 2.2 : Different Elements of Human Computer Interaction	16
Figure 2.3 : Flexibility as the Multiplicity of Ways for Information Exchange	16
Figure 2.4 : Building Construct of Cognitive Psychology	17
Figure 2.5 : Conceptual Linking between SF and UF	19
Figure 2.6 : Distribution Channels for Airline Reservation Systems	22
Figure 2.7 : U.S. Travel Market by Channel, 2008 and 2011	25
Figure 2.8 : Lowest Airfare Search Results through Matrix Display	27
Figure 2.9 : Alternative Airport Search to Find the Lowest Possible Fare	27
Figure 2.10: Hotel Search, Results Display & Sorting	28
Figure 2.11: Opaque Fares Offered by Priceline for Discounted Tickets	29
Figure 2.12: Opaque Fares Offered by Hotwire for Discounted Tickets	30
Figure 2.13: Flexible Date Search Results by Hotwire	30
Figure 2.14: Ticket Reservation through Intermediate	38
Figure 3.1 : Research Methodology	44
Figure 3.2 : Research Model for the Hypothesis to be Tested	51
Figure 3.3 : User's Analysis of the Participants from UTP	53
Figure 3.4 : User's Analysis of Participants from Travel Agencies	56
Figure 3.5 : User's Analysis of Participants from UTP for Study 2	56
Figure 3.6 : User's Analysis of Participants from Online Travel Forums	63
Figure 3.7 : User's Analysis of Participants from In-depth Interviews	63
Figure 3.8 : User's Analysis of Participants from Focus Group	63
Figure 3.9 : User's Analysis of Participants for Transformation Scale	67
Figure 3.10: Proposed FOARS after Integrating Opaque Fair into SBTs	71
Figure 3.11: Increased Usability by Improving SF and Users Decision	71
Figure 3.12: Prototype of a Flexible Booking Window for Flexible Travelers	75
Figure 3.13: Prototype of a Flexible Booking Window for Inflexible Travelers	75
Figure 4.1 : Box Plot showing Satisfaction Level with Existing SBTs	89
Figure 4.2 : Means Plot on Satisfaction Level with Existing SBTs	91
Figure 4.3 : Error Bar on Satisfaction Level with Existing SBTs	92

Figure 4.4 : Box Plot on Satisfaction Level with Existing OTAs
Figure 4.5 : Means Plot on Satisfaction Level with Existing OTAs
Figure 4.6 : Error Bar on Satisfaction Level with Existing OTAs
Figure 4.7 : Means Plot on the Recommendation of Integrating OTA Features 101
Figure 4.8 : Data Distribution after Classifications of Users'
Figure 4.9 : Scatter Plot between Users' Flexibility and SQAs
Figure 4.10: Scatter Plot between Users' Flexibility and External Variables
Figure 4.11: Effect of User Classification of FB on Effectiveness
Figure 4.12: Effect of Existing & Proposed Systems on Effectiveness
Figure 4.13: Interaction Effect of Existing & PS on Effectiveness
Figure 4.14: Effect of User Classification of FB on Efficiency
Figure 4.15: Effect of Existing and Proposed Systems on Efficiency
Figure 4.16: Interaction Effect of Existing & Proposed Systems on Efficiency 126
Figure 4.17: Effect of User Classification of Flexible Behavior on Satisfaction 128
Figure 4.18: Effect of Existing and Proposed Systems on Satisfaction
Figure 4.19: Interaction Effect of Existing and PS on Satisfaction
Figure 4.20: Linear Relationship between Effectiveness and Efficiency
Figure 4.21: Linear Relationship between Effectiveness and Satisfaction
Figure 4.22: Linear Relationship between Efficiency and Satisfaction
Figure 4.23: Bar Chart Showing Effectiveness of the Proposed System
Figure 4.24: Bar Chart Showing Efficiency of the Proposed System
Figure 4.25: Bar Chart Showing Satisfaction of the Proposed System
Figure 5.1 : Comparison of Existing and Proposed Reservation Approaches 166
Figure 5.2 : Home Page for Flexible Online Airline Reservation System
Figure 5.3 : Booking Window for Flexible Online Airline Reservation System 168
Figure 5.4 : Flight Search Window for Flexible Travelers
Figure 5.5 : Travelers' Details Window for Flexible Travelers
Figure 5.6 : Payment Method for Flexible Travelers
Figure 5.7 : Immature Ticket for Flexible Travelers
Figure 5.8 : Flight Search Window for Inflexible Travelers
Figure 5.9 : Travelers' Details Window for Inflexible Travelers
Figure 5.10: Payment Method for Inflexible Travelers
Figure 5.11: Confirmed Ticket for Inflexible Travelers

LIST OF ABBREVIATIONS

ARS	Airline Reservation System
ACM	Association for Computing Machinery
B2C	Business to Consumer
CRS	Computer Reservation System
CI	Confidence Interval
FTB	Flexible Traveling Behavior
FOARS	Flexible Online Airline Reservation System
FG	Focus Group
FR	Functional Requirements
GDS	Global Distribution System
GUI	Graphical User Interface
HCI	Human Computer Interaction
ID	In-depth
NFR	Non-Functional Requirements
OTA	Online Travel Agencies
OCBT	Online Corporate Booking Tool
OARS	Online Airline Reservation System
OTF	Online Travel Forums
PU	Perceived Usability
PLF	Passenger Load Factor
SBT	Self-booking Tool
SF	System Flexibility
SABRE	Semi-Automated Business Research Environment
SIGCHI	Special Interest Group on Human Computer Interaction
SQA	Service Quality Attributes
TAM	Technology Acceptance Model
TPB	Theory of Planned Behavior
UF	User Flexibility
UFS	User Flexibility Score

CHAPTER 1

INTRODUCTION

1.1 Research Background

Airline Reservation Systems are the computerized systems that are used for storing and retrieving information in order to conduct air travel related transactions [1]. Most of the airlines have their own Self-booking Tools (SBTs) also referred to as Online Corporate Booking Tools (OCBTs) that provide an opportunity to their clients to make online reservations [2]. Any sale made through airline's offices directly or through their SBTs is referred to as direct sale. On the other hand, Global Distribution Systems (GDSs) connects airline offices with the offline and Online Travel Agencies (OTAs) [3]–[5]. GDSs book and sell tickets for multiple airlines. Any sale made through GDSs is referred to as sale through intermediaries.

Airlines opt for selling tickets through direct channels and also through intermediaries. However, in order to differentiate their reservation channel from others and to increase the direct sale, airlines have invested heavily in deploying a range of tactics, such as, featuring their web site URLs across the marketing and advertising communications, Web fares, reward mileage bonuses, and negative incentives for non-preferred booking channels [6]. In addition to this many airlines also dispose off their distressed inventory by providing last-minute sale discounts in order to secure incremental revenue, where airline offers its unsold inventory at heavily discounted prices before it perishes. This selling approach is adopted because it does not disrupt the existing distribution channels or retail pricing structure [7] and becomes a productive source of incremental revenue for the airline. The best part of this form of selling is that, although travelers enjoy highly discounted fares, they do not have to make predictions or face extremities in predicting specifies of their traveling itineraries. For example, Malaysian airline has launched last minute flights in 2008 to increase the revenue and average load factor [8]. According to the managing director of Malaysia Airline, Datuk Seri Idris Jala "*the everyday low fares will create new demand for people who do not fly with Malaysia Airline*", hence potentially increasing the average passenger load factor from 70% on each flight by filling up unsold seats. Besides all these tactics made by the airlines, however, more than 50% sale is done through GDSs [9]. Furthermore, researchers [10] reported that this type of direct selling at the last-minute could be very risky for the airline, since the potential travelers may prefer waiting for last-minute sales and not purchase in anticipation of heavy discounts [10]. Such a condition may put an airline in a very risky position with potential possibility of revenue loss. That is why this practice is substantially criticized by analysts and researchers, who refer to it as a vivacious cycle of price degradation that can eventually destroy the airlines [11].

The second way to dispose-off distressed inventory is through opaque channels. The term 'opaque inventory' indicates selling of unsold travel inventory at heavily discounted price and it is called as being opaque, because some of the attributes of the service supplier such as, name in the case of airline or hotel etc., are kept hidden and only revealed to the traveler once the purchase has been materialized [12]. It is called opaque selling because of its innovative mechanisms for marketing and price discrimination [13], [14]. Opaque inventory selling is like a box, full of surprises and travelers who are interested in buying opaque inventory products are high in price sensitivity and low in specifies of travel plan. Thus this form of selling immediately captures the attention of travelers who would like to keep their travel expenses within limited budgets. As for the airlines, in order to minimize effect of price degradation on their revenue, accepted the role played by opaque selling intermediaries, so as to meet uncertain demand situations [15], [16].

But opaque selling has its own share of demerits. Firstly, opaque selling through intermediaries yields higher incremental revenue due to the uncertainty in demand, this means in case of no or little demand uncertainty, selling through the opaque channel will faintly increase profits for the airline, when comparing to direct selling channel [13]. Moreover, opaque channel on one side increase sales by attracting price-sensitive travelers who may otherwise not purchase, at the same time it also causes

reduction in sales of the transparent online SBTs of airlines and of the offline channels such as traditional travel agencies [12]. Earlier opaque selling intermediaries performed much better because there was little or no competition. However, today with the boom in Online Airline Reservation Systems (OARS), the price of discounted products may not vary much because potential buyers will be dispersed by different opaque selling intermediaries that will be standing against each other, striving to steal market share and forming a tacit collusion to keep prices high so as to make profits [12]. This will minimize incremental revenue of a particular airline [17]. In fact, with growing competition in opaque selling intermediaries, product differentiation has become difficult [18], so does branding and building customer loyalty [19].

While explaining the reasons, why people prefer intermediaries over online booking systems, researchers [20], [21] argued that earlier Business to Consumer (B2C) systems were not flexible as they work for simple closed requests, i.e. a request that can be directly mapped into formalized terms or predefined parameters, such as dates, airports, flights etc. Furthermore, these systems could break down for more complex requests, i.e. a request where customer is flexible with regards to attributes such as date and destination. Therefore, Malizia and Olsen [20] have recommended an information system between a customer and booking system to replace intermediaries. However, the solution recommended by these researchers only covers pre-sale flexibility issues. In other words, these systems provide flexibility only in terms of providing general information, which could be useful in taking decision with respect to pre-sale flexibility. So the question is if current SBTs are flexible enough? The term flexibility here should not only be related to the booking. In case of 'e-ticketing', problems arise when a traveler changes his/her mind or if the airline decides to make changes with regards to times, dates, destinations, after receiving a final confirmation of the booking. Therefore, actual replacement of human agent with a virtual intermediate system could be attainable in a post-sale flexibility scenario, if it is really supported in terms of 'flexibility'.

According to the Special Interest Group on Human-Computer Interaction (SIGCHI) of the Association for Computing Machinery (ACM) "Human Computer

Interaction is a discipline concerned with the design, evaluation and implementation of interactive computing systems for human use and with the study of the major phenomena surrounding them." And within the study of HCI, human actions are processed by computers; as a result interaction occurs between the two. This means, humans make computer perform operations. Therefore, it is necessary to understand such an interaction in the contexts of flexibility as well. For that reason, flexibility can be discussed from these two different perspectives, i.e. (1) System's Flexibility (Computer) and (2) Users' Flexibility (Humans).

1.2 Problem Statement

In designing a flexible system, it is inadequate to understand and provide only the System Flexibility. It is equally important to also address the flexibility on the users' side. Researcher [22] argued that even though internet is referred as a major technological innovation of today, its success heavily depends upon assimilation of customer expectations and preferences into the design and content of websites. In airline industries, Users' Flexibility is more prevailing as most airlines adopted last-minute ticket selling strategy and opaque selling through which travelers can enjoy highly discounted fares at the price of their flexibility on, for instance, the traveling dates, time or itineraries. Previous Human Computer Interaction (HCI) researches, however, rarely addressed the users' perspectives flexibility when designing systems interface for usability and flexibility of Online Airline Reservation Systems. Therefore, it is important to design self-booking tools in view of customers' preferences, expectations and online usage behavior in order to increase the Perceived Usability of such systems.

Usability of a system can be evaluated on the basis of performance of its different functions and from literature it is noticeable that flexibility of a system is one of the guiding principles that provide support to achieve, develop or improve its usability. Therefore, it is very critical for the organizations to design flexible systems that are easy to use and can accomplish all the requirements by way of offering customizability. And one should not overlook, that flexibility is mirrored in functional requirements as well [23]. A system is considered usable if users can accomplish their

tasks easily. Similarly, a system is considered functional if it offers all the functions required by a user to perform their tasks [24].

Norman's philosophy [25] says that users are good in adapting to a system. However, research shows users' dissatisfaction with existing Online Airline Reservation Systems in terms of task completion [26]. Therefore, researchers are eager to find out ways for improving online usability of the systems, how users' Perceived Usability of a system is formulated by its flexibility functions [27]–[31]. In addition to techniques, methods and guidelines proposed for designing usable systems, HCI researchers have also long argued on the importance of human factors in designing and implementation of user-centred designs. According to Nielsen [32], *"users experience usability of a site before they have committed to using it and before they have spent any money on potential purchases"*. This indicates users' Perceived Usability in online digital environments is an important determinant for evaluating their satisfaction in the same environment. The existing literature can be divided into the following four research aspects where researchers currently are focusing upon to determine usability of online systems:

- Usability Perception by Performing Content Analysis
- Usability Perception through User's Internet Adoption
- Usability Perception based on Users' Preferences and Expectations
- Usability Perception based on Online Behavior of Web Users

This research uses a blended approached and combines the above four research areas to determine usability perception of the Online Airline Reservation Systems. It is important because customers' usability expectation and preferences from Online Airline Reservation Systems lacks research and empirical findings. Law and Leung [33] had emphasized upon the need to investigate expectations of airline customers that book their itineraries through their online self-booking tools. Moreover, the existing evaluation of online tourism websites is performed by researchers and not by customers. It leads to a dilemma and research gap that does not potentially address expectations of travelers. This research therefore examines travelers' expectations, preferences and online behavior (User's Flexibility) and aligns that with designing of flexible Online Airline Reservation Systems (System's Flexibility) and users' as evaluators of the online systems to determine its usability (Perceived Usability).

1.3 Research Aims and Objectives

This research aims to provide a framework for designing a more flexible Online Airline Reservation Systems through investigating the associations between System's Flexibility, Users' Flexibility and Perceived Usability of Online Airline Reservation Systems. This was an exploratory approach and would lead to a better understanding of the interrelationship between System's Flexibility, Users' Flexibility and Perceived Usability of Online Airline Reservation Systems; and provide a basis for future studies to formally develop design guidelines and/or usability metrics in the flexibility context. To aid this aim the following research objectives are defined to address the corresponding research questions/hypotheses:

1. To assess user needs (System's Flexibility and Users' Flexibility) associated with Online Airline Reservation Systems.

The research questions are:

RQ1: What are the issues with flexibility of Online Airline Reservation Systems, whether or not flexibility is one of the reasons for users not using such systems?

RQ2: To what extend flexible users can compromise with service quality attributes of Online Airline Reservation Systems?

RQ3: How users' satisfaction with an existing SBTs is rated against their choice of OTA feature and reflected in their integration assessment of the same for making SBTs more flexible Online Airline Reservation Systems?

 To propose a framework for designing more flexible Online Airline Reservation Systems while classifying users on the basis of their Flexible Traveling Behavior. The research questions are:

RQ4: How users' perception on factors influencing Flexible Traveling Behavior and Flexible Online Airline Reservation Systems is determined?

RQ5: How to classify Users' on the basis of their Flexible Traveling Behavior into High, Medium and Low flexible and how to investigate interrelationships among Users' Flexibility, System's Flexibility and Perceived Usability of existing Online Airline Reservation Systems?

3. To study the interrelationship between System's Flexibility, Users' Flexibility and Perceived Usability of Online Airline Reservation Systems and to determine the Perceived Usability of the existing and proposed systems.

The research questions are:

RQ6: How do service quality attributes of airlines and external variables jointly predict flexible behavior of travelers?

RQ7: How does user Perceived Usability with the existing and the proposed system differs?

RQ8: Is there a multivariate main effect of user's Flexible Traveling Behavior (High, Medium and Low) on effectiveness, efficiency and satisfaction of the proposed system?

1.4 Research Methodology

The methodology of this thesis consists of three phases which are described below:

• Phase I: Assessing User Needs (System's Flexibility & Users' Flexibility)

Phase I was designed to achieve the 1st research objective. The existing Online Airline Reservation Systems were used to assess the System's Flexibility and Users' Flexibility. In the existing Information Systems research and literature, no study has been found to address consumer behavior on opaque selling with respect to Online Airline Reservation Systems. Therefore, this phase will help to explore the small but growing literature in designing of Online Airline Reservation Systems by modeling upon flexible behavior of travelers.

Three pilot studies were conducted in this phase as shown below:

- 1. A study to investigate issues with flexibility and if flexibility is the reason for not using Online Airline Reservation Systems.
- 2. A study to investigate users' flexible behavior in terms of compromising on the service quality attributes of an airline.
- 3. A study to examine if integration of OTAs features can make SBTs more Flexible Online Airline Reservation Systems?
- Phase II: Classification of Users (Interrelationship Testing of Variables)

Phase 2 was designed to attain the 2nd research objective. This phase intends to carry out an extensive relationship testing of variables and their sub-measuring constructs so as to evolve a framework for designing of Flexible Online Airline Reservation Systems. Two detailed studies were conducted in this phase with the following study objectives:

- A qualitative enquiry to explore the concept of users' perception on factors influencing Flexible Traveling Behavior and Flexible Online Airline Reservation Systems.
- A study to classify Users' on the basis of their Flexible Traveling Behavior (High, Medium, Low) and to investigate interrelationships among System's Flexibility, Users' Flexibility and Perceived Usability of existing Online Airline Reservation Systems.
- **Phase III:** Case Study (Testing the Framework)

Phase III of this study is related to a design case study and the corresponding analysis to conquer the 3rd research objective. Participants were requested to complete the usability evaluation of the existing and proposed interfaces.

Quantitative technique was used to analyze the data collected in the context of System's Flexibility, Users' Flexibility and Perceived Usability of the systems.

A detailed research methodology is presented in Chapter 3.

1.5 Scope of Research

Global Distribution Systems allow users to make reservations, from hotel booking to car rentals, from railway reservation to e-ticketing. However, the scope of this research is limited to Online Airline Reservation Systems only, as the research focuses upon designing a more flexible Online Airline Reservation Systems in lieu of users' flexible behavior. Thus, the scope of this research is further limited to the Malaysian perspective.

Forrester research [33], estimates seventy million consumers searching online for travel plans in July 2006, thus making online travel bookings the single largest component of e-commerce. Different users have different needs, interests and wishes to be served and system's effectiveness, efficiency and satisfaction may vary from one user to another based on their usability perception. For some users a system may be very effective but this may not be true for all. Therefore, usability of any website cannot be improved without considering consumer intend or user behavior. Furthermore, clear understanding of consumer intent and behavior in the case of online airline ticket shopping and elsewhere cannot be achieved without considering the factors that affect purchase decisions [34]-[36]. The reason is selling products online are very different from selling in physical market and this requires a clear understanding of online customer interest due to absence of face-to-face interaction with customers [20], [21], [37]. Internet marketing strategies can be adjusted if cyber marketers know what the consumers want and how they reach their decisions. Similarly, such an understanding will help Web designers to develop sites making that are not only popular but also flexible and effective for sales [38], [39].

The anticipation of travelers for low fares is an extremely important concern, that airlines are faced with every day. Anticipation of travelers for low fares, gives an idea about their traveling behavior, which is 'the extent to which a traveler anticipates for a low fare', indicating the extent to which 'a traveler is ready to compromise on flying conditions', and thus becoming flexible in accepting what is being offered to them by an airline. As discussed in Chapter 2, Section 2.3.1, airlines are not in a strategic position to offer low cost fares directly to their customers for selling left over inventories, due to a number of potential threats to their revenue generation. Therefore, the only business model that addresses this concern is opaque selling by OTAs, and scope of this research is narrowed down to examining the flexible behavior of Malaysian travelers in order to design more flexible Online Airline Reservation Systems that may increase the Perceived Usability of Online Airline Reservation Systems.

1.6 Research Contributions

The first contribution of this research is the development of a framework that could be used for further studies and to design more flexible Online Airline Reservation Systems. The framework is a general framework that can be applied to different reservation systems; however, this research particularly addresses the airline reservation systems.

The second contribution is towards the development of users' flexibility measuring scale. The non-availability of an absolute scale to measure flexibility turns the investigation into a cumbersome effort for researchers and practitioners. It is difficult to even make any rough assumptions about the extent to which the users would like to have additional flexibility features in online reservation systems. Therefore, this research builds on previous and ongoing work within the disciplines of *Human Computer Interaction* by introducing psychometric scales to measure users' flexibility in terms of compromising on service quality attributes of an airline.

The third contribution of this research is within the area of *Operational Management*, as it introduces a new approach of reservations to increase the Passenger Load Factor (PLF). Under utilization of the resources, such as, air plane capacity is one reason of low PLF and increased number of flights. Using traditional

airline reservation systems, airlines either cancel the flight at 11th hour or send flight with minimal profit margin, even if they receive lesser bookings in any particular flight. One solution to address this concern is by leveraging upon travelers' flexibility. In this research it is proposed to send flights fully occupied, which in turn could reduce the number of flights that actually take-off per week.

The fourth contribution of this research is to add up in the growing literature. The concept of SBTs for disposing off their opaque inventory directly has not been adequately considered in *Information System Research and Literature*. The concept requires extensive research especially in academic discipline [40] and as highlighted by Jerath *et al.* [41] a number of studies have focused upon airline revenue management systems, however attempts to empirically verify those findings are a few. Furthermore, the existing opaque selling literature lies at the intersection of consumer behavior and revenue management operational strategies [40]. However no study has been found to address consumer behavior on opaque selling with respect to Online Airline Reservation Systems as most recent papers as well as researches fall within the marketing domain [40]. This research therefore contributes to the small but growing literature in designing of Online Airline Reservation Systems by modelling upon flexible behavior of travelers.

The fifth contribution of this research is the IS theory, based on empirical findings and analysis in support of the proposed framework for Flexible Online Airline Reservation Systems. Theory building from case studies is considered to produce novel theory, and is testable with constructs that can be readily measured and hypothesis that can be proven false.

The sixth contribution of this research is the development of a prototype in terms of proposed Graphical User Interfaces (GUI) to handle flexible and inflexible travelers differently. If a system provides ranges of dates as flying and source destination options at different fares, flexibility of the system is enhanced in its Perceived Usability in the eyes of the flexible travelers.

Finally, this research provides empirical results of the real case studies on the existing and proposed system.

11

1.7 Organization of the Thesis

This dissertation is divided into the following 6 chapters:

Chapter 1 provides a research overview. It gives the research background while defining the problem statement, research aim and objectives, research methodology, research scope and contribution. Finally, it outlines the overall chapters of the dissertation.

Chapter 2 presents the literature review. This chapter introduces flexibility of Online Airline Reservation Systems from two different perspectives (i.e. System's Flexibility and Users' Flexibility) and also provides a conceptual linking between the two. Furthermore, different aspects of Perceived Usability have been discussed in this chapter in order to provide a conceptual linking between the usability of Online Airline Reservation Systems and Flexible Traveling Behavior of the travelers.

Chapter 3 describes the research methodology of the thesis. The overall methodology is divided into three phases. Phase I addresses user needs (System's Flexibility and Users' Flexibility), Phase II gives classification of users and interrelationship testing of variables and, Phase III provides a case study.

Chapter 4 reports the statistical analysis of the research. It includes results of different pilot studies and the case study. It follows the research questions to organize the results obtained through corresponding hypotheses.

Chapter 5 presents the discussion of the dissertation. This chapter follows the same pattern by discussing and elaborating facts related to individual research questions reported in Chapter 4. Moreover, recommendations for the proposed Flexible Online Airline Reservation Systems are also discussed in this chapter.

Chapter 6 summarizes the thesis. It highlights the research and emphasizes on the importance of the proposed framework. Moreover, major findings of the work and recommended directions for the future work are also presented in this chapter.

CHAPTER 2

LITERATURE REVIEW

2.0 Chapter Overview

In this chapter System's Flexibility, Users' Flexibility and Perceived Usability of the Online Airline Reservation Systems is presented. Section 2.1 is dedicated to explore flexibility from two different perspectives (i.e. System's Flexibility and Users' Flexibility) and to find out the conceptual link between the two. Section 2.2 covers different aspects of usability perception and provides a conceptual link between the usability of Online Airline Reservation Systems and Flexible Traveling Behavior of the travelers. Section 2.3 is devoted to explore the relationship between flexibility and usability and also to study the role of opaque selling in System's Flexibility and User's Flexibility. Section 2.3 presents the summary of the chapter.

2.1 Flexibility and Online Airline Reservation Systems

Flexibility of a system is one of the guiding principles that provide support to achieve, develop or improve its usability. It is very critical for the organizations to design flexible systems that are easy to use and can accomplish all the requirements by way of offering customizability. Given the importance of internet shopping as a source of income for the airlines and high user demand, in-depth research is required.

2.1.1 Flexibility Concepts

The notion of flexibility has been addressed in many disciplines and from many different perspectives. The oxford university's dictionary on Business and Management defines flexibility as "the ability to adapt an operating system to respond

to changes in the environment" [42]. In case of manufacturing, one may distinguish eleven different classes of flexibility: machine, material handling, operation, process, product, routing, volume, expansion, program, production and market flexibility [43]–[45]. In the discipline of systems engineering, the flexibility of a system is understood as "the ability to respond to change" [46]. Product design literature defines flexibility "as the ability of companies to frequently upgrade their products to meet the rapidly changing technologies" [47], [48]. Each of the above definition defines flexibility in a different perspective, but the fundamental meaning of this term remains consistent across all definitions which: "able to flex."

As defined by the Special Interest Group on Human-Computer Interaction (SIGCHI) of the Association for Computing Machinery (ACM) "Human Computer Interaction is a discipline concerned with the design, evaluation and implementation of interactive computing systems for human use and with the study of the major phenomena surrounding them." In HCI, human actions are processed by computers; as a result interaction occurs between the two. This shows that humans make computer perform operations, therefore, it is very important to understand human computer interaction in the context of flexibility as well. Hence, flexibility can be discussed from these two different perspectives, i.e. (1) System's Flexibility (Computer) and (2) Users' Flexibility (Humans) as shown in Figure 2.1.



Figure 2.1: Two Different Perspectives of Flexibility

2.1.1.1 System's Flexibility

Within the HCI discipline, System's Flexibility is referred to its ability to respond to internal or external changes. However, the ambiguous characteristic of word "flexibility" [46] has forced authors to explain flexibility differently as shown in Table 2.1.
Human computer interaction studies are conducted to *develop* or *improve* the safety, utility, effectiveness, efficiency, usability, appeal of the systems that include computers as shown in Figure 2.2. Of which, Usability of the systems is described by researchers as a "measure of the ease with which a system can be learned and used, its safety, effectiveness and efficiency, and attitude of its users towards it" [49]. While, ISO defines Usability as "the extent to which a product can be used by specified users to achieve specified goals with effectiveness, efficiency and satisfaction in a specified context of use" (ISO 9241-11). The principles that provide support to achieve, develop or improve Usability of the system include, (1) Learnability, which is the ease with which users can use the system effectively, (2) Robustness, which is the level of support provided to the user to achieve its goals and (3) Flexibility, which is basically multiplicity of ways the user and the system exchanges information.

Author	Definition of Flexibility	Reference
Nilchiani	We define flexibility as the ability of a system to respond to potential internal or external changes affecting its value delivery, in a timely and cost-effective manner. Thus, flexibility is the ease with which the system can respond to uncertainty in a manner to sustain or increase its value delivery. It should be noted that uncertainty is a key element in the definition of flexibility. Uncertainty can create both risks and opportunities in a system, and it is with the existence of uncertainty that flexibility becomes valuable.	[50]
Saleh	Flexibility should be sought when: 1) the uncertainty in a system's environment such that there is a need to mitigate market risks, in the case of a commercial venture, and reduce a design's exposure to uncertainty in its environment, 2) the system's technology base evolves on a time scale considerably shorter than the system's design lifetime, thus requiring a solution for mitigating risks associated with technology obsolescence.	[51]
Ross	The only difference between flexibility and adaptability is the location of the change agent with respect to the system boundary: inside (adaptable) or outside (flexible). Of course the system boundary could be redefined, changing a flexible change into an adaptable one, or vice versa. The fungible nature of the definition is often reflected in colloquial usage and sometimes results in confusion. If the system boundary and location of change agent are well- defined, confusion will be minimized.	[52]

Table 2.1:	Definitions of	of Flexibility	in the Context	of System	Engineering
				~	<u> </u>



Figure 2.2: Different Elements of Human Computer Interaction

Of the three principles, Flexibility is related to taking input/output in different forms and examined with respect to (1) Dialogue initiatives, (2) Multi-threading, (3) Task migratability, (4) Substitutivity and (5) Customizability as shown in Figure 2.3. Customizability refers to adaptability of interfaces to suit different needs, and it is achieved by way of (i) adaptability, where users can adapt the user interface, (ii) adaptivity, where the user interface can be adapted by the system and (iii) personalization, where the user interface is tailored towards the individual user. While system driven interaction hinders its flexibility, user-driven interaction is considered to be strongly favourable.



Figure 2.3: Flexibility as the Multiplicity of Ways for Information Exchange

2.1.1.2 Users' Flexibility

The User's Flexibility is nothing but users' ability to rapidly change from one course of action to another, i.e. "flexible behavior", and it is referred as a hallmark of human cognition system [53]. Webster's Dictionary defines cognition as "the act or process of knowing in the broadest sense; specifically, an intellectual process by which knowledge is gained from perception or ideas". Empirical research into cognition is usually scientific and quantitative, and involves formation of mental models to describe or explain certain behaviors. In context of flexible behavior of users, human cognition system may thus be examined from the perspective of cognitive psychology. As mentioned earlier, Human-Computer Interaction (HCI) research is intended to explain interaction between humans and the computer technology. And in order to provide a scientific explanation to human behavior (e.g. user interface design, information visualization, etc) many principles, theories and concepts from cognitive psychology are deployed in HCI [54]–[60] such as Perception, Categorization, Memory, Knowledge Representation, Language and Thinking as shown in Figure 2.4.



Figure 2.4: Building Construct of Cognitive Psychology

Thinking refers to any intellectual or mental activity resulting in ideas or arrangements of ideas and within the context of HCI, thinking simulates human behavior, which is eventually translated as an action taken on part of users, in the form of making choices, performing logical operations, formation of concepts, problem solving and decision making [61]. For understanding flexible human behavior, decision making is an important reflection of users thought process. Decision making is a mental process which results in selection of a course of action among several alternatives. At the end of every decision making process, an output is produced in the form of a final choice or selection, which can be in the form an action, or an opinion of choice. Decision making process is an active research area since it examines decisions of users in context of their unique set of needs and preferences, therefore, reflection of users' flexible behavior can be seen in the decisions they make [62], [63].

2.1.1.3 Conceptual Linking between System's Flexibility and Users' Flexibility

In order to understand users' flexible behavior, it is first important to understand the contexts that govern users' behavior towards being flexible. In case of System's Flexibility, users interact with computer systems in order to accomplish tasks. While the System's Flexibility is reflected in its customizability features, therefore, developing an understanding of system's customizability in terms of affecting users flexible behavior, requires a science base in the form of systematic knowledge of what governs user's flexible behavior' and influencing upon their decision making process as shown in Figure 2.5. Thus, three variables have been identified in this basic conceptual framework: (i) System's Flexibility, (ii) Users' Flexibility and (iii) System's Usability.

From literature review, it is found out that users' flexible behavior is reflected in their decision making process, while System's Flexibility is translated in its customizability features.

- 1. System's Flexibility has a linear relationship with User's Flexibility due to the following assumptions:
 - A flexible system (customizable) can reinforce users' flexible behavior by influencing upon their decision-making, even if they were inflexible or partially flexible initially [25].

- On the contrary, if a user is flexible with respect to making decisions, it cannot still reinforce System's Flexibility through customizability, even if it was inflexible or partially flexible, initially.
- System's Flexibility is one of the principles that provide support to achieve, develop or improve usability of the system. System's flexibility thus has a linear relationship with System's Usability.
- 3. Users' flexible behavior in terms of their decision making influence upon the usability of a system. Thus Users' Flexibility has a linear relationship with System's Usability.



Figure 2.5: Conceptual Linking between System's Flexibility and Users' Flexibility

From literature it is concluded that flexibility can be discussed from two different aspects, one from System's perspective (Computers), and second from User's perspective (Human). System's Flexibility translates into its customizability in achieving the defined usability objectives of the system which are effectiveness, efficiency and satisfaction.

Likewise, in context of User's Flexibility, this section concludes that different users may have different needs, interests and wishes to be served and system's effectiveness, efficiency and satisfaction may vary from one user to another based on their usability perception. For some users a system may be very effective but this may not be true for all. This drives the need of integrating cognitive ergonomics into the framework, to understand Users' Flexibility in designing of systems. Moreover, from literature it is found that Users' Flexibility is reflected in their decision making behavior. Further elaboration and validation of the conceptual linking between System's Flexibility, Users' Flexibility and Usability can be found in the methodology chapter Section 3.2.1.

2.1.2 Airline Reservation Systems

Airline Reservation Systems (ARS) keep record of airline schedules, fare tariffs and passenger reservations. ARS are developed to enable productive and effective flight reservations for an airline. ARS eventually evolved into the Computer Reservation System.

2.1.2.1 Computer Reservation Systems

Computer Reservation Systems (CRSs) are the computerized systems used for storing and retrieving information such as, airline reservation systems, car rental systems and, hotel reservation systems [1].

CRSs became increasingly popular due to their immense potential in handling of reservations and companies could foresee an increase in their yield matrices. However, CRSs offer advantages but the strength of their positivity depends upon how well and at what level systems have been integrated [64].

CRSs are equipped with enhanced functionalities and features that provide companies with an integrated one stop solution to manage sales, customer relationship management, marketing plans, resource planning and personalized customer care and attention. With these features, CRSs help in processing reservations and at the same time support decision making processes. The development of CRSs started at the beginning of the 60s when the first electronic travel booking was launched by Semi-Automated Business Research Environment (SABRE). Subsequently, American Airlines and IBM joined hands with SABRE to launch first of many airlines owned and operated CRSs.

Initially, these systems were used at airlines' basic and internal reservation centres, but its true potential was realized quickly and travel agencies grabbed the opportunity for its deployment. It helped travel agencies immensely in terms of providing accurate schedules to travelers, fares, instant information on availability of seats and extended efficiency internally as well with respect to strengthening the distribution channels. However, CRSs came with their share of drawbacks as well. CRSs did help in reducing the costs of travel agencies when compared to manual reservation systems that were based on telephone confirmation and checking [65], however, they were still criticized as "inflexible dinosaurs" since they were not adaptable enough to meet growing business demands, that requires robustness with regards to offering services and additional features for reducing high distribution costs in a more flexible manner [1]. Airlines, being the true originator of CRSs enjoyed more competitive edge than problems that had arisen due to inflexibility of CRSs. Especially when comparing with travel agencies, airlines were in a more control situation with respect to scheduling of flights and could even influence upon market share. For example, in 1985, U.S. travel agency sales had risen to \$54 billion, which was more than nine in 10 agencies with sales greater than \$1 million and had deployed CRSs. Travel Agency revenue had surged 400% over the same period, while agency employment increased by only 20% [9].

In early nineties, with the consortium of four large CRSs companies. Global Distribution Systems (GDSs) had emerged into the scene. It is important to mention here that CRSs are not to be confused with GDSs since they are electronically connected to one another. CRSs run on mainframes, minicomputers or microcomputers and are connected through data communication links to terminals within various branches of the company for bookings. On the other hand, GDSs are the systems that book and sell tickets for multiple airlines and use internet gateways to allow users for making reservations, from hotel booking to car rentals, from railway reservation to e-ticketing as shown in Figure 2.6. The emergence of GDSs and their

connectivity with CRSs has brought hundreds of thousands of travel agents and other distributors with thousands of suppliers on one single platform, hence resulting in improved efficiency, facilitating control and rapid response time to both customers and management [3]–[5]. This is particularly true in case of global tourism industry since it heavily deploys CRSs to process their reservations through GDSs to perform basic functions of reservation process, such as product presentation, reservation, fare quote & ticketing and additional services [1]. This is reflective in tourism or hospitality industry where over the years, electronic reservation systems have provided greater operational benefits in terms of yield management, e-marketing strategies as well as productivity benefits [64]. Likewise travel agencies through GDSs enjoy the freedom to make reservations directly from their terminal with any airline, on any continent. This saves much of their coordination time and effort that is required in settling negotiations.



Figure 2.6: Distribution Channels for Airline Reservation Systems

As mentioned earlier, GDSs has brought hundreds of thousands of travel agents and distributors from different countries and continents in nexus, it has thus acquired the status of the nervous system in global reservation system. The GDSs generated more than \$9.6 billion in revenue and more than 1.1 billion transactions in 2008, just over 2,100 transactions per minute. The infrastructure of each GDS can support volumes far greater than this [9] since airlines and other distributors that are in nexus with GDSs and provide access to schedule and fare to travel agencies – both offline and online [9].

2.1.2.2 Self-Booking Tools vs. Online Travel Agencies

A travel agency, also called a travel bureau, is defined as business that attends to the details of transportation, itinerary, and accommodations for travelers [2]. Travel agency acts as an agent, just like a retail storefront that books and sells tickets on behalf of many airlines. Traditional travel agencies hold a large portion of travel booking industry, due to a number of factors such as, face to face service to customers, provision of personalized services and realistic solutions for providing reservation arrangements, comfort in country of destination, and special packages or promotional deals. On the negative front, traditional travel agencies are blamed for practicing restrictive practices such as racking whereby they promote traveling brochures of those companies only who pay them highest commission. The traveler is unaware of possible alternative options and considers them to be the only once or best option available.

On the contrary, an Online Travel Agencies (OTAs) operate through a travel website on the World Wide Web, dedicated to providing updated travel related information, guidance and travel reviews [2]. The travelers interact with the virtual interface of the online travel agency which allows them to search and book their travel plans. The online reservation process does not involve personalized attention on behalf of online travel agency and still this does not seem to be a matter of concern for travelers. According to Forrester research [33], approximately seventy million consumers searched for travel plans online in July 2006, thus making online travel bookings the single largest component of e-commerce. This also makes online travel agencies an important part of the overall equation for flexible reservation systems for airlines. The recent growing acceptance of online travel agencies is credited to their

meta-search engines feature that provides fare aggregators to travelers. Meta-search engine as the name indicates conducts search across multiple independent search engines and gets live availability of flights through "screen scraping" process, which crawls through the airline websites and extracts content by way of human-readable HTML feed. The content extracted from various airlines website is then displayed to the users in the form of fare aggregation, i.e. all results on one screen. According to PhoCusWright Report 2009 [9], the overall share of online travel agencies in US travel market alone was 13% in 2008 and projected to touch 16% in 2011 as shown in Figure 2.7. On the contrary, the share of conventional traveling agency was 33% in 2008 and is projected to suffer a decline 3% in 2011. This is further justified from Yahoo Travels' claim which says that 76% of all online travel purchases occur as a result of search function. Jupiter Research in its Travel Consumer Survey published in 2004 pointed out that "nearly two in five online travel consumers say they believe that no one site has the lowest rates or fares." This therefore created a niche research dimension for OTAs to look at different ways for integrating additional features into their reservation system so as to optimize aggregate travel search and provide lowest rates from multiple travel sites, to eliminate travelers' verification need from site to site.

It is also interesting to note from the chart as shown in Figure 2.7 that supplier branded websites also will experience an estimated increase of 3% in 2011. Suppliers branded websites are Self-Booking Tools (SBTs) providing direct linkage of the passenger with the airline industry. They provide carrier-direct bookings facility to travelers, without having them going through the hassle of other intermediaries. These booking are just like going to the reservation office of a specific airline physically and are popular among travelers who remain loyal to their favourite brands of airlines and prefer to travel only through them. Another strong reason for travelers to opt for SBTs is their ability to earn flying rewards, which ultimately makes them more loyal towards a particular brand of airline. Likewise, in order to differentiate their reservation channel from others, airlines have started to invest heavily in their online SBTs capabilities, offering more features and convenience for travelers such as tracking, managing and redeeming air miles. Moreover, airlines also invest massively on branding of their image and securing loyalty of customers by offering reward mileage bonuses. In addition to this some airlines have gone to the extent of imposing fees on GDS bookings for their carrier. For example Lufthansa airline in its Preferred Fares Program launched in 2008, imposed fees of €4.90 per ticket for travel agencies in Austria, Germany and Switzerland that made reservations through GDSs [6].



Figure 2.7: U.S. Travel Market by Channel, 2008 and 2011 (Projected) Source: PhoCusWright U.S. Online Travel Overview Ninth Edition

On the contrary, OTAs cash upon the nexus of CRSs and GDSs and act as a central hub for price differentiation and comparison. They provide discounted fares and 24 hours service. Their fixed costs are lowest, since there is no requirement as such to set up physical offices with state-of-the-art facilities at prime locations. Their success is derived by innovational strategies, as a result they hold lion's share, 50% (average 2006, 2007 & 2008) in the travel industry.

A comparison chart on SBTs and OTAs is presented and discussed as shown in Table 2.2. The table presents innovative attributes and function that have contributed immensely towards the popular acceptance of OTAs over Airlines' SBTs and are also widely common among travel companies in recent years [15], [16]. Functions such as product presentation, reservation, quoting & ticketing, post-sale services, low fare notification, dynamic packaging and flexible alternative date search are also performed by SBTs, however, OTAs get an edge over SBTs in terms of providing matrix display, opaque fares, alternative airport search and hotel search.

Feature	Description	SBTs	OTAs
Product presentation	is the presentation of services or products in all aspects of travel industry.	~	\checkmark
Reservations	is used for making reservations for the offered services and products.	~	\checkmark
Quoting & ticketing	relates to providing fare quotes and generate receipts for the given services and products.	\checkmark	\checkmark
Additional services	post-sale features and user prompting for their guidance throughout the reservation process.	~	\checkmark
Matrix display	using this feature, users may click on any cell within the matrix to sort airfare search results by price, airline and number of stops. It was initially introduced by Orbitz but these days it has become a standard for all OTAs.		~
Alternative airport search	allow travelers to search across multiple departure and arrival airports so as to find the lowest possible fare or most convenient schedule.		~
Hotel search, results display & sorting	 this feature allow travelers to display, sort and compare options from hundreds of possible hotel search results. address or landmark search and sorting map-based search results display traveler reviews included with the results multiple sorting options, including price, star rating, brand, guest rating and amenities 		~
Opaque fares	were initiated by Priceline's Name Your Own Price airfare bidding model. In this feature users are offered heavily discounted tickets with not specified time or flight number. They are usable at the discretion of the airline.		~
Low-fare notifications	a feature in which customers are intimated via email to opt for specific promotional deals.	\checkmark	\checkmark
Flexible & alternative date search	allow users to search and compare flight options across multiple departures and return dates so as to find the lowest possible fare.	1	√
Dynamic packaging	initially made famous by Expedia where users are allowed to shop for multiple components in a single search, such as "Flight and Hotel".	~	~

Table 2.2: Comparison Chart on Offered Features by SBTs and OTAs [1], [9]

Matrix Display – Orbitz is the pioneer in OTAs who initiated the concept of matrix display. This feature allows users to click on any particular airline offered fare to see the further details such as departure and arrival timings. As shown in Figure 2.8, there are number of airfares from New York to Los Angles offered by multiple carriers on the specified dates e.g., departure: 29 December 2010; arrival: 29 January

2011. In the matrix, carriers are organized in multiple columns, stops in multiple rows and the airfares are placed against airlines and stops.

lick Search Vacation	Packages Ho	tels 1.01	Cars	Cruises	Activities	Deals		~	~			
	100 - 11	11-	From: New Y To: Los Ange Click a fare	ork (41 Local les (AJ Local to see airli	ons), NY (N) lons), CA (L4 nes and tim	с ю еs. Ф	E	Search paramete				
NO FLIGHT BOOKING FEES	Find flights by: Airline Stops Price	Multiple Carriers	A ^k A American Airlines	Virgin America	US Airways	jetBlue JetBlue Alrwava	a Airlitati Airwaya	United Airlines	Deita Air Lines	Alaska Alaska Artines	Continental Airlines	Erontie Airline
UNBEATABLE VALUE	Non-stop	\$477 total <u>3595</u>	\$527 total <u>\$545</u>	\$665 totel <u>5838</u>	\$884 tatel <u>\$905</u>	\$655 total <u>5878</u>		\$711 total <u>\$733</u>	\$724 total <u>5745</u>	\$761 total 3783	\$767 total 5785	1.12
O Learn more	<u>1 siop</u>	\$467 total 3505	\$527 total 3559	\$530 total <u>5552</u>	\$521 Iotal <u>\$563</u>		\$656 lotal <u>\$591</u>		\$724 total <u>\$755</u>		\$1,112 total <u>\$1.143</u>	\$838 total <u>323</u>
	2+ stops			\$530								

Figure 2.8: Lowest Airfare Search Results through Matrix Display [66]

Alternate Airport Search – Alternate airport search feature helps users to find the lowest possible fare or most convenient schedule across multiple departure and arrival airports. Using this feature as shown in Figure 2.9, one can click on the checkbox "include nearby airports" to see the search results on the specified as well as nearby airports. Orbitz provides the flexibility of choosing "include nearby airports" for source and destination airports.

	Lowest price	Departure time	Shortest flight		Airpart codes
Change Search	Showing lowest	priced flights (45 figh	a out of 358 Iolali	See at 218 metre	Sponsored Links
From City tame or 20000 NYC O Incl. nearby airports LAX incl. nearby airports Leave 12/25/10 Depart T	Select	\$530 - 532 taxes & fe Alternate Airpo Search Option Change planes. Time to Vir cart:1:40pm Sat wei:3:00pm Lat	es = \$562 USD per person NO, BOOKING(FEES)	PRICE ASSURANCE	Malaysia Airlines Creac Pares to Many Destinations For Limitad Ferico Only, Book Nowl www.or. Malaysia.Ardines.com Fiv: to L.A. from \$49" Los Angeles Rates Just Drocced Book Naviso Loo In the Best Deals LowFares.com/Los-Angeles Filahts Airline tickets los angeles Permanent for close KLM dises Breas deals now at KLM com
Anytime 💌		<u>р</u> т.,	conory 1br 25min Arbus A320 <u>View seata</u>		AirAsia Cheap Fares

Figure 2.9: Alternative Airport Search to Find the Lowest Possible Fare [66]

Hotel Search – OTAs offer hotel search feature that provides the option to display, sort and compare hundreds of possible hotel search results. As shown in Figure 2.10, this feature allows address or landmark search and sorting, map-based

search results display, traveler reviews included with the results and multiple sorting options, including lowest price, distance, star rating, brand and amenities.



Figure 2.10: Hotel Search, Results Display & Sorting to Compare Options from Hundreds of Possible Hotels [66]

Opaque Fares – Opaque selling intermediaries have become an established distribution channel for the travel industry [17]. This form of opaque selling came into limelight in 1998 when priceline.com's, *Name-Your-Own-Price* emerged with an opaque selling business model where both the itinerary information and the identity of the airline carrier were hidden from the traveler, until the bid was purchased as shown in Figure 2.11. Next, major U.S airlines established Hotwire to compete in the opaque segment of intermediaries. However, it came up with a different opaque selling business model, which was not based on bidding mechanism but rather posting a price for an offer that concealed key itinerary information and airline identity as shown in Figure 2.12 and 2.13. Since then many opaque selling intermediaries have appeared in the international travel market, such as cheaptickets.com, onetravel.com etc. and popularized this selling mechanism [67], [68]. However, the common aspect of all such opaque selling intermediaries is that they are based on hiding descriptive attributes of the service to be provided; as a result travelers cannot fully predict the ultimate service provider or the airline.

priceline.com[.]

Name Your Own Price[®] and Save up to 40% on Flights¹ New York City. NY to Los Angeles. CA. Wed, Dec 29 to Sat, Jan 29 - 1 ticket change cities

Select Departure and Arrival Airports

Please note that Name Your Own Price flights are economy (coach) class only

Departure Airports		Arrival Air	ports			
Y	CHOOSE THESE AIRPI	ORTS FOR THE	BEST DEAL V			
Newark Intl. NJ (EWR)		🗐 Los Ar	ngeles Intl. CA (LAX)			
New York City - John F Kenn New York City - La Guardia. Newburgh - Stewart Intl. NY (White Plains - Westchester (Islip - Long Island MacArthur	edy Intl. NY (JFK) NY (LGA) SWF) County, NY (HPN) NY (ISP)	 Ontario Intl. CA (ONT) Burbank Glendale Pasadena. CA (BUR) Long Beach - Daugherty Field. CA (LGB) Santa Ana - John Wayne / Orange County. CA (
ame Your Own Price						
Name Your Own Price (Per Rou	od-trip Ticket) \$.00	BEST & QUALITY			
otal charges, including taxes and fees	, are displayed on the	following page				
riease enter passenger name(s) exa ige of 18 must be accompanied by a f you already have a priceline profile	ictly as they appear o in adult Note: Meal a and would like to acc	n a driver's lici ind seat prefer ess your store	ense or other official photo ences are not guaranteed id flight preferences, click i	here to sign in		
Adult Passenger 1: Under 1	8? 🗖					
First Name:	Middle Name:		Last Name:	Suffix		
Please provide your TSA Secure Fl	light information now.	Learn more.				
Date of Birth	Gender					
Month 👻 Day 👻 Year 👻	🔿 Male 🔘	Female				
Infants: Number of unticketed passengers u	nder age 2. 0 👻					
	prov	next				

Figure 2.11: Opaque Fares Offered by Priceline for Discounted Tickets without

Specifying Carrier, Time and the Route [69]

Home Hotels	Cars	Flights	Vacations	Cruises	Deals					Help Ce	nter	Planning Too
Velcome to F Find the lowest fares in a	lexi 30-day ra	ble Dat	te Sea	urch								
1. Location From See a NEW YORK CITY, NEW YORK	K K	To [LOS ANGELE	S. CALIFORN	si oden ĝi G	2. Travel date Trip length 14-15 nights 2-4 nights Weekend 3-5 nights 4-6 nights 5-7 nights 6-8 nights 8-10 nights	5	raveling so (2/29/10 earch us te	ameti an 30 de Have	ime be d lot. e F	tween 27/11 e on bertain da	3. Pi	ind a flight O
You enter your deaired trip length Date Search finds all the best far	and a trav	ei window up to 1 1 those dates with	10 days, Flexicie 1 one click, You	get -	12-14 nights 12-14 nights 10-12 nights 12-14 nights 12-14 nights	10 147 10 506	300 Mg 19 101	10	122	148 148 14 15 16 218 26 25	a mi v	100 340 545
Savings Put more money back in your poo	ket for you	r snip			San a	n 140	wight trave Werd The Her 12	Feb	5.0 25	Forg Lowest New — No Typic accelet		The Game Ration proj
Choice we search every roundtrip flight : Convenience	n your trav	ei window. You e	see the lowest f	ares.	\$99*	CALL N N	Desert das de la		38 3/6 100 4			
How many searches would you t saves you the trouble.	so la fina th	e cest fare? Flexi	ble Dale Search		1179 1 16 1504 3	n tes ta M	M7 342	274-	108			

Figure 2.12: Opaque Fares Offered by Hotwire for Discounted Tickets with Specified Time or Flight Number [70]



Figure 2.13: Flexible Date Search Results on the of 14-16 Nights' Trip Length [70]

In this section four OTA features have been identified and discussed that are not integrated into SBTs of the airlines mostly because of the practical implication of each OTA feature, which is not very feasible for the SBTs to opt for. However, it is essential in this research to understand if the integration of same OTAs features could make SBTs flexible OARS, and if the answer was yes then how come that could be done. Therefore, preliminary but comprehensive researches need to be conducted with airline executives to examine their subjective satisfaction with existing SBTs by the airlines.

Furthermore, it is also important to explore the recommendations for making SBTs flexible Online Airline Reservation Systems in order to increase the usability of online reservation systems. Further elaboration of the preliminary research can be found in the methodology chapter Section 3.2.3.

2.2 Usability of Online Airline Reservation Systems

Researchers argued that, the design of a website determines online purchase decisions and revisit intentions [71]–[73]. The essence of quality for a successful website has been addressed by many authors time-to-time [74]–[78]. The quality of a website is referred to its usability and functionality [24], [79]. A website will be considered as useable if users can accomplish their tasks easily. Similarly, a website will be considered as functional if it offers all the functions required by a user to perform their tasks [24]. Thus quality of a website can be evaluated on the basis of different functions they offer and the performance of those functions. Therefore, it is very critical for the organizations to design a website that is easy to use and accomplish all the requirements.

In the last twenty years different lines of research have focused on identifying certain factors influencing acceptance of information systems and have provided models and theoretical proposals. Social Cognitive Theory [80], Diffusion of Innovation Theory [81], the Theory of Reasoned Action (TRA) / Theory of Planned Behavior (TPB) [82]–[84], the Triandis Model [85], Human Computer Interaction research [86], the Technology Transition Model [87], and Social Network Theory [88] are representative examples.

In particular, Technology Acceptance Model (TAM), as introduced by Davis [79] where external variables have been identified as factors that tend to influence upon systems perceived ease of use and perceived usefulness. In other words, the model explains how users accept and use technology on the basis of perceived usefulness and perceived ease-of-use of the system influenced by external factors. The contextual interpretation of any event is determined by contextual factors that reinforce viewers' schemas, formulate characteristics of the surrounding environment and ensure

effective collaboration between the two. In case of news processing, for example, the contextual factors that reinforce viewers' schemas are their lifestyle, political socialization, prior knowledge and life experiences, current needs for various types of information, and attitudinal factors such as interest in news and perceived credibility of sources [89]. During the past two decades, TAM is considered the most prominent, powerful and parsimonious theory for describing an individual's acceptance of information systems [90]–[93]. Many other researchers have also contributed to the list of external variables [94]–[96] since original TAM model has more than seven hundred citations to its credit and has been adapted and extended in many ways to-date.

TAM and TPB, have received considerable attention from the scientific community and its use has been extended to the study of tourism services [63]. However, TAM and TPB have successfully explained behavioral intentions, previous research pointed out that TAM and TPB's fundamental constructs do not reflect the specific influences of usage-context factors that may alter users acceptance. Usage-context factors are based on users' contextual interpretations that are based on their attitude or belief. In case of Online Airline Reservation Systems, it is suggested that TAM and TPB should be considered with more belief-related variables. Therefore, in our framework, self-determined psychological factors that may influence upon perceived flexible personality of travelers by way of reinforcing their purchase decisions are adapted from the TAM.

2.2.1 Usability Perception

Customers' acceptance of the internet, as a suitable medium for booking their itineraries, has been accelerated due to the structural changes in the aviation industry [97]. Likewise, research shows users dissatisfaction with existing Online Airline Reservation Systems in terms of task completion [26]. Therefore, researchers are eager to find out ways for improving online usability of the systems, how users' Perceived Usability of the system is formulated by its flexibility functions [27]–[31]. In addition to techniques, methods and guidelines proposed for designing usable systems, HCI researchers have also long argued on the importance of human factors

in designing and implementation of user-centered designs. According to Nielson [32], "users experience usability of a site before they have committed to using it and before they have spent any money on potential purchases". This indicates users' Perceived Usability in online digital environments is an important determinant for evaluating their satisfaction in the same environment. The existing literature can be divided into four research aspects where researchers currently are focusing upon to determine Perceived Usability of online systems.

2.2.1.1 Usability Perception by Performing Content Analysis

Many worldwide researchers in tourism industry have examined Perceived Usability by performing content analysis of websites' features [98]–[103] which involves technical assessment of the basic content and hypertext structure of the websites. For example, Morrison *et al.* [104] provided a comprehensive study of different websites evaluation approaches by categorizing them into four groups. The grouping was based upon determining "effectiveness and efficiency" and "Why and When" in evaluation of the websites. Similarly, Law and Leung [101] examined 30 different North American Online Airline Reservation Systems for evaluating the range of comprehensive online reservation services provided by each of them. Their research provides useful sets of attributes for consideration. Their findings which were based on those useful sets of attributes showed significant differences user's satisfaction with each website. Furthermore, Scharl *et al.* [105] changed the traditional evaluation techniques done by human experts and introduced an automated tool for the systematically evaluation of websites.

2.2.1.2 Usability Perception through User's Internet Adoption

The second area of research is based upon examining Perceived Usability of online system through user's internet adoption practices [97]. One major pitfall of this kind of research is that it primarily focuses upon use of internet technologies instead of evaluating websites for determining Usability Perception of users. The investigation

methodologies typically involve interviews or surveys of tour operators, hotels and destination marketing organizations [106], [107].

2.2.1.3 Usability Perception based on Users' Preferences and Expectations

The third area of research is based upon investigating web users' characteristics, preferences, and expectations [108] from online systems, and to compare with the Perceived Usability of the systems. For example, researchers have investigated the demographic differences between the "lookers" and "bookers" [109], [110] in online reservation systems and how their Perceived Usability different from one another due to the contextual factors involved. Then there are a few academic studies where researchers have investigated customer preferences and expectations for tourism websites [97]. However, no study has been undertaken to examine customer preferences and expectations from online airline reservation websites [97].

2.2.1.4 Usability Perception based on Online Behavior of Web Users

The fourth most important and sophisticated area of research is the investigation of online behavior of web users [97] and how their behavior is related to determining perceived usefulness of the system. Some researchers have paid attention to online users' search behavior on information [111], [112]. Bai *et al.* [113] studied online travel behavior of US college students. Rudstorm and Fagerberg [114] adopted quasi-experimental methodology to investigate customer's behavior and to explore emerging concept of socially enhanced travel booking. In addition to this, Klein *et al.* [115] and Marcussen [116] have examined online behavior of European travelers that have possibly led to slower adoption trend of online air travel bookings in Europe. Their findings showed that lack of relevant information, price dispersion, product complexity, and the usability of online booking tools were the rationale behind customers' such behavior.

2.2.2 Conceptual Linking between PU and FTB and FOARS

To discover the relationship between Perceived Usability (PU), Flexible Traveling Behavior (FTB) and flexible Online Airline Reservation Systems (FOARS), it is necessary to study usability in the context of flexibility from two different perspectives, i.e. from user's traveling behavior and flexibility of the reservation systems.

Users traveling behavior is molded by a number of important personality relevant determinants, both internal and external in characteristics. While, traveling consciousness, self-efficacy in digital skills and self-belief as flexible travelers are internal personality relevant determinants influencing directly upon travelers flexible behavior, societal influences, attribution and prior experiences are external personality relevant determinants that indirectly influence upon travelers flexible behavior. Moreover, external determinant may not necessarily always have the same influence every time, depending upon the situation the traveler is in.

As for System's Flexibility is concerned, system's perceived flexibility is reflected in its Perceived Usability. Perceived Usability is a combination of system's effectiveness, efficiency and satisfaction. End user support or user prompting is considered to be a supporting characteristic that substantially augments efficiency and effectiveness of the system, while multiple options directly influence upon user satisfaction. If a system provides ranges of dates as flying and source destination options at different fares, flexibility of the system is enhanced in its Perceived Usability in the eyes of the users. This is because if a user chooses a flying option 'A' from a given one or two options, he has not made a flexible decision. But if he chooses the same flying option 'A' from a variety of given flying options, he is likely to enjoy extra satisfaction that he will get from the flexibility of the system and also in his purchase making decision. Likewise, if a system offers multiple flying options, they will also influence upon users' decision and make them change their mind to opt to fly from option 'B' instead of 'A'. This will again have positive influence upon user's satisfaction from the system - Perceived Usability. More details on this study can be found in chapter 3 Section 3.3.1.

35

2.3 Usability Vs. Flexibility

Researchers point out [117] that opaque products are flexible in characteristics; therefore, a seller is in a unique position to offer horizontally differentiated products to customers upon purchase due to the flexibility of assigning pre-determined products to the customer. Opaque selling became popular since they offered a very unique price discrimination mechanism [118] and could generate incremental revenue for the airline by deliberating upon price sensitive consumers [119]. In very short time, opaque selling has attained the status of a competitive lever for the airline, signifying that an airline could suffer revenue loss to its competitors by not opting to offer opaque offers [120].

The concept of online airlines reservation systems or SBTs for disposing off their opaque inventory directly has not been adequately considered in information system research and literature. As highlighted by Jerath *et al.* [10] a number of studies have carried out with primary focus upon airline revenue management systems, however attempts to empirically verify those findings are a few. However, some recent research by Jiang [118], Fay [121], Granados et al [122], Jerath *et al.* [10] have made strong argument in favour of this connotation and given theoretical justifications in support of the argument. Likewise, recent research on opaque selling has rather adopted an objective discourse to empirically validate revenue management theories (see, for example, Puller *et al.* [123]). Granados *et al.* [122], [124] has compared price elasticities of the offline, online transparent and opaque channels. Their findings suggest that opaque selling mechanism has high price elasticity. Again this does not adequately address the research gap on opaque selling through SBTs.

It is believed offering opaque selling through SBTs will not reduce profits, as in the case of direct last-minute selling undertaken by an airline, because it holds a critical position to replicate its profit margin by setting high price of opaque selling. But on a more fare note, airlines may like to introduce opaque selling directly through their SBTs so as to attract additional sufficient number of customers and minimize the effect of price degradation, as discussed earlier. Thus it is believed that if an airline opts for opaque selling through its SBTs, it can enhance its profit through market expansion and also by enhancing price discrimination of one's existing customer base [121]. Moreover, in order to test applicability of opaque selling, empirical research on opaque selling intermediaries show that it depends upon certain factors, e.g. demand characteristics (Jerath et al. [10], Fay and Xie [18], Granados et al. [122]), product characteristics [118], consumer loyalty [121], industry structure [120] and competition [10], [125]. However, this research adopts an approach based on behavioral characteristics. This is because literature review has highlighted that direct last-minute selling, although could lead to severe consequences for the airline, is still preferred over opaque selling through an opaque intermediary because of high expectations of customers on little service differentiation [10]. And research on opaque selling shows it has been preferred over direct last minute selling with an increase in high demand situations and this has been the primary factor for airlines to opt for opaque selling intermediaries [10]. However, if an airline opts for an opaque fare selling it can position itself more competitively than opaque selling intermediaries because of its knowledge and accurate resource information and management [10]. Thus if an airline opts to adopt an opaque selling mechanism, it will be in a win-win situation, whereby it will employ its own resources and provide opaque selling directly to price-sensitive customers, who do not wish to anticipate hidden characteristics of the service to be provided to them, a major concern in opaque selling through intermediaries. Not only this, it will also add brand customer loyalty to its credit as well.

The proposed model in this thesis thus integrates opaque selling mechanics into the framework for designing of flexible Online Airline Reservation Systems. Thus, it is important to discuss mechanics of opaque selling from same two perspectives discussed earlier, System's Flexibility and User's Flexibility.

2.3.1 Opaque Selling Mechanics in System's Flexibility

In order to determine the role of opaque fare in making airlines SBTs (system) flexible and improve their usability it is first and foremost important to discuss flexibility of SBTs.

Earlier B2C systems were considered to be not very flexible as mentioned by Olsen & Malizia [20], [21] in terms of handling more open requests that could not be possibly mapped directly into the formalized terms offered by the Web-interface. They would work for simple closed requests, i.e., a request that could be mapped directly into formalized terms or pre-defined parameters such as dates, airports, flights, etc. The system could break down for more complex closed requests, i.e., where the customer is flexible with regards to attributes such as destination and dates [20], [21]. Olsen & Malizia has recommended an information system, as an intermediate between a customer and booking system, that would provide the user with all the necessary data and support, on a mere button-click, after the initial data has been fed into the system, see Figure 2.14.



Figure 2.14: Ticket Reservation through Intermediate **Source:** Flexible User Interfaces for B2C Systems [20]

As per the above diagram, the role of intermediate system which is an SBT in this case is to provide detail information to users, which could be useful in making good decisions [20], [21]. Therefore, SBTs being intermediate information systems serve the purpose of human agents in order to map formalized terms into closed requests. And the existing SBTs cover mostly pre-sales flexibility [126]. The term "flexibility" here should not only be related with the booking of a ticket in case of online booking system. If we say, our booking/reservation system is flexible then ideally speaking, it should also support flexibility with regards to flexible features of the system. Since the essence of a successful quality website has been addressed by many authors from time-to-time [74]–[78] and the quality of a website is referred to its usability and functionality [24], [76]. Likewise, in case of usability of SBTs, users have reported to be unsatisfied when they are flexible with regards to traveling attributes such as destination and dates [20], [21], [26] and systems are not. That is why many researchers are trying to find out ways so as to improve online usability [26]–[30] and

HCI researchers have long argued on the importance and criticality of human factors' study to the successful design and implementation of technological devices so as to improve system's usability by enhancing its flexibility.

In context of integrating opaque mechanics into existing SBTs of airlines, we know from literature that when comparing last-minute direct sales of an airline, with those offered through an opaque selling intermediary, researchers have found out that opaque channel increases total demand [17] since customers are contended to comprise on hidden attributes in anticipation of heavy discounts. This anticipation of travelers for low fares is an extremely important concern, that airlines are faced with every day. As mentioned earlier, airlines are not in a strategic position to offer low cost fares directly to dispose-off their distressed inventories, mainly due to potential threat to their revenue generation, therefore, the only business model that successfully addresses this concern is opaque selling through intermediaries or OTAs. However by opting for OTAs, such as Priceline for example, the traveling attributes such as the airline or the route to be flown are hidden and research shows frustration experienced by travelers when they end up flying a much more circuitous route, than they might have wished and also not necessarily with the most preferred carrier. Moreover, even if an airline sells its distressed inventory through opaque selling intermediaries, it does not add brand loyalty of travelers to its credit, as travelers are likely to remain impartial to the carrier they fly with under opaque selling [79]. Likewise, research shows that more airline products become opaque or hidden in nature, higher the dissatisfaction of the traveler becomes with the quality of airline solution or service [79]. Therefore, it is argued that opaque selling mechanics if adopted by existing SBTs, they will not only become flexible reservation systems and could even bring similar or higher incremental revenue for the airline.

2.3.2 Opaque Selling Mechanics in User's Flexibility

Anticipation of travelers for low fares not only indicates why opaque selling intermediaries became so popularly accepted, but also an idea about traveling behavior of customers, which is 'the extent to which a traveler anticipates for a low fare', indicating the extent to which 'a traveler is ready to compromise on flying

conditions', and thus becoming flexible in accepting what is being offered to them by an airline. The idea of flexibility is not something new in airline industry, it was proposed by Schwieterman [127] who had emphasized upon segmentation of the market between discretionary and non-discretionary travelers using time flexibility. More, recent research by Garrow [128], and Carroll *et al.* [129] also deploy time flexibility as a value driver. However, there appears to be no research done to date, in which destination flexibility is used as a value driver [79]. By designing SBTs in view of User's Flexibility, they will be provided additional detailed information and choices unquestionably useful in making good decision [20], [21].

2.4 Chapter Summary

As said earlier, this research is undertaken to examine applicability of opaque selling mechanics on SBTs and have discussed the same in context of System's Flexibility and User's Flexibility to increase the Perceived Usability of Online Airline Reservation Systems. The first two research areas of Usability Perception i.e. Usability Perception by Performing Content Analysis and Usability Perception through User's Internet Adoption are related to supply oriented studies, the other two studies i.e. Usability Perception based on Users' Preferences and Expectations and Usability Perception based on Online Behavior of Web Users are demand driven since they consider online services and features used by travelers when making traveling and purchasing decisions. There are also some researchers who have combined some or all four research areas for investigating Usability Perception of online systems. For example, Benckendorff and Black [99] have used surveys of regional tourism organizations as well as website evaluations methods. Nysveen [108] conducted surveys and obtained results from both web users and tourism businesses. Their research objective was to investigate gaps between customer preferences and actual website offerings.

This research uses a blended approached and combines the four research areas of Usability Perception to determine Perceived Usability of the Online Airline Reservation Systems. It is important because customers' usability expectation and preferences from Online Airline Reservation Systems lacks research and empirical findings. Law and Leung [101] had emphasized upon the need to investigate expectations of airline customers that book their itineraries through their online self-booking tools. Moreover, the existing evaluation of online tourism websites is performed by researchers and not by customers. It leads to a dilemma and research gap that does not potentially address expectations of travelers. Even though internet is referred as a major technological innovation of today, its success heavily depends upon assimilation of customer expectations and preferences into the design and content of websites [101]. For airlines to run successful business, in spite of their products are sold online or through more traditional channels, it is important to design their self-booking tools in view of customers' preferences, expectations and online usage behavior.

Thus, behavioral characteristics in terms of making travelers flexible are appealing and an important area of research. And if it can be determined, i.e., what makes a traveler flexible on the basis of his/her behavioral characteristics. it could give crucial insight for designing of Flexible Online Airline Reservation Systems, based on SBTs offering opaque selling. The existing opaque selling literature lies at the intersection of consumer behavior and revenue management operational strategies [40]. However no study has been found to address consumer behavior on opaque selling with respect to online airline reservation system as most recent paper and researchers are done within the marketing domain [40]. The research therefore examines travelers' expectations, preferences and online behavior (User's Flexibility) and aligns that with designing of flexible Online Airline Reservation Systems (Systems Flexibility) and users' as evaluators of the online systems to determine its usefulness through effectiveness, efficiency and satisfaction (Perceived Usability).

This anticipation of travelers for low fares is an extremely important concern, that airlines are faced with every day. As mentioned earlier, airlines are not in a strategic position to offer low cost fares directly to sell their left over inventories, due a number of potential threats to their revenue generation, therefore, the only business model that addresses this concern is opaque selling by OTAs. The concept of online airlines reservation systems or SBTs for disposing off their opaque inventory directly has not been adequately considered in information system research and literature. The concept requires extensive research especially in academic discipline [40] and as highlighted by Jerath *et al.* [10] a number of studies have focused upon airline revenue management systems, however attempts to empirically verify those findings are a few. Likewise, recent research adopts an objective discourse to empirically validate revenue management theories (see, for example, Puller *et al.* [123]). This does not adequately address the research gap on opaque selling. Moreover, the only known studies on opaque selling are of Granados *et al.* [122], [124] who have compared price elasticities of the offline, online transparent and opaque channels. Their findings suggest that opaque selling mechanism has high price elasticity.

Literature also highlights that direct last-minute selling, although could lead to severe consequences for the airline, it is still preferred over selling through an opaque intermediary in case of high expectations of customers and customer expecting little service differentiation [10]. However, research on opaque selling also shows it has been preferred over direct last minute selling with an increase in high demand situations and this has been the primary factor for airlines to opt for opaque fare intermediaries [10]. However, if an airline opts for an opaque fare selling it can position itself more competitively than opaque selling intermediaries because of its knowledge and precise information on availability of its own resources [10]. This could put an airline in a win-win situation, whereby it will employ its own resources and provide opaque selling directly to price-sensitive customers, who do not wish to anticipate hidden characteristics of the service to be provided to them, a major concern in opaque selling through intermediaries.

Anticipation of travelers for low fares, gives an idea about their traveling behavior, which is 'the extent to which a traveler anticipates for a low fare', indicating the extent to which 'a traveler is ready to compromise on flying conditions', and thus becoming flexible in accepting what is being offered to them by an airline. The existing opaque selling literature lies at the intersection of consumer behavior and revenue management operational strategies [40]. However no study has been found to address consumer behavior on opaque selling with respect to online airline reservation system as most recent paper and researchers are done within the marketing domain [40]. This research therefore contributes to the small but growing literature in designing of Online Airline Reservation Systems by moulding upon flexible behavior of travelers.

CHAPTER 3

METHODOLOGY

3.0 Chapter Overview

In this chapter, the overall methodology of the thesis is described which is divided into three phases. Each phase contains one core research objective which is achieved through the corresponding research questions and hypotheses. Section 3.1 is devoted to the organization of the phases named as research methodology. Section 3.2 covers Phase I that consists of 3 studies to investigate and discuss the user needs associated to System's Flexibility and Users' Flexibility. Section 3.3 is dedicated to Phase II which contains 2 further studies to classify users and to investigate the interrelationship of the variables. At the end of Phase II, an overview of the study results are also presented in order to facilitate the description of the methods used in this research work. Section 3.4 presents Phase III which contains a case study in order to test the proposed framework. Section 3.5 presents the statistical formulas that are used in this research and Section 3.6 summarizes the chapter.

3.1 Research Methodology

The methodology of this thesis consists of three phases which are described below. The overall research methodology and a complete list of research questions and the corresponding hypothesis are shown in Figure 3.1 and Table 3.1 respectively.

Phase I: Assessing User Needs (System's Flexibility & Users' Flexibility)

Phase II: User's Classifications (Interrelationship Testing of Variables)

Phase III: Case Study (Testing the Model)



Figure 3.1: Research Methodology

Research Phase Study **Research Questions Hypotheses** Analysis **Objectives** I To assess user **RO1:** What are the issues H₁: Non-functional Requirements are 1 needs (System's with flexibility of Online perceived to have an impact on the Flexibility and Airline Reservation Systems, usability of OARS. Users' Flexibility) whether or not flexibility is H₂: Functional Requirements are associated with one of the reasons for users perceived to have an impact on the Correlation Online Airline not using such systems? usability of OARS. Analysis, Reservation H₃: The perceived flexibility of OARS Reliability Systems. affects the usability of such systems. Analysis, H₄: Functional Requirements of OARS Descriptive Analysis. are inversely associated with the flexibility of the systems. H₅: The availability of resources and skills set influence upon the usability of OARS. **RO2:** To what extend flexible H₆: The level of satisfaction 2 with SBTs different users can compromise with existing is for service quality attributes of respondents with different attitudes Online Airline Reservation Users' towards Flexibility in ANOVA & Post-Systems? compromising on SQAs of the airline. H₇: The level of satisfaction with Hoc existing OTAs is different for respondents with different attitudes towards Users' Flexibility in compromising on SQAs of the airline.

Table 3.1: Research Questions and Hypotheses

Phase	Study	Research Objectives	Research Questions	Hypotheses	Analysis
	3		RQ3: How users' satisfaction with an existing SBTs is rated against their choice of OTA feature and reflected in their integration assessment of the same for making SBTs more flexible Online Airline Reservation Systems?	H ₈ : Users' satisfaction with existing SBTs is different across their choice of four OTA features for making SBTs more FOARS.	Two-Way ANOVA
II	4	To propose a framework for designing more flexible Online	RQ4: How users' perception on factors influencing flexible traveling behavior and flexible OARS is determined?		Emerging Theme Analysis
-	5	Airline Reservation Systems while classifying users on the basis of their Flexible Traveling Behavior.	RQ5: How to classify Users' on the basis of their Flexible Traveling Behavior into High, Medium and Low flexible and how to investigate interrelationships among	H ₉ : Users can be classified on the basis of their Flexible Traveling Behavior.	Unidirectional Scale
	the basis of their Flexible Traveling Behavior. Medium and Low fl how to interrelationships System's Flexibilit Flexibility and Usability of existin			H ₁₀ : User's Flexible Behavior and their Perceived Usability is correlated.	Correlation Analysis
		System's Flexibility, Users' Flexibility and Perceived Usability of existing Online	H ₁₁ : User's Flexible Behavior and System's Flexibility is correlated.	Correlation Analysis	
		Airline R	Airline Reservation Systems?	H ₁₂ : Perceived Usability of OARS is not affected by users' Flexible Traveling Behavior after adjusting for the effect of the covariate, System's Flexibility.	ANCOVA

 Table 3.1: Research Questions and Hypotheses (continues)

 Table 3.1: Research Questions and Hypotheses (continues)

Phase	Study	Research Objectives	Research Questions	Hypotheses	Analysis
III	Case Study	To study the interrelationship between System's Flexibility, Users' Flexibility and Perceived Usability	RQ6: How do service quality attributes of airlines and external variables jointly predict flexible behavior of travelers?	H_{13} : Flexible behavior of travelers cannot be predicted by service quality attributes and external variables.	Multiple Regression Analysis
		of Online Airline Reservation Systems and to determine the Perceived Usability	RQ7: How does user Perceived Usability with the existing and the proposed system differs?	H ₁₄ : User Perceived Usability with existing and proposed systems is different across the three groups.	Two-Way ANOVA
		of the existing and proposed systems.	RQ8: Is there a multivariate main effect of user's Flexible Traveling Behavior (High, Medium and Low) on effectiveness, efficiency and	H_{15} : There are differences among effectiveness, efficiency and satisfaction caused by the users' Flexible Traveling Behavior.	MANOVA
			satisfaction of the proposed system?	H_{16} : Effectiveness, efficiency and satisfaction in the proposed FOARS is highest for users with highest flexible behavior.	Post-Hoc

3.1.1 Phase I: Assessing User Needs (System's Flexibility & Users' Flexibility)

In Chapter 2, two different perspectives of flexibility, i.e. (1) System's Flexibility (Computer) and (2) Users' Flexibility (Humans) were discussed. Therefore, in Phase I, existing Online Airline Reservation Systems (Interface A) were used to assess the System's Flexibility and Users' Flexibility. Three studies were conducted in this phase as follow:

- 1. A study to investigate issues with flexibility and if flexibility is the reason for not using Online Airline Reservation Systems.
- 2. A study to investigate users' flexible behavior in terms of compromising on service quality attributes of an airline.
- 3. A study to examine if integration of OTAs features can make SBTs more flexible in the context of Online Airline Reservation Systems?

Section 3.2 will discuss Phase I in detail.

3.1.2 Phase II: User's Classification (Interrelationship Testing of Variables)

Phase II was designed to carry out extensive relationship testing of variables and their sub-measuring constructs towards developing a framework for designing a more Flexible Online Airline Reservation System. Two detailed studies were conducted in this phase with the following research objectives:

- To explore the concept of users' perception on factors influencing Flexible Traveling Behavior and Flexible Online Airline Reservation Systems.
- To classify Users' on the basis of their Flexible Traveling Behavior (Low, Medium, High) and to investigate interrelationships among System's Flexibility, Users' Flexibility and Perceived Usability of existing Online Airline Reservation Systems.

Section 3.3 will discuss Phase II in detail.

3.1.3 Phase III: Case Study (Testing the Proposed Framework)

Phase III is related to the design of a case study (paper prototype) and the corresponding analysis to answer the final research questions. Participants were requested to complete the usability evaluation of the existing and proposed interfaces (Interface B). Quantitative technique was used to analyze the data collected in the context of System's Flexibility, Users' Flexibility and Perceived Usability of the systems. Section 3.4 will discuss Phase III in detail. The following research questions were addressed:

- 1. How do service quality attributes of airlines and external variables jointly predict the flexible behavior of travelers?
- 2. How does user Perceived Usability with existing and proposed systems differs?
- 3. Is there a multivariate main effect of user's Flexible Traveling Behavior (High, Medium and Low) on effectiveness, efficiency and satisfaction of the proposed system?

3.2 Phase I: System's Flexibility and Users' Flexibility

Three studies were conducted in this phase as shown below using the existing OARS to assess the System's Flexibility and Users' Flexibility.

3.2.1 Study 1: Issues with Flexibility

This study is to address the 1st research question.

RQ1: What are the issues with flexibility of Online Airline Reservation Systems, whether or not flexibility is one of the reasons for users not using such systems?

3.2.1.1 Rationale

The rationale behind pilot study 1 was to assess users' needs from the System's Flexibility perspective. The primary objective of Requirements Engineering (RE) is to build a foundation of a product that satisfies the customers' needs and interests. In addition to pure functionality, customers want to see non-functional characteristics in the systems, such as security, stability, usability and high performance. Such Non-Functional Requirements (NFRs) have become essential for the success [130] of today's businesses.

The role of NFRs for the success or failure of any Transaction Processing System is as important as in any other systems. In the case of electronic commerce, security and privacy of consumers' sensitive personal data are one of the major concerns [131]. Consumers' lack of full adoption of electronic commerce solutions is not merely due to the concern on security and privacy, but also caused by a variety of non-functional characteristics such as flexibility, consistency, learnability, and reliability. This study specifically addresses the notion of flexibility of such systems, trying to establish how much the success of Business to Consumer (B2C) e-commerce hinges on them.

The main objective of pilot study 1 in phase 1 was to investigate customers' concerns for not using Online Airline Reservation Systems, and to study the relationship between flexibility and the use of Online Airline Reservation Systems. Therefore, this study focuses on the following sub research questions:

- What are the reasons for not using Online Airline Reservation Systems?
- Do customers have concerns with the flexibility of current Online Airline Reservation Systems?
- Does the perceived flexibility of Online Airline Reservation Systems significantly affect the usability of such systems?
In this phase, the methodology of the study followed quantitative research based on the self-reporting questionnaire and testing of hypothesis as shown in Figure 3.2. The hypotheses were:

H₁: Non-functional Requirements (NFRs) are perceived to have an impact on the usability of OARS.

 H_2 : Functional Requirements (FRs) are perceived to have an impact on the usability of OARS.

H₃: The perceived flexibility of OARS affects the usability of such systems.

H₄: Functional Requirements of OARS are inversely associated with the flexibility of the systems.

 H_5 : The availability of resources and skills set influence upon the usability of OARS.



Figure 3.2: Research Model for the Hypothesis to be Tested

3.2.1.3 Validity

A questionnaire was designed in order to test the model and the above hypotheses (Appendix A). A mixture of two approaches-adaptive (questions from existing literature) and development (questions as per the required scenario) – was used to prepare the questionnaire. Relevant and useful questions were adapted from the literature review [132]–[134].

In order to ensure that items on the questionnaire were related to the constructs being measured, the Content Validity Index (CVI) was used. According to Burns and Grove [135] Content Validity is obtained from three sources: literature, representatives of the relevant populations, and experts. The Content Validity Index (CVI) developed by Waltz and Bausell [136] was used in this study.

Three human factor students with expertise in usability were asked to rate each item on the questionnaire based on Relevance, Clarity, Simplicity and Ambiguity on the four-point scale. The results of CVI were analyzed and items that had CVI over 0.75 remained and the rest were discarded. The remaining items were modified, based on the experts' opinions.

The questionnaire consists of 13 questions on Non-functional Requirements -NFR (excluding flexibility), 9 questions on Functional Requirements - FR, 11 questions exclusively on Flexibility, and 3 questions on assessing Users' resources & skills when booking or using Online Airline Reservation Systems.

The pilot testing of the questionnaire was also conducted through a series of informal interviews with PhD and Master level students at Universiti Teknologi PETRONAS before sending the questionnaire to the target audience.

3.2.1.4 Sample Size

To analyze the model under consideration and to study the factors that impact the use of online reservation systems, more than 200 copies of the questionnaire were emailed or hand-distributed to various students and faculty members of two Universities namely, Universiti Teknologi PETRONAS (UTP) and Universiti Teknologi Mara (UiTM). The analysis of the participants is shown in Figure 3.3.



Figure 3.3: User's Analysis of Participants from UTP

3.2.1.5 Response Rate

140 responses were received, yielding a response rate of 70% of the total population surveyed. Out of 140, 78 respondents reported to be inexperienced with online ticket buying; the remaining 62 respondents considered themselves experienced.

3.2.1.6 Scale

Each construct included questions presented in a five-point Likert mode, ranging from "strongly agree" to "strongly disagree." Respondents' responses were scored as: for the "strongly agree" response was assigned a score of 5, while for the "strongly disagree" response was assigned a score of 1. Consequently, users' gaining higher scores in a certain scale showed stronger preferences toward the specific scale.

3.2.1.7 Analysis

In order to test influence of variables on another, 5 research hypotheses were designed in this study that was correlated using Pearson Correlation Coefficients. Pearson Correlation (Section 3.5.2) is regarded as the most familiar measure of examining dependence between two quantities. It indicates the strength of a linear relationship between two variables; however its value generally does not completely characterize their relationship [137]. Additional analyses were performed on data, by computing descriptive analysis (Section 3.5.1) and Reliability Analysis (Section 3.5.3). Data analysis carried out will be discussed in Chapter 4.

3.2.2 Study 2: Users' Flexible Behavior in Terms of Compromising on SQAs

This study is to address the 2^{nd} research question.

RQ2: How flexible users are in term of compromising on Service Quality Attributes (SQAs) of Online Airline Reservation Systems?

3.2.2.1 Rationale

The rationale behind study 2 was to assess users' needs from the Users' Flexibility perspective. A mixture of two approaches – adaptive (questions from existing literature) and development (questions as per the required scenario) – was used to prepare the questionnaire (Appendix B). Relevant and useful questions were adapted from the literature review [134], [138], [139].

3.2.2.2 Methodology

Study 2 followed a quantitative research methodology based on self-reporting questionnaire and testing of hypothesis. The null and alternate hypotheses for the study are given below:

 H_6 : The level of satisfaction with existing SBTs is different for respondents with different attitudes towards Users' Flexibility in compromising on service quality attributes of the airline.

 H_0 : The level of satisfaction with existing SBTs is same for respondents with different attitudes towards Users' Flexibility in compromising on service quality attributes of the airline.

H₇: The level of satisfaction with existing OTAs is different for respondents with different attitudes towards Users' Flexibility in compromising on service quality attributes of the airline.

 H_0 : The level of satisfaction with existing OTAs is same for respondents with different attitudes towards Users' Flexibility in compromising on service quality attributes of the airline.

3.2.2.3 Validity

For the second pilot study, after ensuring Content Validity using Waltz and Bausell [136] scale, Criterion Validity was computed through Concurrent validity to ensure whether the questionnaire is truly measuring users' satisfaction with existing SBTs and OTAs.

In psychometrics, Criterion Validity is a measure of how well one variable or set of variables predicts an outcome based on information from other variables, and will be achieved if a set of measures from a personality test relate to a behavioral criterion on which psychologists agree [140]. Criterion Validity was ensured through implementing concurrent validity of the measuring constructs, i.e. Users' Flexibility in terms of compromising on service quality attributes of airlines (Users' Flexibility) and effectiveness, efficiency & satisfaction (Perceived Usability), by taking feedback of 12 randomly selected users. Concurrent validity is particularly useful to demonstrate where a test correlated with a measure has previously been validated [141]. In this case, Pilot study 1 had already established a strong correlation of Users' Flexibility with Perceived Usability of Online Airline Reservation Systems (H₅).

Validity check results showed a strong positive correlation of r = 0.354, p < 0.05 between Users' Flexibility in terms of compromising on service quality attributes of

airlines and Satisfaction. Essentially, this means Users' Flexibility in terms of compromising on service quality attributes of airlines can be used to predict their satisfaction from existing online reservation systems, both SBTs and OTAs.

3.2.2.4 Sample Size

250 copies of the questionnaire were hand-distributed to various students at Universiti Teknologi PETRONAS and four travel agencies in Malaysia located in the city of Ipoh, Perak. The analysis of the participants from travel agencies and those respondents from Universiti Teknologi PETRONAS are shown in Figure 3.4 and Figure 3.5 respectively.



Figure 3.4: User's Analysis of Participants from Travel Agencies



Figure 3.5: User's Analysis of Participants from UTP for Study 2

3.2.2.5 Response Rate

169 responses were received, yielding a response rate of 67.60% of the total population surveyed. Out of 169, 106 (63%) respondents were from Universiti Teknologi PETRONAS, while the remaining 62 (37%) respondents were travelers visiting traveling agencies.

3.2.2.6 Scale

Each construct included questions presented in a five-point Likert mode, ranging from "strongly agree" to "strongly disagree." Respondents' responses were scored as: for the "strongly agree" response was assigned a score of 5, while for the "strongly disagree" response was assigned a score of 1. Consequently, users' gaining higher scores in a certain scale showed stronger preferences toward the specific scale.

3.2.2.7 Analysis

In order to test the stated alternate and null hypothesis H_6 and H_7 in this study, Oneway Analysis of Variance (ANOVA) was computed to determine the satisfaction mean of users with existing SBTs and OTAs. ANOVA (see Section 3.5.4) is a statistical method used to compare the means of two or more groups.

ANOVA for H_6 was computed to determine satisfaction mean of users with existing self-booking tools of the airline, who at the same time reported their flexibility level in terms of compromising on service quality attributes of the airline. The respondents had to select from the three given options of, (1) Can compromise on service quality attributes, (2) May compromise on service quality attributes, (3) Cannot compromise on service quality attributes. This was followed by ANOVA for H_7 to determine satisfaction mean of users with existing online travel agencies of the airline, who at the same time reported their flexibility level in terms of compromising on service quality attributes of the airline. The respondents had to select from the three given options of, (1) Can compromise on service quality attributes, (2) May compromise on service quality attributes, (3) Cannot compromise on service quality attributes. Data analysis carried out will be discussed in Chapter 4.

3.2.3 Study 3: Integration of OTAs Features can make SBTs more FOARSs

This study is to address the 3rd research question.

RQ3: How users' satisfaction with an existing SBTs is rated against their choice of OTA feature and reflected in their integration assessment of the same for making SBTs more flexible Online Airline Reservation Systems?

3.2.3.1 Rationale

The rationale behind study 3 was to investigate if the integration of some OTA features could make SBTs a more Flexible Online Airline Reservation Systems and how that can be achieved? Therefore, a preliminary but comprehensive focus group research was conducted with airline executives, using quantitative survey method to:

- examine their subjective satisfaction with existing self-booking tools provided by the airlines,
- report if the group approve or disapprove the proposed idea of the four OTA features (Matrix Display, Opaque Fare, Alternate Airport Search, Hotel Search Facility) integrated into SBTs so as to make them Flexible Online Airline Reservation Systems, and
- recommend an OTA feature of their choice, for making SBTs Flexible Online Airline Reservation Systems.

3.2.3.2 Methodology

Study 3 followed a quantitative research methodology based on self-reporting questionnaire to test the hypothesis. The study used the same data set as obtained in

study 2. Therefore, the steps required for validity, sample size, response rate and scale used in the methodology can be seen in study 2.

The null and alternate hypothesis for the study is given below:

H₈: Users' satisfaction with existing SBTs is different across their choice of four OTA features for making SBTs more Flexible Online Airline Reservation Systems.

H₀: Users' satisfaction with existing SBTs is same across their choice of four OTA features for making SBTs more Flexible Online Airline Reservation Systems.

3.2.3.3 Analysis

In order to test if the respondents approve or disapprove the idea of OTA features integrated into SBTs so as to make them Flexible Online Airline Reservation System, means plot was examined. This was followed by two-way ANOVA analysis to test the satisfaction level of existing Online Airlines Reservation Systems. One of the basic assumptions before performing any analysis of variance is to check for normality of sampling distribution of mean. The sample size for this study (n) was 169, and according to central limit theorem if a random sample of size n is > 30 and it is derived from an infinite population with finite standard deviation, then the standardized sample mean converges to a standard normal distribution [142].

To perform a Two-way ANOVA, respondents were requested to indicate if integration of OTAs features will make Online Airlines Reservation System more flexible or not and also to recommend an OTAs feature (from the given four options of Opaque Fare, Matrix Display, Hotel Search Facility, Alternate Airport Search) for integration into SBTs.

The F-statistic was interpreted in analysis of variance since it is a ratio of the explained variability to the unexplained variability (taking into account the degrees of freedom). A larger F-statistic indicates that more of the total variability is accounted for by the model [143]. Data analysis carried out will be discussed in Chapter 4.

3.3 Phase II: Users' Classification and Interrelationship Testing of Variables

Two studies were conducted in this phase to classify users on the basis of their flexible traveling behavior.

3.3.1 Study 4: Users' Perception on Factors Influencing Flexible Traveling Behavior

This study is to address the 4th research question.

RQ4: How users' perception on factors influencing Flexible Traveling Behavior and Flexible Online Airline Reservation Systems is determined?

3.3.1.1 Rationale

The rationale behind this study was to investigate and explore the concept of users' perception on factors influencing Flexible Traveling Behavior and Flexible Online Airline Reservation Systems.

3.3.1.2 Methodology

In order to gather an in-depth understanding of human behavior and the reasons that govern such behavior, qualitative research was conducted. This qualitative exploratory study adopted a grounded theory approach to investigate the users' perception on their Flexible Traveling Behavior and Flexible Online Airline Reservation Systems. Grounded Theory is a research method in which the theory is developed from the data, rather than the other way around [144], since it is an appropriate way to research a previously little studied area in Information Systems research. According to Strauss [145], "A grounded theory is one that is inductively derived from the study of the phenomenon it represents. That is, it is discovered, developed, and provisionally verified through systematic data collection and analysis of data pertaining to that phenomenon."

Moreover, this methodology provides an ideal and flexible guideline to analyze qualitative data and equip researchers with necessary understanding of underlying concepts to build theories through successive levels of data analysis [146]. Researchers [147]–[149] have recognized this method as an authentic research tool in qualitative data analysis due to its procedural credibility.

The population of this study consisted of travelers who had experience in purchasing tickets through airlines Self-Booking Tools (SBTs) and Online Traveling Agencies (OTAs). This was an important consideration, because travelers with experience in purchasing tickets through SBTs and OTAs could very well understand and relate to what being 'Flexible Traveling Behavior' mean from users' perspective as well as from systems' perspective.

The data was collected from three methods. (1) Two online travel forums, <u>http://www.travelblog.com</u> and <u>www.travellerspoint.com/</u> (2) Semi-structured indepth interviews and (3) Focus group. Use of online surveys to collect data has become a popular choice for researchers, with special reference to tourism data [150]. This was mainly due to the flexibility, reach and robustness offered by visual medium of internet. Likewise, in depth interviews and focus group were essentially required in this research, so as to build deeper understanding of respondents' perspective on Flexible Traveling Behavior and Flexible Online Reservation Systems, which otherwise may not be possible to obtain through online travel forums alone. Moreover, in short interviews researcher is in a position to pick up non verbal cues and even rephrase questions so as to personalize them and make respondents feel at ease to answer them.

The following questions were raised at travel forums:

- Which factors influence upon your Flexible Traveling Behavior?
- Which factors influence upon your perception of a Flexible Online Airline Reservation System?

The semi structured in-depth interview of actual travelers was conducted at Kuala Lumpur International Airport from 11–13th March, 2011. A realistic, flexible and ethically accepted approach was adopted to identify potential research participants. Since this research involved travelers and reservation systems, therefore, Kuala Lumpur International Airport (KLIA) was visited for consecutive three days in order to approach a number of volunteer travelers.

Finally, the focus group interviews were conducted with 3 managers, 5 junior executives and 3 technical experts of three local airlines, namely (1) Malaysian Airline, (2) Fire Fly, and (3) Air Asia. The interviews were held from 15–25th March, 2011.

3.3.1.3 Validity

In order to ensure study's trustworthiness, two methods were employed i.e. Triangulation and Negative Case Analysis.

With regards to triangulation, the three sources and three different data collection methods; online travel forums, semi-structured in-depth interviews and focus group, were employed. This was important to see that data obtained from different independent data sources converged on something similar, or at least do not oppose to each other [151]. The data was analyzed by two authors independently and then discussed together to derive emerging themes, categories and to also ensure credibility. Negative case analysis was performed on the initial derived emerging themes [146], [152]. The purpose was to see if the characteristics of the derived emerging theme sufficiently inculcated the true essence of whole research and were applicable to all cases.

3.3.1.4 Sample Size

31 respondents of the two questions were from travelblog's and travellerspoint forums. 28 travelers were interviewed, and each interview lasted from 10-15 minutes.

Finally, the focus group interviews were conducted with 3 managers, 5 junior executives and 3 technical experts of three local airlines, namely (1) Malaysian Airline, (2) Fire Fly, and (3) Air Asia. The interviews were held from 15–25th March, 2011 (permission letter enclosed see Appendix F). The analysis of the participants from the two online travel forums, semi-structured in-depth interviews and the Focus group are shown in Figure 3.6, Figure 3.7 and Figure 3.8 respectively.



Figure 3.6: User's Analysis of Participants from Online Travel Forums



Figure 3.7: User's Analysis of Participants from In-depth Interviews



Figure 3.8: User's Analysis of Participants from Focus Group

3.3.1.5 Response Rate

43 responses were received from online travel forums. Out of 43, 12 cases were rejected due to the ambiguous respondents yielding a response rate of 72% of the total responses. 28 responses in-depth interviews and 11 responses from focus group were collected with no rejected cases. The demographics of online travel forums, in-depth interviews and focus group can be seen in Chapter 4.

3.3.1.6 Scale

The analysis of interview transcripts was based on an inductive approach which is meant to identify emerging patterns in the data by using thematic codes. Inductive analysis looks for emerging patterns, themes and categories through analysis of data and opposes imposition of the same, prior to data collection and analysis [153].

3.3.1.7 Analysis

The data collected from three different sources was examined for triangulation. It depicted a similarity pattern, especially in case of data collected from online travel forums and in-depth interviews of travelers at KLIA.

Data analysis of later, however, provided a more detailed perspective of travelers flexible behavior by incorporating socio-economic factors and societal influences. The focus group, being technical experts, however significantly contributed towards identifying factors that may influence upon perceived flexibility of reservation systems. After giving much thought process to results as shown in Chapter 4, Section 4.4.1, 6 themes emerged under factors influencing upon Flexible Traveling Behavior and 3 themes emerged under factors influencing upon perceived flexibility of a reservation system as shown in Table 3.2.

	Emerging Themes							
_	Factors Influencing Upon Flexible Traveling Behavior	Factors Influencing Upon Perceived Flexibility (PF) of an Online Airline Reservation System						
1.	Travelers' flexible behavior is moulded by their traveling consciousness	1. Systems perceived flexibility is influenced by its Perceived						
2.	Travelers' flexible behavior is moulded by their belief that they have the	Usability. 2. Systems perceived flexibility is						
3.	Travelers' flexible behavior is moulded by their self-belief as flexible travelers.	3. Systems perceived flexibility is influenced by comparison of						
4.	Travelers' flexible behavior is moulded by societal influences.	features on the actual level of effect regarding to complete the						
5.	Travelers' flexible behavior is moulded by how they attribute a cause to their traveling behavior.	reservation process.						
6.	Travelers' flexible behavior is moulded by their prior traveling experiences.							

Table 3.2: Emerging Themes on Factors Influencing upon FTB and PF

3.3.2 Study 5: A Study to Classify Users' on the Basis of Flexible Traveling Behavior

This study is to address the 5th research question.

RQ5: How to classify Users' on the basis of their Flexible Traveling Behavior into High, Medium and Low flexible and how to investigate interrelationships among System's Flexibility, Users' Flexibility and Perceived Usability of existing Online Airline Reservation Systems?

3.3.2.1 Rationale

The study purpose was to (i) classify users' on the basis of their Flexible Traveling Behavior (Highly Flexibile, Medium Flexible, Low Flexible) and (ii) to investigate interrelationships among System's Flexibility, User's Flexibility and Perceived Usability of the Online Airline Reservation Systems.

3.3.2.2 Methodology

The quantitative research methodology was adopted using survey to address the following four hypotheses.

H₉: Users can be classified on the basis of their Flexible Traveling Behavior.

 H_{10} : User's Flexible Traveling Behavior and their Perceived Usability is correlated.

H₁₁: User's Flexible Traveling Behavior and System's Flexibility is correlated.

 H_{12} : Perceived Usability of Online Airline Reservation Systems is not affected by users' Flexible Traveling Behavior after adjusting for the effect of the covariate, System's Flexibility.

3.3.2.3 Validity

In order to ensure that items on the questionnaire were related to the constructs being measured, the Content Validity Index (CVI) was used. According to Burns and Grove [135] Content Validity is obtained from three sources: literature, representatives of the relevant populations, and experts. The Content Validity Index (CVI) developed by Waltz and Bausell [136] was used in this study.

Three human factor students with expertise in usability were asked to rate each item on the questionnaire based on Relevance, Clarity, Simplicity and Ambiguity on the four-point scale. The items that had CVI over 0.75 remained and the rest were discarded. The remaining items were modified, based on the experts' opinions.

3.3.2.4 Sample Size

To investigate the above hypotheses, 90 random cases were selected for validating the results of transformation scale and to perform preliminary interrelationship of variables before performing the final analysis on the data set of 273 responses.

3.3.2.5 Response Rate

Out of 273, 90 random cases were selected as shown in Figure 3.9. Randomly selected cases will be used for transformation scale.



Figure 3.9: User's Analysis of Participants for Transformation Scale

3.3.2.6 Scale

Five-point Likert mode was used ranging from "strongly agree" to "strongly disagree." Respondents' responses were scored as: for the "strongly agree" response was assigned a score of 5, while for the "strongly disagree" response was assigned a score of 1. Consequently, users' gaining higher scores in a certain scale showed stronger preferences toward the specific scale.

3.3.2.7 Analysis

Section B of the questionnaire consisted of questions on Service Quality Attributes (SQAs) of an airline that users' are ready to forgo or compromise on, in order to become flexible travelers. By opting to become flexible travelers, users would get discounted fares at the cost of being unaware of three hidden characteristics of their traveling itinerary (1) their seat details/confirmation, (2) date of flying confirmation and (3) time of flying confirmation. All questions posted in this section were unidirectional and designed as such that the extent to which a user would agree with a particular statement reflected his/her flexible behavior in terms of flying on flexible dates/times. The following unidirectional scale was used, where 1 denoted 'Highest' and 5 denoted 'Least' flexibility in terms of compromising on service quality attributes of an airline as shown in Table 3.3.

Rate	your priorities for the following	Highest	High	Neutral	Low	Least
serv	ice quality attributes of an airline in					-
term	s of their importance.		2	3	_4	5
1	Flying date confirmation					
2	Flying carrier confirmation					
3	Flying time confirmation					
4	Number of stop-over					
5	Number of connected flights					
6	Ticket class (economy/business)					
7	Seat specifications					
8	Discounted Airfares					
9	Destination/Source Airport					
10	Immediate confirmation of itinerary on purchase of ticket					
9 10	Destination/Source Airport Immediate confirmation of itinerary on purchase of ticket					

Table 3.3: Uni-directional Scale to Measure Users' Flexibility

Transformation Scale – Once the score of the respondents was recorded, it required adoption of a validated mechanism to test the hypothesis H_9 , whereby individual ratings of participants on 10 service quality attributes could be transformed into singular unit, such as a score or product.

In a study conducted on aesthetic appraisal of 5 most popular aviation websites in the world, a similar methodology based on transformation of ratings into a product score, has been used [154]. Respondents in the study evaluated 5 popular SBTs of airlines on the given parameters by assigning a score of 1 (Very Low) to 5 (Very High). Then their score was transformed into a product, by examining frequency of occurring of each rating (1, 2, 3, 4, 5) and multiplying that frequency (e.g., 1 occurs two times, 2 occurs zero times, 3 occurs four times, 4 occurs four times, 5 occurs zero times) with that of transforming scale product. The final total is added to obtain a unique product score, which in case of given example 2 as shown in Table 3.4. Data analysis carried out will be discussed in Chapter 4.

Dating	VL	L	N	Н	VH
Kaung	1	2	3	4	5
Scale	-2	- l	0	+1	+2
Rating Frequency	2	0	4	6	0
Product (Rating Frequency * Scale)	-4	0	0	6	0
Product Score			2		-

Table 3.4: Transformation Scoring Scale

 H_{10} investigated interrelationship between users' Flexible Traveling Behavior (FTB) and their Perceived Usability of FOARS. For this investigation, Kendall's Tau (see Section 3.5.9) method for examining bivariate correlations was selected because it essentially met the nonparametric conditions of the study. First, the study used a small data set of 62 respondents only, reporting their Perceived Usability. Secondly, users were classified into three categories of flexible behavior (high, medium, low) on the basis of their Users' Flexibility score. The correlation analysis was performed and significance level was interpreted. Data analysis carried out will be discussed in Chapter 4.

 H_{11} investigated interrelationship between users' FTB and System's Flexibility (comprised of adaptability, adaptivity, and personalisation). Pearson Correlations was computed and significance level was checked and interpreted accordingly. Data analysis carried out will be discussed in Chapter 4.

 H_{11} called for the further investigation in order to ascertain interrelationship between users' Flexible Traveling Behavior and System's Flexibility. Consequently, in H_{12} , Perceived Usability of Flexible Online Airline Reservation Systems (PU) was included as a dependent variable to investigate how the two variables, (1) Users' Flexible Traveling Behavior and (2) System's Flexibility together predict Perceived Usability of Online Airline Reservation System. Moreover, since the interrelationship between users' Flexible Traveling Behavior and Perceived Usability of Flexible Online Airline Reservation Systems is already ascertained in H_{10} , therefore, users' Flexible Traveling Behavior was taken as a fixed factor and System's Flexibility was included as a Covariate, to reduce within group error variance and to eliminate confounds. ANOVA includes one or two continuous variables that predict the outcome or dependent variable. However, continuous variables such the one that are not part of the main experimental manipulation but may have an influence on dependent variable are known as covariates in the Analysis of Covariance (ANCOVA). ANCOVA takes into account confounding variables to give a clear measure of effect of the experimental manipulation, and the analysis is performed as such to first examine influence of independent or fixed factor (users' FTB) on dependent variable (Perceived Usability of Flexible Online Airline Reservation Systems) and then experiment is manipulated by introducing a covariate (System's Flexibility).

One important assumption was checked before performing ANCOVA, i.e., independence of the covariate and treatment effect. This assumption requires that covariate (System's Flexibility) should not be different across three users' Flexible Traveling Behavior Groups in the analysis. To meet this assumption, One Way Independent ANOVA was performed, with Perceived Usability of Flexible Online Airline Reservation System across three groups as an independent variable, and System's Flexibility as an outcome variable. This analysis should be non-significant to meet the assumption of ANCOVA. This was followed by performing ANCOVA results are interpretation. Data analysis carried out will be discussed in Chapter 4.

3.3.3 Research Way Forward

The Matrix display and hotel search features are unique in the sense of incorporating multiple sources reservation information, however from integrating into the proposed framework of SBTs perspective, they are not very feasible since it requires merger of multiple information resources, that might not be acceptable to an airline due to its privacy policy and other regulations. On the contrary, alternate airport search is related to provide additional information, and the extent to which a traveler is willing to be flexible in identifying his/her destination sources. This feature seems practical and has implications for integration into SBTs. Finally, unlike other OTA innovations, the opaque fare mechanism depends on hidden characteristics of the traveling plan, thus leveraging upon traveling behavior of leisure travelers, who are always up for

grabs and less sensitive to traveling plans. The findings of the studies showed that this OTA feature to be the most recommended for integration into SBTs as shown in Figure 3.10.



Figure 3.10: Proposed FOARS after Integrating Opaque Fares into SBTs

Since we are investigating flexibility from both, system point of view and users' perspective, integration of opaque fares concept into SBTs would increase the usability of the system by way of improving System's Flexibility and by making users' more flexible in their decision making as shown in Figure 3.11.





As discussed above, flexibility is referred to its ability to respond to internal or external changes. Change can be defined as the transition over time which requires change agent. Researchers argued that "if the change agent is external to the system, then the change under consideration is a *flexible*-type change" [12]. Therefore, in case of SBTs incorporated with opaque fares would serve the role of external change agent

by way of providing flexibility in users' decision making. Similarly, "if the change agent is internal to the system, then the change under consideration is an *adaptable*-type change" [12]. Thus the provision of opaque fares into SBTs also serves the role of internal change agent by way of providing the capability of accepting changed decisions. If no change agent exists, then the system is rigid (no change can occur). Since provision of opaque fares could make users flexible and also increases the adaptability of the system, it is expected that the usability of the system would be enhanced.

3.4 Phase III: Case Study to Test the Proposed Framework

Phase III was designed to study the interrelationship between System's Flexibility, Users' Flexibility and Perceived Usability of Online Airline Reservation Systems and to determine the Perceived Usability of the existing and proposed systems. In order to investigate the relationship between Perceived Usability and travelers Flexible Traveling Behavior (FTB) with existing and proposed systems the following three research questions were investigated:

- **RQ6:** How do Service Quality Attributes of airlines and External Variables jointly predict flexible behavior of travelers?
- **RQ7:** How does user Perceived Usability with existing and proposed system differs?
- **RQ8:** Is there a multivariate main effect of user's Flexible Traveling Behavior (High, Medium and Low) on effectiveness, efficiency and satisfaction of the proposed system?

3.4.1 Rationale

The case study was designed in order to investigate the relationship between user Perceived Usability and travelers Flexible Traveling Behavior with the existing and the proposed Online Airline Reservation Systems.

3.4.2 Methodology

The quantitative research methodology, based on investigating three (03) research questions through survey questionnaires was adopted. Users' were given hard copies of the questionnaire with following sections.

Section A of the questionnaire was designated to collect demographic profile of the respondents. Section B consisted of service quality attributes of an airline that users' are ready to forgo or compromise in order to become flexible travelers. By opting to become flexible travelers, users will get discounted fares at the cost of not knowing their seat confirmation, date of flying confirmation and time of flying confirmation. All questions posted in this section were uni-directional and designed as such that the extent to which a user would agree with a particular statement reflected his/her flexible behavior in terms of flying on flexible dates/times. Section C consisted of questions that particularly addressed the external factors that may influence upon the Perceived Usability of the system. The details of each measuring construct have already been discussed in Chapter 2. Section D consisted of questions to measure Perceived Usability of the system.

To answer the above research questions the following hypotheses were created:

- **H**₁₃: Flexible behavior of travelers cannot be predicted by the Service Quality Attributes and External Variables.
- H₁₄: User Perceived Usability with existing and proposed systems is different across the three groups.
- H₁₅: There are differences among effectiveness, efficiency and satisfaction caused by the users' Flexible Traveling Behavior.
- H₁₆: Effectiveness, efficiency and satisfaction in the proposed FOARS is higher for users with highest flexible behavior.

3.4.3 Validity

Validity technique known as Item Discrimination Index was employed in this study [155]. Item Discrimination Index indicates how adequately an item separates or discriminates between high scorers and low scorers on an entire test [157]. It is a measure of difference between the proportion of high scorers answering an item correctly and the proportion of low scorers answering the item correctly. Andy Field [156] argues that item discrimination means that respondents with different score should also differ in the construct of researchers' interest. Kelley [158] suggested that item discrimination should be based upon following two corollaries and pose unidirectional questions to respondents, so that the degree of their agreement with a particular statement could be used to discriminate them with respondents with certain levels of disagreements over the same statement.

- Respondents with the same score should be equal to each other along the measured construct.
- Respondents with different scores should be different to each other along the measured construct.

To meet the item discrimination validity requirement, the questionnaire was designed with uni-directional questions. This paved way to discriminate respondents on the basis of *the degree to* which they agree being flexible travelers or not.

3.4.4 Sample Size

To investigate the above three (03) research questions, more than 500 travelers were requested to fill in the survey questionnaire during the Malaysian Association of Tour and Travel Agents (MATTA) fair 2011. MATTA fair is a world known exhibition in the tourism industry which is held every year. Three (03) days (11-13th March 2011) were spent at Putra World Trade Centre (PWTC), Kuala Lumpur (Appendix G).

3.4.5 Response Rate

273 responses were received, yielding a response rate of 54.6% of the total population surveyed. However, 23 cases were dropped due to the ambiguous respondents yielding a response rate of 50%.

3.4.6 Paper Prototype

A paper prototype as shown in Figure 3.12 and Figure 3.13 was developed to get the users' response to classify them into different categories.

9 9		- Annotes					and the second			-
eps	rt :	1	8 7 <u>. V</u> .		15	-		Wate		
look	ing ty	pe :	0	lexib	le	0	Inflexibl	e		
lo. 0	f day	stob	e flex	lble.			1			
	The second s		Contraction of the local division of the loc							
Mon	Tue	Wed	Thu	Fri	Sat	Sun		6.00	distants fi	
Mon	Tue	Wed 1	Thu 2	Fri 3	Sat 4	Sun 5		Ava	ilablefi	ight
Mon	Tue	Wed 1 8	Thu 2 9	Fri 3 10	4 11	5 12	Adult	Ava L	ilablefi 🕜 🗋	ight
Mon 6 13	Tue 7 14	Wed 1 8 15	7hu 2 9 16	Fri 3 10 17	4 11 18	5 12 19	Adult Childi	Ava : : ren :	labiei) (7) (7)	ight
Mon 6 13 20	Tue 7 14 21	Wed 1 8 15 22	7 hu 2 9 16 23	Fri 3 10 17 24	4 11 18 25	5 12 19 26	Adult Childi Infanl	Ava : : ren :		ight

Figure 3.12: Prototype of a Flexible Booking Window for Flexible Travelers

Book a Flight - One Way Ticket–	
From :	
To	4
Depart :	
Booking type :) Flexible Ø Inflexible	
Select exact date :	
Adult : 👔 💽 💙 Children : 👔 💽 Infants : 👔 🚺	
Book Now	

Figure 3.13: Prototype of a Flexible Booking Window for Inflexible Travelers

3.4.7 Usability Evaluation of Prototype

The usability of a prototype named as interface B was inspected through heuristic evaluation. 3 HCI graduates who had usability evaluation and testing experiences served as expert evaluators to evaluate the usability. An evaluation sheet was provided to each evaluator that contained a series of usability guidelines based on the well-recognized Research-Based Web Design and Usability Guidelines book [159]. The selected guidelines were adapted to be presented in a checklist format (Appendix H). The evaluators were requested to assign each guideline a problem severity rating from 0 (no problem) to 4 (usability catastrophe) based on the frequency, impact and persistence of the problem. According to Nielsen and Mack [160], using the mean of a set of severity ratings from three evaluators is satisfactory for many practical usability inspection purposes.

Furthermore, the evaluators were also requested to provide redesign suggestions for each problem identified. Based on all evaluators' evaluation and the results from studies, a new interface B (Prototype) was designed.

3.4.8 Scale

Questionnaire was designed using psychometric scales, which are commonly used in psychological research [79]. This technique was employed because psychometric scales tend to prompt an individual to respond to various questions that pertain to a given context and according to Davis [79] "*responses of individuals are an indication of their internal belief*". The participants were provided a hardcopy of the questionnaire for indicating their response by evaluating the prototype by using a 5 point likert scale ranging from (1) "Strongly Disagree" to (5) "Strongly Agree" (Appendix C).

3.4.9 Analysis

The analysis on RQ6, RQ7 and RQ8 are as follows:

Analysis of RQ6 – The first research question in the case study was investigated by performing statistical procedure in two steps. In Step 1, Pearson Correlation Coefficients of the ten quality service attributes along with external variables was computed to determine their association with flexible behavior of travelers and to also ascertain their individual range and strength of association. In Step 2, Multiple Regression Analysis (MRA) was performed to determine how service attributes quality and external variables jointly determine flexible behavior of travelers. MRA predicts values on a quantitative outcome variable, using several other predicting variables [161]. Data analysis carried out will be discussed in Chapter 4.

Analysis of RQ7 – In order to investigate the second research question of the case study i.e. how Perceived Usability of existing and proposed systems differs among users' with high, medium and low Flexible Traveling Behavior?, a two-way ANOVA analysis was performed. As mentioned above, one of the basic assumptions before performing any analysis of variance is to check for normality of sampling distribution of mean. The sample size for this study (n) was 250, and according to central limit theorem if a random sample of size n is > 30 and it is derived from an infinite population with finite standard deviation, then the standardized sample mean converges to a standard normal distribution [142].

Two-way ANOVA tested the *Effectiveness*. *Efficiency and Satisfaction* of the existing and proposed systems for travelers with High, Medium and Low Flexible Traveling Behavior. The F-statistic was interpreted in analysis of variance since it is a ratio of the explained variability to the unexplained variability (taking into account the degrees of freedom). A larger F-statistic indicates that more of the total variability is accounted for by the model [143].

Analysis of RQ8 – Third research question of the case study was examined with the help of MANOVA to see if there is a multivariate main effect of user's Flexible Traveling Behavior on the proposed system's effectiveness, efficiency and satisfaction. Multivariate normality requires that any linear combination of the dependent variables must be distributed normally. This assumption was checked by examining pair wise nonlinear relationships between dependent variables using scatter plots. The second assumption of multivariate analysis is homogeneity of the covariance matrices. It was met by examining Box's M, which tests the hypothesis that the covariance matrices of the dependent variables are significantly different across levels of the independent variable as show in Chapter 4, Section 4.8.1.2.

The overall F test for the three dependent variables was examined in Multivariate Tests by analyzing the statistic called Wilks' lambda (λ), and the F value associated with that which is significant at p <001. Lambda is a measure of the percent of variance in the Dependent Variables that is *not explained* by differences in the level of the Independent Variable. If the overall F test is significant, then it is a common practice to go ahead and look at the individual dependent variables with separate ANOVA tests. This was followed by a univariate ANOVA and Post-hoc multiple comparison tests that shows statistically significant effect on three dependent variables. Data analysis carried out will be discussed in Chapter 4.

3.5 Statistical Methods

In this thesis, different statistical methods have been chosen to analyse the data. The methods were, descriptive analysis, pearson coefficient correlation, reliability analysis, analysis of variance, analysis of covariance, multivariable analysis of variance, multiple regression analysis, Post Hoc Scheffe's Test and Kendall's Tau. The following subsection describes the statistical formulas used in this research.

3.5.1 Descriptive Analysis

Descriptive analysis is used to describe and summarize a collection of data in a clear, understandable and meaningful manner which allows simple interpretation of the data. There are two basic approaches, one is numerical and, second is graphical. Using the first approach i.e. numerical method, one might compute statistics such as mean and standard deviation. In graphical method one might create plots that contain detailed information about the distribution. Stem and leaf display and a box plot are famous plots in graphical method [162].

3.5.2 Pearson Coefficient Correlation

The correlation coefficient was developed by Karl Pearson from the idea introduced by Francis Galton in 1880s [163], [164]. Sometime, it is also termed as "Pearson's r". In statistics, correlation (linear dependence) measures the degree of association between two variables X and Y. It is a value between -1 and +1 inclusive. A positive value implies a positive association between the two variables (large values of X tend to be associated with large values of Y and small values of X tend to be associated with small values of Y). A negative value implies a negative or inverse association between the variables (large values of X tend to be associated with small values of Y). A negative value implies a negative or inverse association between the variables (large values of X tend to be associated with small values of Y and vice versa). It is defined as the covariance of the two variables divided by the product of their standard deviations.

3.5.3 Reliability Analysis

In statistics, reliability is treated as the consistency of a set of measurements or it is also referred as a measuring instrument which is used to describe a test. It is inversely related to the random error [165]. Reliability is said to be sample dependent as it is the property of the scores of a measure rather than the measure itself. Cronbach's alpha is the most common consistency measure which is typically interpreted as the mean of all possible split-half coefficients [166]. Cronbach's alpha is a more generalized form of Kuder-Richardson Formula 20 [166] used for estimating internal consistency. Reliability may be defined as the proportion of true score variability that is captured across respondents, relative to the total observed variability [167].

3.5.4 ANOVA

The aim of Analysis of Variance generally known as ANOVA is to test the significant difference between group means. ANOVA is commonly used if the user needs to compare performance of more than two parameters. ANOVA generalizes t-test to more than two groups by providing a statistical test of whether or not the means of several groups are all equal. Therefore, the advantage of ANOVA over t-test is

ANOVA can detect the effect of interaction between variables, and to test more complex hypothesis about the existing problem [168]. If the result indicates a significant difference, then it would be followed by post-hoc test to identify which mean of result is different.

ANOVA is a technique that was firstly proposed by R. A. Fisher in a 1918 article "*The Correlation Between Relatives on the Supposition of Mendelian Inheritance*". In 1921, his first application of the ANOVA was published which was later included in his book "*Statistical Methods for Research Workers*". ANOVA is used to compare group means [169]. ANOVA uses two hypotheses to determine the result, namely null hypothesis and alternate hypothesis.

3.5.5 ANCOVA

The aim of Analysis of Covariance generally known as ANCOVA is to compare one variable in 2 or more groups taking into account variability of other variables, called covariate [170]. ANCOVA is a technique that sits in between analysis of variance and regression analysis [171]. It combines one-way or two-way ANOVA with linear regression. In other words, it is a General Linear Model (GLM) with a continuous outcome variable and two or more forecaster variables where at least one is continuous and one is categorical. Continuous variable is always quantitative or scaled while the categorical variable is nominal or non-scaled.

3.5.6 MANOVA

Multivariate analysis of variance generally known as MANOVA is a generalized form of ANOVA which is used to analyze data that involves two or more than two dependent variables [172]–[175]. There are three basic advantages of MANOVA analysis. Firstly, it helps in finding the interactions among dependent variables, secondly, it helps in finding the interaction among independent variables and, thirdly, it helps in finding the effect of independent variable(s) on the dependent variable(s) [176].

3.5.7 Multiple Regressions

The term regression was first time introduced by Francis Galton in 1900s [177]. Galton's work was later extended by Udny Yule [178] and Karl Pearson [179] to a more general statistical context. In statistics [161], this technique is used to predict the relationship between one dependent variable and one or more independent variable(s). This technique is able to form a method of least square where sum of squared residuals between the regression plane and the observed values of the dependent variable are minimized [180].

3.5.8 Post Hoc Scheffe's Test

Post hoc tests are useful to explore the differences among means. It provides specific information on which means are significantly different from each other. Therefore, Post hoc tests are performed where researchers has already conducted F-test with a factor that consists of more than two means [181]. In statistics, there are many procedures to perform Post hoc tests, however, Scheffe's techniques is a most popular and flexible method introduced by Henry Scheffe. It is a method to adjust significance levels in linear regression analysis for multiple comparisons.

3.5.9 Kendall's Tau

Kendall's Tau is used to measure the strength of the relationship between the two variables. It is a measure of correlation which is carried out on the ranks of the data [182]. For any sample of n observations, there are [n (n-1)/2] possible comparisons of points (X_1, Y_1) and (X_3, Y_3) . Assume C is a number of pairs that are concordant, and D is a number of pairs that are not concordant than Kendall's Tau can be calculated [183].

3.6 Chapter Summary

In this chapter, we discussed the overall methodology of the thesis whereby existing Online Airline Reservation Systems were examined to assess the flexibility and usability of the systems. This chapter was divided into three phases and each phase contains one core research objective which was achieved through quantitative and qualitative techniques to assess System's Flexibility, Users' Flexibility and Perceived Usability of the systems. A redesign solution for enhanced usability for more Flexible Online Airline Reservation Systems was developed based on HCI guidelines and the flexibility tactics used in online travel agencies. A new Flexible Online Airline Reservation System design was applied, which led to a proposed interface with the integration of opaque mechanism. The two interfaces were used in the case study. Participants were requested to complete the evaluation of the existing and proposed interfaces.

CHAPTER 4

RESULTS AND ANALYSIS

4.0 Chapter Overview

The research methodology discussed in Chapter 3 was divided into three phases addressing one core research objective with corresponding research questions in each phase. The same pattern is followed in this chapter to organize the results obtained through the corresponding hypotheses. To achieve the 1st research objective, Section 4.1, 4.2 and 4.3 are devoted to presents the results obtained from the three studies conducted in Phase I answering the corresponding research questions RQ1, RQ2 and RQ3, respectively. To attain the 2nd research objective, Section 4.4 and 4.5 are dedicated for the results obtained from the two studies conducted in Phase II answering the corresponding RQ4 and RQ5, respectively. To conquer the 3rd research objective, Section 4.6, 4.7 and 4.8 presents the results obtained from the case study answering the corresponding research questions RQ6, RQ7 and RQ8, respectively. Section 4.9 summarizes the overall chapter.

4.1 Phase I: User Needs Associated to System's Flexibility

This study is to address the 1st research question.

RQ1: What are the issues with flexibility of Online Airline Reservation Systems, whether or not flexibility is one of the reasons for users not using such systems?

4.1.1 Descriptive and Reliability Analysis

72% of the online experienced ticket buyers found the systems to be consistent with

respect to the usage of terms and the position of messages on the screen. Among the experienced buyers, 71% agreed that the current online reservation systems facilitate learning through textual descriptions and are presented in a manner that is not confusing. 35% of the experienced users found online reservation systems were easy to use or handle with little need to read instructions. 61% of the participants agreed to have no difficulty in learning to operate the system.

Pearson correlation coefficients were computed in order to test the relationships between each factor and the usability of online reservation systems. Table 4.1 shows the average item scores and standard deviations within each of the four groups (FR, NFR, flexibility and required resources/skill set).

Factors	No. of Items	Mean per Factor	S.D.	Cronbach's α
Functional Requirements	9	1.6	0.7	0.74
Non-Functional Requirements	20	2.8	1.0	0.76
Consistency	5	2.2	0.9	0.90
Ease of Use	5	3.2	0.9	0.62
Learnability	4	3.8	1.1	0.93
Security	3	3.4	0.8	0.63
Trust	3	3.6	0.9	0.71
Flexibility	11	2.8	0.7	0.81
Satisfaction	4	3.2	0.8	0.74
User-guidance	4	3.2	0.8	0.76
Simplicity	3	2.3	1.0	0.82
Required Resources & Skill Se	t 5	3.1	0.6	0.66

Table 4.1: Coefficient of the Factors Affecting Usability of Online Systems

The largest group-wise score (3.1 with a standard deviation of 0.6) was attained for Required Resources and Skills. Even though this is almost the mean of the fivepoint Likert scale, this indicates that this factor was of high preference among customers. Ranked next were Flexibility (2.8 with a standard deviation of 0.7) and the other NFRs (2.8 with a standard deviation of 1.0), respectively. This close ranking suggests that a notable preference of customers concerning these two aspects. The lowest item score is that of the FRs (1.6 with a standard deviation of 0.7), indicating that the extent to which online reservation systems may provide incorporation of additional support features such as online cancellation, online modification, online transfer and online reservation, remains uncertain as per the perception of the customers.

Cronbach's α analysis was performed to assess the reliability of the variables used in the research constructs. Reliability analysis was computed on each section/group measuring different attributes associated with the hypothesized research constructs.

- Flexibility: This factor gained the highest Cronbach's α score of 0.81
- NFR: This factor gained the second highest Cronbach's α score of 0.76
- FR: This factor gained the third highest Cronbach's α score of 0.74
- Required Resources and Skills: This factor gained Cronbach's α score of 0.66

4.1.2 Hypothesis Testing H₁

H₁: Non-functional Requirements are perceived to have an impact on the usability of OARS.

The factor of NFR was differentiated into separate dimensions to capture the customer's perception of consistency, ease of use, learnability, security, and trust. Table 4.1 shows that the construct "Learnability" among the NFRs had the highest mean of 3.8 and a standard deviation of 1.1, which may indicate an inclination of the users toward learning how to handle advanced features of online reservation systems. Moreover, this construct also has the highest Cronbach's α score of 0.93. The items in the Learnability construct were related to the satisfaction of users with reading text on the screen, the sequence of screens, the organization of information and supportive information such as online help, messages, and documentation provided by the systems before, during, and after completing the tasks. This is followed in rank by the extended construct of "Trust" with a mean of 3.6 and a standard deviation of 0.9. The extended construct of Trust was related to the risk in providing personal accounts information online and the reliability on experienced travel agents in finding better flights and packages. The survey indicates that customers bestowed upon traveling agents greater trust and reliability believing they could help in finding them better

traveling packages. This construct has the third highest Cronbach's α score of 0.71. The extended construct of "Consistency" has the second highest Cronbach's α score of 0.90. The questions asked with regards to this construct included various aspects of how information is presented in the interface and used by customers of online reservation systems. The items in the Consistency construct addressed finding the consistency in use of terms throughout the system, the positioning of messages on the screeens, organization of screens, and the use of text-based instructions.

The correlation between NFR and the usage of the systems was observed to be r = 0.7 with p<0.01. This shows the significance of the relationship; the *hypothesis is accepted*.

4.1.3 Hypothesis Testing H₂

H₂: Functional Requirements are perceived to have an impact on the usability of OARS.

Table 4.2 presents the flexibility of existing systems. 66% of the experienced users stated that they never tried to make online changes in their traveling schedule. 15% of the respondents reported to have been successful, 11% reported to have been unsuccessful, and the remaining 8% reported to have not seen such an option in existing online systems. 64% of the respondents claimed that they never tried online cancellation. 8% reported to have been successful in making cancellation changes, 16% reported to have been unsuccessful, whereas the remaining 12% reported to have not seen such an option in the systems. 80% of the respondents claimed that they never tried an online transfer of a ticket. 1% reported to have been successful, 11% reported to have been unsuccessful, whereas the remaining 8% reported to have not seen such an option in the system. 72% of the respondents answered that they never tried online correction of errors. 9% reported to have been successful, 15% reported to have been unsuccessful, whereas the remaining 4% reported to have not seen such an option in the system. Lastly, 58% of the experienced users reported that they never tried reservations in online systems. Interestingly, no one reported to be successful, whereas the remaining 15% reported to have not seen such an option in the system.
Flexibility of Existing Systems	Never	Unable to do so	Option not available	Successful
Have you ever tried to make changes in your traveling dates online?	66%	11%	8%	15%
Have you ever tried to cancel your ticket online?	64%	16%	12%	8%
Have you ever tried to transfer your ticket to someone else online?	80%	11%	8%	1%
Have you ever tried to correct typos errors online?	72%	15%	4%	9%
Have you ever tried to reserve a ticket for few days with the intension to buy it later?	58%	27%	15%	0%

Table 4.2: Flexibility of Existing Systems

The relationship between FR and the usage of the systems was observed to have r = 0.18 with p>0.01. This shows that the relationship is not significant; the *hypothesis* is rejected.

4.1.4 Hypothesis Testing H₃

H₃: The perceived flexibility of OARS affects the usability of such systems.

The factor of flexibility was investigated from different viewpoints to examine the perception of the customers concerning satisfaction, user guidance, and simplicity of use. 45% agreed that the data entry was flexible, 35% perceived flexible user guidance, and 53% saw clear indications for completing the process. 40% claimed to be satisfied with the total number of steps required to accomplish the task. Furthermore, 48% were satisfied with the ease of completing the tasks and 41% with the total time systems take to complete the tasks. 30% of the respondents agreed that the online systems support quick and easy recovery from mistakes, while 32% of the respondents did not agree. The remaining 38% of the respondents remained neutral to this. 30% of the respondents agreed that the current online reservation systems provide effective linkages with other travel-related partners (e.g. links to other airline reservation system in case of connected flights). 25% did not agree, whereas the remaining 45% remained neutral on this. 67% of the respondents claimed to have

been uncertain about how to fix errors if these occur; 85% of the respondents preferred travel agents to make changes in their flight schedule. 72% of the respondents preferred travel agents for reserving tickets. 51% of the inexperienced respondents refrained from online shopping so as to avoid making online payments. 51% of the respondents considered travel agents to be more reliable. Two extended constructs of the "Flexibility" factor, namely, "User-guidance" and "Satisfaction" have the highest mean score of 3.2. The reliability of both the constructs is also observed to be good with a Cronbach's α score of 0.76 and 0.74, respectively.

The relationship between perceived flexibility of online reservation systems and the usage of such systems was observed to be r = 0.69 with p<0.01. This shows a high significance of the relationship; the *hypothesis is accepted*.

4.1.5 Hypothesis Testing H₄

H₄: Functional Requirements of OARS are inversely associated with the flexibility of the systems.

The relationship between FR of online reservation systems and the flexibility of the systems was observed to have r = 0.28 with p<0.01. This shows that the relationship is mildly significant; the *hypothesis is accepted*.

4.1.6 Hypothesis Testing H₅

H₅: The availability of resources and skills set influence upon the usability of OARS.

10% of the respondents claimed to have no Internet connection available; 30% claimed to have no or little knowledge of online reservation systems; the remaining 60% reported to have no credit or debit card available.

The relationship between available resources and the usage of online reservation systems is observed to have r = 0.32 with p<0.01. This shows that the relationship is mildly significant; the *hypothesis is accepted*.

4.2 Phase I: Users' Flexibility in Terms of Compromising on SQAs

This study is to address the 2nd research question.

RQ2: To what extend flexible users can compromise with service quality attributes of Online Airline Reservation Systems?

4.2.1 Assumptions in ANOVA to Test H₆

Before performing ANOVA, a very basic assumption of ANOVA was checked, i.e. absence of outliers. Box-plot of the sample distribution was examined since it is a useful standard in data interpretation, reveals data symmetry, skewness and the presence of outliers. Moreover, it also facilitates in comparing more than one population without knowing anything about the underlying statistical distributions of those populations.

4.2.1.1 Box-and-Whisker Plot (H₆)

In case of satisfaction level with existing SBTs, respondents who reported that they 'can compromise' on SQAs of the airline have a median at 3 (black line) as shown in Figure 4.1.



Users' flexibility in compromising on service quality attributes

Figure 4.1: Box Plot showing Satisfaction Level with Existing SBTs

This represents neutral satisfaction level and at the same time indicates 50% of the data is greater than this value. Users' with any lesser satisfaction with existing SBTs are represented everything above median black line, while the users with higher satisfaction are represented everything below median black line. As shown by the top 'whisker', this group has greatest values but no outliers. Hence the data is normally distributed.

In case of satisfaction level with existing SBTs, respondents who reported that they 'may compromise' on SQAs of the airline have a median at 2 (black line). This represents high satisfaction level and at the same time indicates 50% of the data is greater than this value. Users' with any lesser satisfaction with existing SBTs are represented everything above median black line, while the users with higher satisfaction are represented everything below median black line. As shown by the top 'whisker', this group has greatest values and an outlier. The majority of the data is normally distributed.

In case of satisfaction level with existing SBTs, respondents who reported that they 'cannot compromise' on SQAs of the airline have a median at 2 (black line). This represents high satisfaction level and at the same time indicates 50% of the data is greater than this value. Users' with any lesser satisfaction with existing SBTs are represented everything above median black line, while the users with higher satisfaction are represented everything below median black line. As shown by the top 'whisker', this group has greatest values but no outliers. Hence the data is normally distributed.

4.2.1.2 Means Plot (H₆)

The means plot as shown in Figure 4.2 shows that there is apparently an enormous difference between the satisfaction level of the three respondents groups, which appears not be the actual case. Therefore as a follow-up, the same results will be analyzed in a different chart to see the difference between the groups.

Satisfaction level with existing Self Booking Tools (SBTs)



Figure 4.2: Means Plot on Satisfaction Level with Existing SBTs

4.2.1.3 T-Test (H₆)

In this case the three groups are significantly different using a t-test (t=36.760, df=169, p=0.000) as shown in Table 4.3. 95% Confidence Interval (CI) is probability that the interval contains the true mean.

		Test Value = 0							
	t	df	Sig. (2- tailed)	Mean Difference	95% Confid of the I	dence Interval Difference			
How flexible you are	36.760	169	.000	2.07	Lower 1.96	Upper 2.18			

Table 4.3: T-Test on User's Flexibility with SBTs

4.2.1.4 Error Bars (H₆)

The same results are now reproduced in the error bars, with 95% confidence intervals to have an idea of the variation in sample distribution. CI of the groups is closely related to the results of the analysis of variance for these groups. The confidence interval for each graph below shows a linear pattern of the sample distribution which otherwise appeared to be showing huge variations in the simple means plot.

In error bars we intend to see if the mean of one group is included in the confidence interval of the other two groups - if so then there is likely no difference among the groups. Moreover, it is not relevant whether the error bars 'overlap' but whether the mean of one group 'overlaps' with the error bars of the other. The confidence intervals can overlap by as much as 25 percent of their total length and still show a significant difference between the means for each group.



Users' flexibility in compromising on service quality attributes

Figure 4.3: Error Bar on Satisfaction Level with Existing SBTs

In Figure 4.3, 95% CI tells us that the satisfaction level of existing SBTs for the users who "can compromise" on SQAs of the airline is probably between 2.7 and 3.35, with group mean of 3. Likewise, for users who "may compromise" it is probably between 2.4 and 2.83, with group mean of 2.6, and for users who "cannot compromise" it is probably between 1.8 and 2.3, with group mean of 2.07.

The group means of users' who 'may compromise' shares a degree of confidence interval overlap with users who 'can compromise', thus the two groups may not necessarily be different from one another. Moreover, the group mean of users' who 'cannot compromise' does not share any degree of confidence interval overlap with either of the two groups, therefore, this particular group appears to be significantly different from the rest of the sample population. However, post-hoc tests can confirm this.

4.2.2 Hypothesis Testing H₆

H₆: The level of satisfaction with existing SBTs is different for respondents with different attitudes towards Users' Flexibility in compromising on SQAs of the airline.

4.2.2.1 One-way Analysis of Variance (H₆)

To test the hypothesis, one-way analysis of variance was used to determine the satisfaction mean of users with the existing self-booking tools of the airline and at the same time report their flexibility level in terms of compromising on SQAs of the airline as shown in Table 4.4.

<u> </u>	Sum of Squares	df	Mean Square	F	Sig.
Between Groups	20.707	2	10.354	11.250	.000
Within Groups	153.699	167	.920		
Total	174.406	169			

 Table 4.4: One-Way ANOVA on Satisfaction Level with Existing SBTs

The respondents had to select from given three options of, (1) Can compromise on SQAs, (2) May compromise on SQAs, (3) Cannot compromise on SQAs. The analysis showed significant differences among satisfaction mean of the three user groups with existing self booking tools of the airline (F (2,169) = 11.250, p < .001).

The respondents who indicated their flexible attitude as "cannot compromise" on SQAs of the airline, depicted highest level of satisfaction with existing SBTs of the airline (M = 2.06, SD = .916). This was closely followed by satisfaction of the respondents who indicated their flexible attitude as "may compromise" on SQAs of the airline (M = 2.59, SD = .973). The respondents who reported their flexible attitude as "can compromise" on SQAs of the airline, depicted least level of satisfaction with existing SBTs of the airline (M = 3.00, SD = .987).

Since the three user groups differed significantly on satisfaction mean level of existing SBTs of the airline, null-hypothesis is rejected and alternative *hypothesis* H_6 *is accepted.*

4.2.2.2 Post-hoc Scheffe Tests(H₆)

Post-hoc Scheffé tests in Table 4.5 showed that there is a significant difference between the pair of means of the respondents who reported their flexible attitude as "Cannot compromise" on SQAs of the airline with those who "Can compromise"; p=.000(<.001). The same group of respondents also differed significantly from the group of respondents who reported their flexible attitude as "May compromise" on SQAs of the airline, p=.009 (<.01).

Dependent Variabl	e: Satisfaction with Exist	ing SBTs		_		
(I) How Flexible You Are	(J) How Flexible You Are	Mean Difference (I-J)	Std. Error	Sig.	95 Confi Inte	% dence erval
			_		Lower Bound	Upper Bound
Can Compromise	May Compromise	.41	.187	.092	05	.87
	Cannot Compromise	.94(*)	.202	.000	.44	1.44
May Compromise	Can Compromise	41	.187	.092	87	.05
	Cannot Compromise	.53(*)	.172	.009	.11	.96
Cannot Compromise	Can Compromise	94(*)	.202	.000	-1.44	44
	May Compromise	53(*)	.172	.009	96	<u>11</u>

 Table 4.5: Multiple Comparisons on Satisfaction Level with Existing SBTs

* The mean difference is significant at the .05 level.

4.2.2.3 Effect Size for One-way ANOVA(H₆)

From hypothesis testing it is clear that the three groups are different, but this does not confer the strength or the magnitude of this effect. Effect size is measure of the strength of an effect. And since the null-hypothesis has already been rejected, therefore it makes sense to calculate effect-size to determine the size of the effect. The size of the effect is 12% ($\eta^2 = 0.1187$).

4.2.3 Assumptions in ANOVA to Test H₇

Box-plot of the sample distribution was examined to meet the assumption of ANOVA, i.e. absence of outliers. Since it is a useful standard in data interpretation, reveals data symmetry, skewness and the presence of outliers. Moreover, it also facilitates in comparing more than one population without knowing anything about the underlying statistical distributions of those populations.

4.2.3.1 Box-and-Whisker Plot (H7)

In the case of satisfaction level with existing OTAs, respondents who reported that they 'can compromise' on SQAs of the airline have a median at 2 (black line) as shown in Figure 4.4. This represents high satisfaction level and at the same time indicates 50% of the data is greater than this value. Users' with any lesser satisfaction with existing OTAs are represented everything above median black line, while the users with higher satisfaction are represented everything below median black line. As shown by the top 'whisker', this group has greatest values but no outliers. Hence the data is normally distributed.



Users' flexibility in compromising on service quality attributes

Figure 4.4: Box Plot on Satisfaction Level with Existing OTAs

In the case of satisfaction level with existing OTAs, respondents who reported that they 'may compromise' on SQAs of the airline have a median at 2 (black line). This represents neutral satisfaction level and at the same time indicates 50% of the data is greater than this value. Users' with any lesser satisfaction with existing OTAs are represented everything above median black line, while the users with higher satisfaction are represented everything below median black line. As shown by the top 'whisker', this group has greatest values but no outliers. Hence the data is normally distributed.

In the case of satisfaction level with existing OTAs, respondents who reported that they 'cannot compromise' on SQAs of the airline have a median at 3 (black line). This represents high satisfaction level and at the same time indicates 50% of the data is greater than this value. Users' with any lesser satisfaction with existing OTAs are represented everything above median black line, while the users with higher satisfaction are represented everything below median black line. As shown by the top 'whisker', this group has greatest values but no outliers. Hence the data is normally distributed.

4.2.3.2 Means Plot (H₇)

The means plot is shown in Figure 4.5.



User's flexibility in compromising on service quality attributes

Figure 4.5: Means Plot on Satisfaction Level with Existing OTAs

The means plot shows that their apparently enormous difference between the satisfaction level of the three respondents groups, which may appear not to be actual case. Therefore as a follow-up and to backup this, we will analyze same results in a different chart to see the difference between the groups.

4.2.3.3 T-Test (H₇)

In this case the three groups are significantly different using a t-test (t =35.509, df = 169, p = 0.000) as shown in Table 4.6. 95% Confidence interval is probability that the interval contains the true mean. CI of the groups is closely related to the results of the analysis of variance for these groups. The confidence interval for each graph below shows a linear pattern of the sample distribution which otherwise appeared to be showing huge variations in the means plot.

Table 4.6: T-Test on User's Flexibility with Existing OTAs

	Test Value = 0									
	t	df	Sig. (2- tailed)	Mean Difference	95% Confidence Differ	e Interval of the rence				
How flexible you are	35.509	169	.000	2.14	Lower 2.02	Upper 2.25				

4.2.3.4 Error Bars (H₇)

The same results are now reproduced in the error bars as shown in Figure 4.6, with 95% confidence intervals to have an idea of the variation in sample distribution.

In error bars we intend to see if the mean of one group is included in the confidence interval of the other two groups - if so then there is likely no difference among the groups. Moreover, it is not relevant whether the error bars 'overlap' but whether the mean of one group 'overlaps' with the error bars of the other. The confidence intervals can overlap by as much as 25 percent of their total length and still show a significant difference between the means for each group. Any more overlap and the results will not be significant.

Error Bar: Satisfaction Level with existing Online Travel Agencies



Users' flexibility in compromising on service quality attributes

Figure 4.6: Error Bar on Satisfaction Level with Existing OTAs

In Figure 4.6, 95% CI tells us that the satisfaction level of existing OTAs for users who "can compromise" on SQAs of the airline is probably between 1.98 and 2.42, with group mean of 2.2. Likewise, for users who "may compromise" it is probably between 2.03 and 2.44, with group mean of 2.28, and for users who "cannot compromise" it is probably between 2.3 and 2.8, with group mean of 2.6.

The group mean of users' who 'can compromise' shares a certain degree of confidence interval overlap with the error bars for users who 'may compromise', thus the two groups may not necessarily be different from one another. Moreover, the group mean of users' who 'cannot compromise' does not share any degree of confidence interval overlap with either of the two groups, therefore, this particular group appears to be significantly different from the rest of the sample population. However, only with our post-hoc tests, this can be confirmed.

4.2.4 Hypothesis Testing H₇

 H_7 : The level of satisfaction with existing OTAs will be different for respondents with different attitudes towards Users' Flexibility in compromising on SQAs of the airline.

To test this hypothesis, one-way analysis of variance was used to determine satisfaction mean of users with existing online travel agencies of the airline and at the same time report their flexibility level in terms of compromising on SQAs of the airline. The respondents had to select from given three options of, (1) Can compromise on SQAs, (2) May compromise on SQAs, (3) Cannot compromise on SQAs as shown in Table 4.7.

The analysis showed significant differences among satisfaction mean of the three user groups with existing online travel agencies (F (2,169) = 6.728, p = .002 < .01).

-	Sum of Squares	df	Mean Square	F	Sig.
Between Groups	9.429	2	4.715	6.728	.002
Within Groups	117.018	167	.701		
Total	126.447	169			

Table 4.7: ANOVA on Satisfaction Level with Existing OTAs

The respondents who indicated their flexible attitude as "can compromise" on SQAs of the airline, depicted highest level of satisfaction with existing online travel agencies (M = 2, SD = .625). This was closely followed by satisfaction of the respondents who indicated their flexible attitude as "may compromise" on SQAs of the airline (M = 2.17, SD = .794). The respondents who reported their flexible attitude as "cannot compromise" on SQAs of the airline, depicted least level of satisfaction with existing online travel agencies (M = 2.57, SD = .984). Since the three user groups differed significantly on satisfaction mean level of existing online travel agencies, therefore, *hypothesis* H_7 *is accepted*.

4.2.4.2 Post-hoc Scheffe Tests (H₇)

Post-hoc Scheffé tests in Table 4.8 showed that there is a significant difference between the pair of means of the respondents who reported their flexible attitude as "Cannot compromise" on SQAs of the airline with those who "Can compromise"; p = .000 (< .001). The same group of respondents also differed significantly from the

group of respondents who reported their flexible attitude as "May compromise" on SQAs of the airline, p=.031 (< .05).

(I) how flexible you are	(J) how flexible you are	how flexible Mean you are Differenc Error e (I-J)		Sig.	95% Confidence Interval		
				<u>.</u>	Lower Bound	Upper Bound	
Can Compromise	May Compromise	17	.167	.579	59	.24	
	Cannot Compromise	57(*)	.166	.003	98	16	
May Compromise	Can Compromise	.17	.167	.579	24	.59	
	Cannot Compromise	39(*)	.148	.031	76	03	
Cannot Compromise	Can Compromise	.57(*)	.166	.003	.16	.98	
	May Compromise	.39(*)	.148	.031	.03	.76	

Table 4.8: Multiple Comparisons on Satisfaction Level with Existing OTAs

 Dependent Variable: Satisfaction with Existing OTAs

* The mean difference is significant at the .05 level.

4.2.4.3 Effect Size for One-way ANOVA

From our hypothesis testing we know that the three groups are different, but this does not confer the strength or the magnitude of this effect. Effect size is measure of the strength of an effect. And since we have already rejected the null-hypothesis in the both of the above cases, therefore it makes sense to calculate effect-size to determine the size of the effect. The size of the effect is 7.5% ($\eta^2 = 0.0745$).

4.3 Phase I: Users' Satisfaction with SBTs against their rated OTA Feature

This study is to address the 3rd research question.

RQ3: How users' satisfaction with existing SBTs of airlines is rated against their choice of OTA feature and reflected in their integration assessment of the same for making SBTs more FOARS?

4.3.1 Hypothesis Testing H₈

H₈: Users' satisfaction with existing SBTs will be different across their choice of four OTA features for making SBTs more FOARS.

4.3.1.1 Means Plot

The means plot as shown in Figure 4.7 illustrates two lines, red line indicating respondents who consider integration of OTA features into SBTs may not necessarily make them FOARS, while the green one denotes the respondents who think otherwise. Means plot showed that out of the four OTA features investigated in this study, user satisfaction was highest for opaque fare and hotel search facility. Among the two, opaque fare was highly recommended due to being users' absolute satisfaction point.



Figure 4.7: Means Plot on the Recommendation of Integrating OTA Features

Out of the four OTA features investigated in this study, opaque fare and hotel search are the only two OTA features for which the green line is lower than the red line (low value indicates high satisfaction). However, among the two, opaque fare is the most recommended OTA feature for making SBTs flexible, since it reflects absolute highest satisfaction point of the respondents, who thought integration of OTA features into SBTs will make them more FOARS and also among respondents who think otherwise, because it has been considered as the second most important feature for integration, only after Matrix Display.

			Matrix Display	Opaque Fares	Alternative Airport Search	Hotel Search Facility
Do you think	Yes, very	Mean	2.25	1.60	3.00	3.40
OTA features	sure about it.	S. D.	.500	.548	.707	.548
in to existing SBTs will make them more FOARS?	Not so	Mean	2.00	2.06	2.83	3.67
	very sure about it.	S. D.	.707	.827	.753	.577

Table 4.9: Mean Score and SD on the Recommendation of Integrating OTA Features

4.3.1.2 Two-ways Analysis of Variance

A two-way analysis of variance tested the satisfaction level of the representatives with existing OARS and also reported if the integration of OTA features into SBTs would make them more FOARS, and also picked their recommended OTA feature for integration into SBTs. The three different F-tests as shown in Table 4.10 are:

- The first one is the mean satisfaction level different across four proposed OTA features for SBTs, controlling for that effect of sharing, if the chosen OTA feature will make existing SBTs more flexible. The difference in satisfaction level has been found to be statistically significant at p<0.01.
- The second F-test looks at whether respondents who reported that integration of OTA features into SBTs will make them flexible reservation systems, do or do not have different levels of satisfaction with existing OARS, and again the results were significant at p<0.05
- 3. The third F-test examines the interaction effect of the four proposed OTA features and their integration into SBTs for making them FOARS. The finding was significant at p<0.05, suggesting that some combination of OTA features and existing SBTs are related and can influence upon one another, especially in terms of making them more flexible.

Dependent Variable: Satisfaction Level with Existing OARS (SBTs)								
Source	Type III Sum of Squares df		Mean Square	F	Sig.			
Corrected Model	18.729(a)	7	2.676	5.205	.000			
Intercept	269.145	1	269.145	523.552	.000			
Recommended OTA feature for SBTs	18.056	3	6.019	11.708	.000			
Flexible SBTs (Yes/No)	1.827	1	1.827	3.556	.004			
Flexible SBTs * Recommended OTA feature for SBTs	1.012	3	.337	.656	.042			
Error	21.591	42	.514		-			
Total	338.000	50						
Corrected Total	40.320	49						
a R Squared = .465 (Adjusted R Squared	= .375)							

Table 4.10: Two-way ANOVA on the Satisfaction Level with Existing OARSs

The three different F-tests in the two way analysis of variance are discussed as under:

- 1. Respondents, who indicated that integration of OTA's feature into existing SBTs will make them FOARS, shared the highest level of satisfaction when opted for opaque fare as a recommended OTA feature for SBTs. However, respondents who indicated integration of OTA features into existing SBTs may not make them flexible, reported higher level of satisfaction when opted for Matrix Display and Alternate Airport Search as a recommend solution for SBTs (F = 11.708, p = .000 < 0.01).
- Satisfaction level with existing OARS differed significantly (F = 3.556, p = .004 < 0.05) across respondents who indicated whether or not integration of OTA features would make the existing systems more flexible.
- 3. The interaction effect of the four proposed OTA features and their integration into SBTs for making them FOARS was also significant (F = .656, p = .042 < 0.05).

F-Test for respondents, who indicated that integration of OTA's feature into existing SBTs will make them flexible Online Airline Reservation Systems, shared the highest level of satisfaction when opted for opaque fare as a recommended OTA feature for SBTs (F = 11.708, p = .000 < 0.01). Since the satisfaction level of the users differed significantly, therefore, *hypothesis is accepted*.

4.4 Phase II: Users' Perception on Flexible Traveling Behavior

This study is to address the 4th research question.

RQ4: How users' perception on factors influencing flexible traveling behavior and FOARS is determined?

4.4.1 Qualitative Analysis

The demographics of the 14 respondents from travelblog forum and 17 from travellerspoint forum are shown in Table 4.11. The average ages of the 14 male and female respondents in case of Travelblog forum were recorded 37 and 35, respectively. Likewise, the average ages of the 17 male and female respondents in case of Travellerspoint forum were recorded 41 and 37, respectively.

	Travelblog (14)	Travellerspoint (17)
Male	9	10
Female	5	7
Average Age (Male)	37	41
Average Age (Female)	35	37
Countries of Origin	USA, China, Singapore, UK, Kenya, India, Peru, Indonesia, Spain, Thailand, Nepal, Romania, Canada, South Africa	USA, Japan, Malaysia, China, Singapore, Thailand, Australia, Indonesia
Rejected cases of ambiguous respondents	5	7

Table 4.11: Demographics of Online Travel Forums (OTF) Respondents

The responses received through online discussion forums were analyzed first as shown in Table 4.12 in order to investigate the factors influencing on users' flexible traveling behavior and factors influencing upon users perception of a flexible online airline reservation system.

	Reasons/Factors
Which factors influence on users' flexible traveling behavior?	 Promotional packages offered by a particular airline Searching skills to look for best offers and deals online Budgetary constraints and leverages Traveling comfort in services offered Traveling purpose Flying frequency Family and friends
Which factors influence upon your perception of a flexible online reservation system?	 Simplicity Easiness Multiple options

Table 4.12: Results of Online Travel Forums

After data analysis of online travel forum discussion, the semi structured in-depth interviews were conducted; the demographics are summarized in Table 4.13.

No. of respondents (as per same Country of Origin)	Country of Origin	Gender A (A		Age (Avg.)	Flying Frequency / Year (Avg.)	Purchase Experience with SBTs of Airline	Purchase Experience with any popular OTA
6	Malaysia	Male	_4	34	Frequent	Vec	Ves
		Female	_2	30	(Twice a year)	103	
4	USA	Male	2	40	Very Frequent	Ves	Ves
т		Female	2	37	(more than twice a year)	105	
3	UK	Male	2	29	Very Frequent	Ves	Ves
		Female	1	26	(more than twice a year)	105	
2	China	Male	_1	36	Very Frequent	Vec	Ves
ے۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔۔		Female	1	24	(more than twice a year)	103	105
2	India	Male	2	33	Frequent	Vec	Ves
<u> </u>		Female	_0	-	(Twice a year)		
2	Pakistan	Male	2	43	Frequent	Yes	Ves
		Female	0	-	(Twice a year)		
2	Singapore	Male	0	-	Very Frequent	Yes	Ves
<u></u>		Female	2	27	(more than twice a year)		
2	Ireland	Male	_ 1	25	Frequent	Vec	Yes
<u>ل</u>	neianu	Female	1	20	(Twice a year)	105	
2	Thailand	Male	1	31	Frequent	Vac	Yes
<u> </u>	mananu	Female	1	29	(Twice a year)	105	
1	Austrolio	Male	1	44	Very Frequent	Vac	Vec
1	Australia	Female	0		(more than twice a year)	105	105
1	Sudan	Male	1	35	Not Very Frequent	Vec	Vec
1	Juuan	Female	_ 0 _	-	(Once a year)	105	105
1	Mouritine	Male	1	29	Frequent	Vac	 Vec
l		Female	0		(Twice a year)	1 05	1 05

 Table 4.13: Demographics of In-depth (ID) Interview

The responses received through in-depth interviews were analyzed in the second phase. Table 4.14 provides additional information on users' flexible traveling behavior.

	Reasons
Which factors influence on users' flexible traveling behavior?	 Airlines repute Standard conscious Promotional packages offered by a particular airline Searching skills to look for best offers and holiday deals Customer loyalty Traveling comfort in services offered Supporter of Green Environment Patriotism Traveling purpose Flying frequency Budgetary constraints and leverages Occupation Children holidays Traveling mileage Interest in recreation, leisure, and tourism
Which factors influence upon your perception of a flexible online reservation system?	 Provides alternative dates for flying Allows self-adjustments in itinerary Simplicity Easiness Multi-linguistic

Table 4.14: Results of In-depth Interviews

After data analysis of semi structured in-depth interview, focus group interviews were conducted, the demographics are shown in Table 4.15.

Table 4.15: Demographics of Focus Group (FG) Interviews

Meeting With Independent Airline Reservation Offices						
Airlines	No. of Manager	No. of Executives interviewed	No. of Technical Experts			
Malaysian Airline	1	2	1			
Fire Fly	1	2	1			
Aero Asia	1	1	1			

	Reasons
Which factors influence on users' flexible traveling behavior?	 Peak / off travel time Discounted versus Normal Fare Traveling mileage Traveling purpose Working hours Traveling comfort in services offered Children holidays Recreation, leisure, and tourism Promotional Schemes Airlines repute
Which factors influence upon your perception of a flexible online reservation system?	 Easiness Product presentation Post sale features User prompting for their guidance throughout reservation process Matrix display to sort airfares on different dates and destinations Low fare notifications Flexible and alternative date search Dynamic packaging Hotel search display, sort and reservation

Table 4.16: Results of Focus Group

The following 6 themes emerged under factors influencing upon flexible traveling behavior and 3 themes emerged under factors influencing upon perceived flexibility of a reservation system after giving much thought process to results, reviewing literature, discussion with qualitative researchers:

- Theme 1: Travelers' Flexible Behavior is molded by their traveling consciousness.
- Theme 2: Travelers' Flexible Behavior is molded by their belief that they have the required digital skills.
- Theme 3: Travelers' Flexible Behavior is molded by their self-belief as flexible travelers.
- Theme 4: Travelers' Flexible Behavior is molded by societal influences.
- Theme 5: Travelers' Flexible Behavior is molded by how they attribute a cause to their traveling behavior.
- Theme 6: Travelers' Flexible Behavior is molded by their prior traveling experiences.

- Theme 7: Systems Perceived Flexibility is influenced by its perceived usability.
- Theme 8: Systems Perceived Flexibility is influenced by end-user support.
- Theme 9: Systems Perceived Flexibility is influenced by comparison of features on the actual level of effect regarding to complete the reservation process.

4.5 Phase II: Classification of Users on the Basis of Their Flexible Traveling Behavior

This study is to address the 5th research question.

RQ5: How to classify Users' on the basis of their Flexible Traveling Behavior into High, Medium and Low flexible and how to investigate interrelationships among System's Flexibility, Users' Flexibility and Perceived Usability of existing OARS?

4.5.1 Hypotheses Testing H₉

H₉: Users can be classified on the basis of their flexible traveling behavior.

Users' Flexibility Transforming Scale – The transforming scoring scale was accordingly adapted in this study as discussed in Chapter 3 to meet the requirements of item discrimination. The same results in adapted table are shown in Table 4.17.

Users' Flexibility Rating on Service	Highest	High	Neutral	Low	Least
Quality Attributes of Airlines	1	2	3	4	5
Scale	-2	-1	0	+1	+2
Users' Rating Frequency	1	1	4	3	1
Product (Users' Rating Frequency * Scale)	-2	-1	0	3	2
Users' Flexibility Score (UFS)			2	·	·

 Table 4.17: Users' Flexibility Transforming Scoring Scale

The participants scoring on Section 2 of the questionnaire (Appendix D) was transformed using Table 4.17, in order to obtain their unique Users' Flexibility Score. In total, ninety (90) Users' Flexibility Score was recorded and transformed, of which

after extensive filtration of results, 62 cases were retained of those Users' only who had adequately respondents in all 4 sections of the questionnaire.

Range: From our sample data set, the range to classify respondents on the basis of their Users' Flexibility Score is $2 \sim +19$ as shown in Table 4.18. This range has 18 digits in between inclusive of the extreme two ends. So if we divide this range approximately among three groups, we get the following classifications to be assigned to users on the basis of their flexibility score.

Table 4.18: Range for Classification of Users' Flexible Behavior on the basis of their unique Users' Flexibility Score (UFS)

	Classification	Range
1	Low Flexible Behavior	+02 to $+07$
2	Medium Flexible Behavior	+08 to +13
3	High Flexible Behavior	+ 14 to + 19

Based on classification range identified from data distribution and transformation scale adapted from literature users' classifications were made as with High, Medium and Low flexible behavior on the basis of Users' Flexibility Score. Users' with High Flexible Behavior were assigned code 3, Medium Flexible Behavior were assigned code 2, and Low Flexible Behavior were assigned code 1 (Appendix E). Figure 4.8 shows the classification of users on the basis of their Users' Flexibility Scores.



Figure 4.8: Data Distribution after Classifications of Users'

Users' with high (denoted by white line), medium (denoted by green) and low (denoted by black) flexibility in their traveling behavior have different levels of satisfaction. Users with high, medium and low Users' Flexibility Score have different level of Perceived Usability and therefore the hypothesis H_9 is accepted.

4.5.2 Hypotheses Testing H₁₀

H₁₀: User's Flexible Behavior and their Perceived Usability is correlated.

The interrelationship between Users' Flexible Behavior and their Perceived Usability of FOARS is shown in Table 4.19.

Kendall's T	au_b	UFB	PU
Users' Flexible Behavior (UFB)	Correlation Coefficient Sig. (2-tailed)	1.000	.381(*) .049
Users' Flexible Behavior (UFB) Perceived Usability of FOARS (PU)	N	62	62
Perceived Usability of FOARS (PU)	Correlation Coefficient Sig. (2-tailed) N	.381(*) .049 62	1.000 62

* Correlation is significant at the 0.05 level (2-tailed)

Negative Association – Users' Flexible Behavior (UFB) is positively associated with Perceived Usability of FOARS, correlation coefficient r = 0.381, is significant at p<0.05. This means that as one variable increases in value, the second variable also increases in value. This is called a positive correlation. The significance value indicates that the relationship is genuine, hence H_{10} is confirmed.

4.5.3 Hypotheses Testing H₁₁

H₁₁: User's Flexible Behavior and System's Flexibility is correlated.

In order to investigate the interrelationship between Users' Flexible Behavior and System's Flexibility, Pearson's Correlation Coefficients was calculated. Table 4.20 provides a matrix of correlation coefficients for the five variables, (1) Users' Flexible Behavior, (2) System's Flexibility, (3) System's Adaptability, (4) Systems' Adaptivity, (5) Systems' Personalization. It also displays a matrix of significance values for these coefficients.

- Users' Flexible Behavior share a statistically non-significant relationship with System's Flexibility, r = 0.255, p = 0.056 > 0.05.
- Users' Flexible Behavior share a statistically significant relationship with System's Adaptability, r = 0.372, p = 0.000 < 0.001.
- Users' Flexible Behavior share a statistically non-significant relationship with System's Adaptivity, r = 0.151, p = 0.146 > 0.05.
- Users' Flexible Behavior share a statistically significant relationship with System's Personalization, r = 0.314, p = 0.042 < 0.05.
- System's Flexibility share a statistically significant relationship with System's Adaptability, r = 0.560, p = 0.000 < 0.001.
- System's Flexibility share a statistically significant relationship with System's Adaptivity, r = 0.344, p = 0.045 < 0.05.
- System's Flexibility share a statistically significant relationship with System's Personalization, r = 0.222, p = 0.032 < 0.05.
- System's Adaptability share a statistically significant relationship with System's Adaptivity, r = 0.342, p = 0.000 < 0.001.
- System's Adaptability share a statistically significant relationship with System's Personalization, r = 0.326, p = 0.016 < 0.05.
- System's Adaptivity share a statistically non-significant relationship with System's Personalization, r = 0.153, p = 0.074 > 0.05.

The results showed that users' FTB to have a non-significant correlation with System's Flexibility (r = 0.255, p = 0.056 > 0.05). Likewise, Adaptivity (sub-variables of System's Flexibility) also shared a non-significant correlation with users' FTB (r = 0.151, p = 0.146 > 0.05). However, Adaptability (a sub-variable of System's Flexibility) shared a strong positive and significant correlation with users' FTB (r = 0.372, p = 0.000 < 0.001). When the same results were interpreted using r^2 , Adaptability accounted for 26% of the variability in users' FTB.

		Users' Flexible Behavior	System's Flexibility	System's Adaptability	System's Adaptivity	System's Personalization
Pearson	Users' Flexible Behavior	1	0.255	0.372**	0.151	0.314*
Correlation	System's Flexibility	0.255	1	0.560**	0.344	0.222
	System's Adaptability	0.306**	0.560**	1	0.342**	0.326*
	System's Adaptivity	0.151	0.344	0.342**	1	0.153
	System's Personalization	0.177*	0.222	0.326*	0.153	1
Sig. (2-tailed)	Users' Flexible Behavior	-	0.056	0.000	.146	0.042
Sig. (2-tailed)	System's Flexibility	0.056	-	0.000	0.045	0.222
	System's Adaptability	0.000	0.000	-	0.000	0.016
	System's Adaptivity	.146	0.045	0.000	-	0.074
	System's Personalization	0.042	0.222	0.016	0.074	-
N	Users' Flexible Behavior	61	61	61	61	61
	System's Flexibility	61	61	61	61	61
	System's Adaptability	61	61	61	61	61
	System's Adaptivity	61	61	61	61	61
	System's Personalization	61	61	61	61	61

Table 4.20: Pearson Coefficient Correlations to Investigate the Interrelationship between UFB and System's Flexibility

The overall relationship between UFB and SF shows non-significant values. However, there are significant relationships between UFB and the two components of the SF i.e. System's Adaptability and System's Personalization. Therefore, the hypothesis H_{11} is partially accepted.

4.5.4 Hypotheses Testing H₁₂

 H_{12} : Perceived Usability of OARS is not affected by users' flexible behavior after adjusting for the effect of the covariate, System's Flexibility.

4.5.4.1 One-way Analysis of Variance (H_{12})

To meet the assumption of ANCOVA i.e. Independence of the covariate and treatment effect – one way independent ANOVA was performed on Perceived Usability as independent variable and covariate System's Flexibility as an outcome variable as shown in Table 4.21. This analysis should be non-significant to meet the assumption and result showed non-significant effect of Perceived Usability of FOARS on System's Flexibility.

Table 4.21: One Wa	y Independent	ANOVA with PU	and System's F	lexibility
--------------------	---------------	---------------	----------------	------------

	Sum of Squares	df	Mean Square	F	Sig.
Between Groups	39.616	4	9.904	2.476	.261
Within Groups	6.372	56	0.1138		
Total	45.988	60			

Sum of squares between groups for the corrected model is 6.093, which indicates total experimental effect while means square of the model is 3.047, which represents average experimental effect as shown in Table 4.22. Unexplained variance error is the sum of squares within groups; it is 8.128 and explains unsystematic variation within data. The test of whether the group means are the same is represented by the F-ratio for the combined between group effect. The value of F ratio is 22.114, which is significant with p = .000 < 0.001. It is therefore reported after conducting ANOVA that there was a significant effect of Users' Flexible Behavior on their Perceived Usability of FOARS, F (2, 59) = 22.114, p= 0.000 > 0.001.

Source	Type III Sum of Squares	df	Mean Square	F	Sig.
Corrected Model	6.093(a)	2	3.047	22.114	.000
Intercept	584.054	1	584.054	4239.555	.000
Users' Flexible Behavior	6.093	2	3.047	22.114	.000
Error	8.128	59	.138		
Total	894.345	62			
Corrected Total	14.221	61			

Table 4.22: ANOVA with PU and System's Flexibility

a. R Squared = .428 (Adjusted R Squared = .409)

Dependent Variable: Perceived Usability of FOARS

4.5.4.2 Analysis of Covariance (H12)

ANOVA results indicated that an important assumption of ANCOVA has not been violated. Therefore, ANCOVA was performed to first examine influence of independent or fixed factor (Users' Flexible Behavior) on dependent variable (Perceived Usability of FOARS) and then experiment was manipulated by introducing a covariate (System's Flexibility) as shown in Table 4.23.

Source	Type III Sum of Squares	df	Mean Square	F	Sig.
Corrected Model	6.487(a)	3	2.162	16.215	.000
Intercept	8.400	1	8.400	62.992	.000
Users' Flexible Behavior	.394	1	.394	2.952	.047
System's Flexibility	3.852	2	1.926	14.443	.000
Error	7.734	58	.133		
Total	894.345	62			
Corrected Total	14.221	61			
a. R Squared = .456 (Adjusted F	R Squared $= .428$)				

Table 4.23: ANCOVA by Introducing System's Flexibility as Covariate

Looking first at the significance value, it clear that covariate, i.e., System's Flexibility, significantly predicts Perceived Usability of FOARS at F(1, 58) = 2.952, p = 0.47 < 0.05. What is more interesting is that when the effect of System's Flexibility is added, the effect of Users' Flexible Behavior still remains significant (p = 0.000 < .001) towards predicting Perceived Usability of FOARS. The amount of variation accounted for by the model has increased to 6.487 units for the corrected model, of which System's Flexibility now accounts for 3.9 units. Most important, the amount of

variation or unexplained variance in Perceived Usability of FOARS that is accounted for by the covariate has reduced to 7.7 units from 8.1 units.

4.6 Phase III: Users' Flexibility is determined by SQAs and EVs

This study is to address the 6^{th} research question.

RQ6: How do Service Quality Attributes of airlines and External Variables jointly predict flexible behavior of travelers?

4.6.1 Hypothesis Testing H₁₃

H₁₃: Flexible behavior of travelers cannot be predicted by Service Quality Attributes and External Variables.

4.6.1.1 Scatter Plots (H₁₃)

To examine whether linear regression is appropriate, scatter plots as shown in Figure 4.9 and Figure 4.10 were examined of each independent variable against the predicting or the dependent variable. User's Flexibility were treated as dependent variable, SQAs and External Variables were treated as independent variable with N = 250.



Figure 4.9: Scatter Plot between Users' Flexibility and SQAs

The first scatter plot examines SQAs of the airline against users' flexible personality. Each of the points on scatter plot represents a particular observation from the data. The data appears to be normally distributed along with linear regression line and has no obvious outliers. A general trend or relationship between the two variables is also predictable.



Figure 4.10: Scatter Plot between Users' Flexibility and External Variables

The second scatter plot examines external variables against users' flexible personality. The data appears to be normally distributed along with linear regression line and has no obvious outliers. A general trend or relationship between the two variables is also predictable.

4.6.1.2 Pearson Correlation Coefficients (H₁₃)

Correlation of the two independent variables was also computed to determine their association with Users' Flexibility and to further ascertain their individual range and strength of association (see Table 4.24). SQAs share a strong positive correlation with Users' Flexibility, r = .792, highly significant at p < 0.001. Likewise, External Variables also share a strong positive correlation with Users' Flexibility, r = .792, highly significant at p < 0.001. Likewise, External Variables also share a strong positive correlation with Users' Flexibility, r = .795, highly significant at p < 0.001 level.

		Personality in terms of Flexibility?	SQA	External Variables
How do you rate	Pearson Correlation	1	.792**	.795**
your overall	Sig. (2-tailed)		.000	.000
personality in terms of flexibility?	N	250	250	250
SQA	Pearson Correlation	.792**	1	.881**
	Sig. (2-tailed)	.000		.000
	N	250	250	250
External Variables	Pearson Correlation	.795**	.881**	1
	Sig. (2-tailed)	.000	.000	
	N	250	250	250

Table 4.24: Correlations between Flexible Personality, SQA and External Variables

**. Correlation is significant at the 0.01 level (2-tailed).

4.6.1.3 Multiple Regression Analysis (H₁₃)

The value of Multiple Correlation Coefficient (R) between the two independent variables and Users' Flexibility is 0.818 as shown in Table 4.25. The maximum value of multiple correlation coefficients is 1, positive or negative and indicates correlation of all variables for predicting one single outcome, which in this case is 0.818, suggesting a strong relationship of the two independent variables with UF.

Table 4.25: Multiple Correlation Analysis between EV, SQA and Users' Flexibility

Model	R	R Square	Adjusted R Square	Std. Error of the Estimate
1	.818 ^a	.669	.667	.653
a. Predictors	: (Constant), Ext	ernal Variables (EV), SQA	

4.6.1.4 Analysis of Variance (H_{13})

Analysis of Variance tests whether the model is significantly better at predicting the outcome, than using the mean as a best guess. In this analysis, simultaneous test was performed as shown in Table 4.26 to examine, (i) if all of the coefficient values could be zero, and (ii) if they are all able to be zero that would mean that none of the independent variables have a relationship with the dependent variable (null hypothesis). And if null hypothesis is not rejected, it means the model is not useful, as none of the independent variables have a relationship with the dependent variable.

In alternative hypothesis, may be at least one of the independent variable's relationship with dependent variable will not be zero, indicating at least one of the coefficients values is not zero. This model has an F ratio = 250.121 which is highly significant at p <.001. This means that model significantly improves ability to determine users' flexible behavior; therefore, *null hypothesis is rejected* as at least one of the coefficient values is not zero.

Model	Sum of Squares	df	Mean Square	F	Sig.
Regression	213.499	2	106.749	250.121	.000 ^a
Residual	105.417	247	.427		
Total	318.916	249	<u></u>		

Table 4.26: ANOVA with External Variables and SQAs

4.6.1.5 Testing and Interpreting Model Coefficients (H₁₃)

Table 4.27 shows values of coefficients and t-tests.

	Un-sta				
Model	Coefficients		Coefficients	t	Sig.
	B	Std. Error	Beta	-	
1 (Constant)	.119	.240		.497	.620
SQAs (Flying Date Confirmation)	.011	.076	.012	.143	.886
SQAs (Flying Carrier Confirmation)	.073	.079	.074	.930	.353
SQAs (Flying Time Confirmation)	076	.072	091	-1.066	.288
SQAs (No. of Stop Over).	.017	.071	.018	.234	.815
SQAs (No. of connected flights)	.043	.080	.042	.534	.594
SQAs (Ticket Class)	.096	.064	.099	1.505	.134
SQAs (Seat Specification)	.003	.068	.003	.046	.964
SQAs (Last minute discounts)	.120	.066	.128	1.829	.043
SQAs (Confirmation of Origin and Destination Airport)	.095	.061	.113	1.572	.017
SQAs (Immediate Confirmation of Itinerary on purchase)	.106	.069	.119	1.542	.003
Attribution	.073	.067	.074	1.099	.273
Engagement	.171	.067	.166	2.548	.011
Persuasion	.071	.053	.264	1.335	.018
Identity	.108	.066	.125	1.638	.037
Self Efficacy	.090	.056	.070	1.619	.107

Table 4.27: Values of Coefficients an	and T-test
---------------------------------------	------------

a. Dependent Variable: How do you rate your overall personality in terms of flexibility?

4.7 Phase III: Perceived Usability with Existing and Proposed Systems

This study is to address the 7th research question.

RQ7: How does user Perceived Usability with the existing and the proposed system differs?

4.7.1 Hypothesis Testing H₁₄

Dependent Variable: PU - Effectiveness

H₁₄: User Perceived Usability with existing and proposed systems is different across the three groups.

4.7.1.1 Descriptive Statistics on Effectiveness (H₁₄)

Table 4.28 displays means, standard deviations and number of respondents in all classifications based on flexible traveling behavior.

Interface Design	Classification of Flexible Behavior	Mean	Std. Deviation	N
	Least Flexible	4.19	.452	30
Existing	Medium Flexible	3.19	.705	19
-	Highly Flexible	2.96	.850	76
Ť	Total	3.29	.905	125
<u>_</u>	Least Flexible	4.12	.467	30
Proposed	Medium Flexible	3.74	.653	19
_	Highly Flexible	4.03	.593	76
-	Total	4.01	.584	125
Total	Least Flexible	4.16	.457	60
	Medium Flexible	3.46	.725	38
	Highly Flexible	3.50	.904	152
	Total	3.65	.839	250

Table 4.28: Descriptive Statistics on Effectiveness

Respondents with least flexible behavior rated the existing (mean 4.19, S.D 0.452) and proposed systems (mean 4.12, S.D 0.467) relatively higher in terms of their effectiveness. Likewise, respondents with a highly flexible behavior, rated the existing systems lowest in terms of its effectiveness (mean 2.96, S.D 0.850), their rating for the proposed systems is relatively higher in terms of its effectiveness (mean

4.03, S.D 0.593). These means are useful in interpreting the direction of any effects that emerge in further analysis of variance.

4.7.1.2 Two-way Analysis of Variance on Effectiveness (H14)

There is a significant effect of user's classifications on the basis of flexible traveling behavior; since the F-ratio as shown in Table 4.29 is highly significant indicating that the users' high, medium and low flexible traveling behavior is significantly affected by the proposed and existing systems.

Table 4.29: Two-way ANOVA of Effectiveness

Source	Type III Sum of Squares	df	Mean Square	F	Sig.
Corrected Model	65.946 ^a	5	13.189	29.410	.000
Intercept	2493.216	1	2493.216	5559.568	.000
interfacedesign	11.941	1	11.941	26.627	.000
classification	20.260	2	10.130	22.588	.000
interfacedesign * classification	14.003	2	7.001	15.612	.000
Error	109.423	244	.448		
Total	3504.778	250			
Corrected Total	175.369	249			

Dependent Variable: PU - Effectiveness

a. R Squared = .376 (Adjusted R Squared = .363)

4.7.1.3 Error Bars on Effectiveness (H_{14})

Figure 4.11 shows that when the influence of existing and proposed system is ignored, the overall effectiveness of systems is very similar for users' with medium and high flexible behavior, as the means of these two groups are approximately equal. However the perceived effectiveness of the system for users' with least flexible behavior, not only differs from the other two groups, it remains also higher.



Figure 4.11: Effect of User Classification of Flexible Behavior on Perceived Effectiveness

When the effect of existing and proposed systems is examined on perceived effectiveness, there was a significant main effect, F (1, 244) = 26.627, p < .001. The same data when examined in error bars as shown in Figure 4.12, the means of proposed and existing systems were observed to be dissimilar or not equal, indicating a probable significant relationship.



Figure 4.12: Effect of Existing and Proposed Systems on Perceived Effectiveness

4.7.1.4 Interaction Effect on Effectiveness (H_{14})

In Figure 4.13, the effectiveness of the proposed system is higher for users' classified as with medium and high flexible behavior (1 - Least effective, 5 - Highly effective). However, for users with least flexible behavior, the effectiveness of the existing
system is marginally higher than the proposed one. The perceived effectiveness of the existing and proposed system for users with least flexible behavior also shares an interaction effect and F-test results further reveal a significant interaction between the effect of existing and proposed online reservation systems and the user classifications on perceived effectiveness, F (2, 244) = 15.612, p < .001.



Figure 4.13: Interaction Effect of Existing and Proposed Systems on Perceived Effectiveness

4.7.1.5 Descriptive Statistics on Efficiency (H₁₄)

Table 4.30 displays means, standard deviations and number of respondents in all classifications based on flexible traveling behavior.

Interface Design	Classification of Flexible Behavior	Mean	Std. Deviation	Ν	
_	Least Flexible	4.19	.358	30	
Existing	Medium Flexible	3.79	.298	19	
U U	Highly Flexible	3.71	.558	76	
	Total	3.83	.521	125	
	Least Flexible	4.01	.514	30	
Proposed	Medium Flexible	3.84	.661	19	
	Highly Flexible	4.00	.473	76	
	Total	3.98	.514	125	
	Least Flexible	4.10	.448	60	
Tatal	Medium Flexible	3.82	.506	38	
10[11]	Highly Flexible	3.85	.535	152	
	Total	3.91		250	

Table 4.30: Descriptive Statistics on Efficiency

Respondents with least flexible behavior rated the existing (mean 4.19, S.D 0.358) and proposed systems (mean 4.01, S.D 0.514) relatively higher in terms of their efficiency. Likewise, respondents with a highly flexible behavior, although rated the existing systems lowest in terms of its efficiency (mean 3.71, S.D 0.558), their rating for the proposed systems is relatively higher in terms of its efficiency (mean 4.00, S.D 0.473). These means will be useful in interpreting the direction of any effects that emerge in further analysis of variance.

4.7.1.6 Two-way Analysis of Variance on Efficiency (H_{14})

There is a significant effect of user's classifications on the basis of flexible traveling behavior; since the F-ratio as shown in Table 4.31 is highly significant indicating that the users' high, medium and low flexible traveling behavior is significantly affected by the proposed and existing systems.

Dependent Variable: PU - Efficiency					
Source	Type III Sum of Squares	df	Mean Square	F	Sig.
Corrected Model	6.714 ^a	5	1.343	5.377	.000
Intercept	2793.594	1	2793.594	11186.352	.000
interfacedesign	.136	1	.136	.545	.461
classification	3.029	2	1.515	6.065	.003
interfacedesign * classification	2.436	2	1.218	4.878	.008
Error	60.935	244	.250		
Total	3880.556	250	<u> </u>		
Corrected Total	67.648	249			

 Table 4.31: Two-way ANOVA of Efficiency

a. R Squared = .099 (Adjusted R Squared = .081)

4.7.1.7 Error Bars on Efficiency (H_{14})

Figure 4.14 shows that when the influence of existing and proposed system is ignored, the overall efficiency of systems is very similar for users' with medium and high flexible behavior, as the means of these two groups are approximately equal. However the perceived efficiency of the system for users' with least flexible behavior, not only differs from the other two groups, it remains also higher.



Figure 4.14: Effect of User Classification of Flexible Behavior on Perceived Efficiency

When the effect of existing and proposed systems is examined on perceived efficiency, there was not a significant main effect, F (1, 244) = 0.545, p = .461. The same data when examined in error bars Figure 4.15, the means of proposed and existing systems were observed to be similar or equal, indicating a probable insignificant relationship.



Figure 4.15: Effect of Existing and Proposed Systems on Perceived Efficiency

4.7.1.8 Interaction Effect on Efficiency (H_{14})

In Figure 4.16, the efficiency of the proposed system is higher for users' classified as with medium and high flexible behavior (1 - Least effective, 5 - Highly effective).

However, for users with least flexible behavior, the efficiency of the existing system is marginally higher than the proposed one. The perceived efficiency of the existing and proposed system for users with least flexible behavior also shares an interaction effect and F-test results further reveal a significant interaction between the effect of existing and proposed online reservation systems and the user classifications on perceived efficiency, F (2, 244) = 4.878, p < .05.



Figure 4.16: Interaction Effect of Existing and Proposed Systems on Perceived Efficiency

4.7.1.9 Descriptive Statistics on Satisfaction (H_{14})

Table 4.32 displays means, standard deviations and number of respondents in all classifications based on flexible traveling behavior.

Respondents with least flexible behavior rated the existing (mean 4.10, S.D 0.377) and proposed system (mean 4.04, S.D 0.393) relatively higher in terms of their satisfaction, and their satisfaction has also fallen by 6% with proposed systems. Likewise, respondents with a highly flexible behavior, although rated the existing systems lowest in terms of satisfaction (mean 3.44, S.D 0.576), their rating for the proposed systems is relatively higher in terms of satisfaction (mean 3.68, S.D 0.556). These means will be useful in interpreting the direction of any effects that emerge in further analysis of variance.

Interface Design	Classification of Flexible Behavior	Mean	Std. Deviation	Ν
· · · · · · · · · · · · · · · · · · ·	Least Flexible	4.10	.377	30
Existing	Medium Flexible	3.60	.348	19
C	Highly Flexible	3.44	.576	76
	Total	3.62	.573	125
	Least Flexible	4.04	.393	30
Proposed	Medium Flexible	3.80	.582	19
-	Highly Flexible	3.92	.417	76
	Total	3.93	.443	125
	Least Flexible	4.07	.383	60
T 1	Medium Flexible	3.70	.484	38
fotal	Highly Flexible	3.68	.556	152
	Total	3.78	.534	250

Table 4.32: Descriptive Statistics of Satisfaction

4.7.1.10 Two-way Analysis of Variance (H₁₄)

There is a significant effect of user's classifications on the basis of flexible traveling behavior; since the F-ratio is highly significant as shown in Table 4.33 indicating that the users' high, medium and low flexible traveling behavior is significantly affected by the proposed and existing systems.

Tal	ble	4.33:	Two-way	ANOVA	of Satisfaction
-----	-----	-------	---------	-------	-----------------

Dependent Variable: PU - Satisfaction					
Source	Type III Sum of Squares	df	Mean Square	F	Sig.
Corrected Model	16.230 ^a	5	3.246	14.454	.000
Intercept	2646.191	1	2646.191	11782.881	.000
interfacedesign	1.960	1	1.960	8.728	.003
classification	6.952	2	3.476	15.477	.000
interfacedesign * classification	3.311	2	1.656	7.372	.001
Error	54.797	244	.225		
Total	3636.516	250			
Corrected Total	71.028	249			

a. R Squared = .229 (Adjusted R Squared = .213)

4.7.1.11 Error Bars on Satisfaction (H₁₄)

Figure 4.17 represent the error bars on satisfaction.

When the influence of existing and proposed system is ignored, the overall satisfaction of systems is very similar for users' with medium and high flexible behavior, as the means of these two groups are approximately equal. However the perceived satisfaction of the system for users' with least flexible behavior, not only differs from the other two groups, it remains also higher.



Figure 4.17: Effect of User Classification of Flexible Behavior on Perceived Satisfaction

When the effect of existing and proposed systems is examined on perceived satisfaction, there was a significant main effect, F (1, 244) = 8.728, p < .05. The same data when examined in error bars Figure 4.18, the means of proposed and existing systems were observed to be dissimilar or not equal, indicating a probable significant relationship.



Figure 4.18: Effect of Existing and Proposed Systems on Perceived Satisfaction

In Figure 4.19, the satisfaction of the proposed system is higher for users' classified as with medium and high flexible behavior (1 – Least effective, 5- Highly effective). However, for users with least flexible behavior, the satisfaction of the existing system is marginally higher than the proposed one. The perceived efficiency of the existing and proposed system for users with least flexible behavior also shares an interaction effect and F-test results further reveal a significant interaction between the effect of existing and proposed online reservation systems and the user classifications on perceived satisfaction, F (2, 244) = 7.372, p < .05.





Figure 4.19: Interaction Effect of Existing and Proposed Systems on Perceived Satisfaction

4.7.1.13 Levene's Test on Existing Systems (H_{L})

Results of Levene's Test as shown in Table 4.34 show non-significant results (p=.140>.05) indicating homogeneity of variance assumption being met, therefore, post hoc analysis can be performed by using Scheffe Test.

Dependent Variable: Perceived Usability (Effectiveness+Efficiency+Satisfaction)									
F	df1	df2	Sig.						
1.995	2	122	.140						

 Table 4.34:
 Levene's Test on Existing Systems

Tests the null hypothesis that the error variance of the dependent variable is equal across groups. a. Design: Intercept + classification + interfacedesign + classification * interfacedesign

4.7.1.14 Post-hoc Multiple Comparisons on Existing Systems (H_{14})

Post-hoc Multiple Comparisons using Scheffe Test is shown in Table 4.35. In case of existing systems, users' classified as with High, Medium and Low flexible behavior did not differ from one another in terms of rating Perceived Usability of the existing OARS.

 Table 4.35: Post-hoc Scheffe Multiple Comparison Using Schefe to Test Users'

 Classification for Existing Systems

Dependent Variable:	Perceived Usability (Ef	fectiveness, Effi	ciency, Sati	sfaction)) of Existing	Systems	
(I) Classification (J) Classificat		Mean	Std.	Q*-	95% Confidence Interval		
of Flexible Behavior	of Flexible Behavior	(I-J)	Error	51g.	Lower Bound	Upper Bound	
T . T1 *11	Medium Flexible	.2645	.13027	.132	0583	.5873	
Least Flexible	Highly Flexible	.0779	.09580	.719	1595	.3153	
Madium Elastible	Least Flexible	2645	.13027	.132	5873	.0583	
Medium Flexible	Highly Flexible	1866	.11396	.266	4690	.0958	
Highly Flexible	Least Flexible	0779	.09580	.719	3153	.1595	
	Medium Flexible	.1866	.11396	.266	0958	.4690	

Based on observed means. The error term is Mean Square (Error) = .197.

4.7.1.15 Levene's Test on Proposed Systems (H_{14})

Results of Levene's Test as shown in Table 4.36 show non-significant results (p = .112 > .05) indicating homogeneity of variance assumption being met, therefore, post hoc analysis can be performed by using Scheffe Test.

 Table 4.36:
 Levene's Test on Proposed Systems

Dependent Variable:PU4			
F	df1	df2	Sig.
5.799	2	122	.112

Tests the null hypothesis that the error variance of the dependent variable is equal across groups. a. Design: Intercept + classification + interfacedesign + classification * interfacedesign

4.7.1.16 Post-hoc Multiple Comparisons on Proposed Systems (H_{14})

Table 4.37 shows post hoc multiple comparison of Users' classification on the basis of their flexible traveling behavior. In rating Perceived Usability of the proposed FOARS, Users' classified as with Least Flexible in their traveling behavior differed

significantly from Users' classified as with Medium Flexible (at p < .001) and Highly Flexible (at p < .001) and vice versa.

Table 4.37: Post-hoc Multiple Comparison Using to Scheffe Test Users'

Dependent Variable: Perceived Usability (Effectiveness, Efficiency, Satisfaction) of Proposed Systems										
(I) Classification	(J) Classification	ssification Mean		Sia	95% Confidence Interval					
of Flexible Behavior	Behavior	(I-J)	Error	oig.	Lower Bound	Upper Bound				
Loost Elovible	Medium Flexible	.6336*	.14926	.000	.2637	1.0035				
Least Flexible	Highly Flexible	.7911*	.10977	.000	.5191	1.0631				
Medium	Least Flexible	6336*	.14926	.000	-1.0035	2637				
Flexible	Highly Flexible	.1575	.13058	.485	1661	.4811				
Highly Flexible	Least Flexible	7911*	.10977	.000	-1.0631	5191				
	Medium Flexible	1575	.13058	.485	4811	.1661				

Classification for Proposed Systems

Based on observed means. The error term is Mean Square(Error) = .259.

*. The mean difference is significant at the .05 level.

4.8 Phase III: Effect of Users' Flexibility on Proposed Systems

This study is to address the 8th research question.

RQ8: Is there a multivariate main effect of user's Flexible Traveling Behavior (High, Medium and Low) on effectiveness, efficiency and satisfaction of the proposed system?

4.8.1 Hypothesis Testing H₁₅

 H_{15} : There are differences among effectiveness, efficiency and satisfaction caused by the users' Flexible Traveling Behavior.

4.8.1.1 Scatter Plots (H₁₅)

Pair wise nonlinear relationships between dependent variables using scatter plots are shown in Figure 4.20, 4.21 and 4.22. In Effectiveness Versus Efficiency, Effectiveness Versus Satisfaction, Efficiency Versus Satisfaction, Strong, positive, linear relationships is observed as one variable increases in value, the other variable tends to also increase.



Figure 4.20: Linear Relationship between Effectiveness and Efficiency



Figure 4.21: Linear Relationship between Effectiveness and Satisfaction



Figure 4.22: Linear Relationship between Efficiency and Satisfaction

4.8.1.2 Homogeneity of Covariance's (H₁₅)

The second assumption of multivariate analysis was met by examining Box's M, which tests the hypothesis that the covariance matrices of the dependent variables are significantly different across levels of the independent variable as show in Table 4.38.

Box's M	16.935
F	1.337
df1	12
df2	14037.348
Sig.	.189

Table 4.38: Box's Test of Equality of Covariance Matrices

Tests the null hypothesis that the observed covariance matrices of the dependent variables are equal across groups.

Results showed non-significant results (p = .189 > .001) hence indicating that the assumption has not been violated.

The overall F test for the three dependent variables was examined in Multivariate Tests as shown in Table 4.39 by analyzing the statistic called Wilks' lambda (λ), and the F value associated with that. In the case of Independent Variable (IV), User classifications on the basis of their flexible traveling behavior, Wilks' lambda is .667, and has an associated F of 8.971, which is significant at p <001. Furthermore, the partial eta squared (partial ε^2) associated with the main effect of Users' classification is .183 and the power to detect the main effect is 1. Thus, *H*₁₅ was accepted.

Initial interpretation of results based on one-way MANOVA have revealed a significant multivariate main effect for User' classifications, Wilks' $\lambda = .667$, F (6, 240.000) = 8.971, p <. 001, partial $\varepsilon^2 = .183$. Power to detect the effect was 1. Thus, H₁₅ was accepted due to statistically significant impact of Users' classification on three dependent variables measuring users' Perceived Usability of the proposed FOARS. Since, the results for hypothesis testing were statistically significant, so follow-up tests were performed and interpreted.

	Effect	Value	F	Hypothesis df	Error df	Sig.	Partial Eta Squared	Noncent. Parameter	Observed Power ^b
Intercept	Pillai's Trace	.982	2188.873a	3.000	120.000	.000	.982	6566.620	1.000
	Wilks' Lambda	.018	2188.873a	3.000	120.000	.000	.982	6566.620	1.000
	Hotelling's Trace	54.722	2188.873a	3.000	120.000	.000	.982	6566.620	1.000
	Roy's Largest Root	54.722	2188.873a	3.000	120.000	.000	.982	6566.620	1.000
Classification	Pillai's Trac e	.333	8.069	6.000	242.000	.000	.167	48.416	1.000
	Wilks' Lambda	.667	8.971a	6.000	240.000	.000	.183	53.828	1.000
	Hotelling's Trace	.498	9.876	6.000	238.000	.000	.199	59.259	1.000
	Roy's Largest Root	.496	20.013c	3.000	121.000	.000	.332	60.038	1.000

Table 4.39: Multivariate Tests

a. Exact statistic

b. Computed using alpha = .05
c. The statistic is an upper bound on F that yields a lower bound on the significance level.
d. Design: Intercept + classification 134

4.8.1.3 Alpha Adjustment (H₁₅)

The experiment-wise alpha protection provided by the overall or omnibus F test does not extend to the univariate tests. It is thus important to make an alpha adjustment to account for multiple ANOVAs being run. Hence, confidence level is divided by the number of tests to be performed, as in this case, F tests for the three dependent variables is required to be at p < 0.017 (.05/3).

4.8.1.4 Univariate ANOVAs (H15)

Table 4.40 shows that Users' classification on the basis of their flexible traveling behavior have a statistically significant effect on three dependent variables assessing Perceived Usability of FOARS, Effectiveness (F (2, 122) = 28.680; p = .000 < .017; partial ε^2 = .320), Efficiency (F (2, 122) = 10,776; p = .000 < .017; partial ε^2 = .15) and Satisfaction (F (2, 122) = 18.738; p = .000 < .017; partial ε^2 = .235).

Source	Dependent Variable	Type III Sum	df	Mean	F	Sia	Partial Eta	Noncent.	Observed
		of Squares		Square	Ľ	oig.	Squared	Parameter	Powerb
Corrected Model	PU - Effectiveness	32.449a	2	16.225	28.680	.000	.320	57.360	1.000
	PU - Efficiency	5.058c	2	2.529	10.776	.000	.150	21.551	.989
	PU - Satisfaction	9.572d	2	4.786	18.738	.000	.235	37.475	1.000
Intercept	PU - Effectiveness	1080.033	1	1080.033	1909.144	.000	.940	1909.144	1.000
	PU - Efficiency	1377.358	1	1377.358	5868.105	.000	.980	5868.105	1.000
	PU - Satisfaction	1252.057	1	1252.057	4901.941	.000	.976	4901.941	1.000
classification	PU - Effectiveness	32.449	2	16.225	28.680	.000	.320	57.360	1.000
	PU - Efficiency	5.058	2	2.529	10.776	.000	.150	21.551	.989
	PU - Satisfaction	9.572	2	4.786	18.738	.000	.235	37.475	1.000
Error	PU - Effectiveness	69.017	122	.566					
	PU - Efficiency	28.636	122	.235					
	PU - Satisfaction	31.161	122	.255					
Total	PU - Effectiveness	1457.222	125						
	PU - Efficiency	1871.778	125						
	PU - Satisfaction	1680.594	125						
Corrected Total	PU - Effectiveness	101.467	124						
	PU - Efficiency	33.694	124						
	PU - Satisfaction	40.733	124	·····					

 Table 4.40: Univariate ANOVAs

a. R Squared = .320 (Adjusted R Squared = .309)
b. Computed using alpha = .05
c. R Squared = .150 (Adjusted R Squared = .136)
d. R Squared = .235 (Adjusted R Squared = .222)

4.8.2 Hypothesis Testing H₁₆

 H_{16} : Effectiveness, efficiency and satisfaction in the proposed FOARS is highest for users with highest flexible behavior.

4.8.2.1 Descriptive Statistics on Perceived Usability (H₁₆)

The descriptive analysis on Effectiveness, Efficiency and Satisfaction is presented in Table 4.41 and bar charts are shown in Figure 4.23, 4.24 and 4.25, respectively.

Classification of F	lexible Behavior	Mean	Std. Deviation	N
PU - Effectiveness	Highly Flexible	4.16	.457	60
	Medium Flexible	3.46	.725	38
	Least Flexible	3.50	.904	152
	Total	3.65	.839	250
PU - Efficiency	Highly Flexible	4.10	.448	60
	Medium Flexible	3.82	.506	38
	Least Flexible	3.85	.535	152
	Total	3.91	.521	250
PU - Satisfaction	Highly Flexible	4.07	.383	60
	Medium Flexible	3.70	.484	38
	Least Flexible	3.68	.556	152
	Total	3.78	.534	250

Table 4.41: Descriptive Statistics on PU



Figure 4.23: Bar Chart Showing Effectiveness of the Proposed System



Figure 4.24: Bar Chart Showing Efficiency of the Proposed System



Figure 4.25: Bar Chart Showing Satisfaction of the Proposed System

4.8.2.2 Levene's Test of Equality of Error Variances (H_{16})

Homogeneity of variances for the three dependent variables with significant Universate ANOVAs was examined as shown in Table 4.42. The Levene's statistics for the three DVs showed non-significant results (Effectiveness; p = .401 > .05, Efficiency; p = .051 > 0.05, Satisfaction; p = .200 > 0.05). This indicated that the

group variances were equal and post-hoc comparison of pair wise group means could be examined by computing Sheffé test and Games-Howell.

	F	df1	df2	Sig.		
PU - Effectiveness	.921	2	122	.401		
PU - Efficiency	3.048	2	122	.051		
PU - Satisfaction	1.631	2	122	.200		
Tests the null hypothesis that the error variance of the dependent variable is equal across groups.						
a. Design: Intercept + users' classification						

Table 4.42: Levene's Test of Equality of Error Variances

4.8.2.3 Alpha Corrections for Post-hoc Multiple Comparisons (H_{16})

Since Post Hoc multiple comparisons involve 9 tests to be performed therefore, confidence level has been reset at .05/9=.005. Post Hoc multiple comparisons in terms of means scores are shown in Table 4.43.

4.8.2.4 Post-hoc Multiple Comparisons (H₁₆)

Post-hoc Multiple Comparisons using Scheffe Test is shown in Table 4.43. In case of existing systems, users' classified as with High, Medium and Low flexible behavior did not differ from one another in terms of rating Perceived Usability of the existing OARS.

Effectiveness – in terms of effectiveness of the proposed system, users with highly flexible behavior differed significantly from users with medium (p=.000<.005) and least flexible (p=.000<.005) behavior. This is also evident from descriptive statistics table that users with highly flexible behavior rated the proposed systems effectiveness highest (M=4.16, SD = .457) which is way above the average rating by other groups.

Efficiency – in terms of efficiency of the proposed system, users with highly flexible behavior differed significantly from users with least flexible (p=.000<.005) behavior. This is also evident from descriptive statistics table that users with highly flexible behavior rated the proposed systems efficiency highest (M=4.10, SD = .448).

Satisfaction – in terms of satisfaction of the proposed system, users with highly flexible behavior differed significantly from users with medium (p=.004<.005) and least flexible (p=.000<.005) behavior. This is also evident from descriptive statistics table that users with highly flexible behavior rated the proposed systems satisfaction highest (M=4.07, SD = .383) which is way above the average rating by other groups.

Based on these results H_{16} is accepted since users with highly flexible behavior differed significantly from users with medium and low flexible behavior in terms of rating effectiveness, efficiency and satisfaction of the proposed system.

Dependent Variable					Std. Error	Sig.	95% Confidence Interval		
		(1) Classification of Flexible Behavior	(J) Classification of Flexible Behavior	Mean Difference (I-J)			Lower Bound	Upper Bound	
PU -	Scheffe	Lially Dissible	Medium Flexible	1.00*	.221	.000	.45	1.54	
Effectiveness		Highly Flexible	Least Flexible	1.22*	.162	.000	.82	1.63	
		Madium Flavible	Least Flexible	-1.00*	.221	.000	-1.54	45	
		Medium Flexible	Highly Flexible	.23	.193	.499	25	.71	
		L post Flowible	Highly Flexible	-1.22*	.162	.000	-1.63	82	
		Least Flexible	Medium Flexible	23	.193	.499	71	.25	
	Games-	Lighly Etavible	Medium Flexible	1.00*	.182	.000	.55	1.45	
	Howell	Hignly Flexible	Least Flexible	1.22*	.128	.000	.92	1.53	
		Medium Flexible	Least Flexible	-1.00*	.182	.000	-1.45	55	
			Highly Flexible	.23	.189	.458	24	.69	
		Least Flexible	Highly Flexible	-1.22*	.128	.000	-1.53	92	
			Medium Flexible	23	.189	.458	69	.24	
PU - Efficiency	Scheffe	Highly Flexible	Medium Flexible	.40*	.142	.022	.05	.75	
			Least Flexible	.48*	.104	.000	.22	.74	
		Madium Playible	Least Flexible	40*	.142	.022	75	05	
		Media	Medium Flexible	Highly Flexible	.08	.124	.799	22	.39
		Least Flexible	Highly Flexible	48*	.104	.000	74	22	
			Medium Flexible	08	.124	.799	39	.22	
	Games- Howell	Games- Howell Highly Flexible	Medium Flexible	.40*	.095	.000	.17	.63	
			Least Flexible	.48*	.091	.000	.26	.70	
		Medium Flexible	Least Flexible	40*	.095	.000	63	17	
			Highly Flexible	.08	.094	.649	14	.31	
		L aget Flowible	Highly Flexible	48*	.091	.000	70	26	
			Medium Flexible	08	.094	.649	31	.14	

 Table 4.43: Post-hoc Multiple Comparisons using Scheffe

	(I) Classification of Flexible Behavior			Std. Error	Sig.	95% Confidence Interval	
Dependent Variable		(J) Classification of Flexible Behavior	Difference (I-J)			Lower Bound	Upper Bound
PU - Satisfaction Scheffe	Highly Flexible	Medium Flexible	.51*	.148	.004	.14	.87
		Least Flexible	.67*	.109	.000	.40	.94
	Medium Flexible	Least Flexible	51*	.148	.004	87	14
Games- Howell		Highly Flexible	.16	.130	.464	16	.48
	Least Flexible	Highly Flexible	67*	.109	.000	94	40
		Medium Flexible	16	.130	.464	48	.16
	Highly Flexible	Medium Flexible	.51*	.105	.000	.25	.76
		Least Flexible	.67*	.095	.000	.44	.89
	Medium Flexible Least Highly	Least Flexible	51*	.105	.000	76	25
		Highly Flexible	.16	.104	.275	09	.41
	Least Flexible	Highly Flexible	67*	.095	.000	89	44
		Medium Flexible	16	.104	.275	41	.09

Table 4.43: Post-hoc	Multiple Com	parisons using S	Scheffe (continue	e)
	manipic com	Parisons asing r		~)

Based on observed means. The error term is Mean Square(Error) = .255. *. The mean difference is significant at the .05 level.

4.9 Chapter Summary

In this chapter the results obtained from the studies and case study are presented. To achieve this goal, this chapter followed the research objectives as described earlier in Chapter 3 with corresponding hypothesis to organize the results. The results obtained from the studies helped identified assessing user needs in terms of System's Flexibility and Users' Flexibility. This was followed by the classification of users on the basis of their flexible traveling behavior. To validate the proposed framework for the development of flexible online airline reservations systems, a case study was conducted. The case study was conducted using quantitative technique whereby participants were requested to report their effectiveness, efficiency and satisfaction with the proposed systems using prototype. The results are discussed in detail in Chapter 5.

CHAPTER 5

DISCUSSION

5.0 Chapter Overview

This chapter discusses the results obtained in chapter 4 based on five different studies and a case study. The overall research was based on investigating three core research objectives with eight corresponding research questions and sixteen corresponding hypothesis. Results obtained during the study supported most of the hypotheses as shown in Table 5.1. The following sections will discuss these results in detail.

To achieve the 1st research objective, Section 5.1, 5.2 and 5.3 are dedicated for the discussion on the results obtained from the three studies conducted in Phase I answering the corresponding research questions RQ1, RQ2 and RQ3, respectively. To attain the 2nd research objective, Section 5.4 and 5.5 present discussions on the results obtained from the two studies conducted in Phase II answering the corresponding research questions RQ4 and RQ5, respectively. To conquer the 3rd research objective, Section 5.6, 5.7 and 5.8 are devoted for the discussion on the results obtained from the corresponding research questions RQ6, RQ7 and RQ8, respectively.

Finally, Section 5.9 discusses the recommendations for the designing of flexible Online Airline Reservation Systems and Section 5.10 cover the chapter summary.

Research Objectives	Research Questions	Hypotheses	Results
To assess user needs (System's	RQ1: What are the issues with flexibility of Online Airline	H ₁ : Non-functional Requirements are perceived to have an impact on the usability of OARS.	Supported
Flexibility and Users' Flexibility)	Reservation Systems, whether or not flexibility is one of the reasons for	H_2 : Functional Requirements are perceived to have an impact on the usability of OARS.	Not Supported
associated with Online Airline	users not using such systems?	H ₃ : The perceived flexibility of OARS affects the usability of such systems.	Supported
Reservation Systems.		H ₄ : Functional Requirements of OARS are inversely associated with the flexibility of the systems.	Supported
		H ₅ : The availability of resources and skills set influence upon the usability of OARS.	Supported
	RQ2: To what extend flexible users can compromise with service quality attributes of Online Airline Reservation Systems?	H_6 : The level of satisfaction with existing SBTs is different for respondents with different attitudes towards Users' Flexibility in compromising on SQAs of the airline.	Supported
		H_7 : The level of satisfaction with existing OTAs is different for respondents with different attitudes towards Users' Flexibility in compromising on SQAs of the airline.	Supported
	RQ3: How users' satisfaction with an existing SBTs is rated against their choice of OTA feature and reflected in their integration assessment of the same for making SBTs more flexible Online Airline Reservation Systems?	H ₈ : Users' satisfaction with existing SBTs is different across their choice of four OTA features for making SBTs more FOARS.	Supported

Research Objectives	Research Questions	Hypotheses	Results
To investigate users' perception on factors	RQ4: How users' perception on factors influencing flexible traveling behavior and flexible OARS is determined?		Supported
influencing Flexible Traveling	RQ5: How to classify Users' on the basis of their Flexible Traveling	H ₉ : Users can be classified on the basis of their Flexible Traveling Behavior.	Supported
Behavior and to classify them on	Behavior into High, Medium and Low flexible and how to investigate interrelationships among System's Flexibility, Users' Flexibility and	H ₁₀ : User's Flexible Behavior and their Perceived Usability is correlated.	Supported
the basis of their Flexible Traveling		H ₁₁ : User's Flexible Behavior and System's Flexibility is correlated.	Partially Supported
Behavior.	Perceived Usability of existing Online Airline Reservation Systems?	H_{12} : Perceived Usability of OARS is not affected by users' Flexible Traveling Behavior after adjusting for the effect of the covariate, System's Flexibility.	Supported
To study the relationship between users' Perceived Usability and travelers Flexible Traveling Behavior with existing and proposed Online Airline Reservation Systems.	RQ6: How do service quality attributes of airlines and external variables jointly predict flexible behavior of travelers?	H_{13} : Flexible behavior of travelers cannot be predicted by service quality attributes and external variables.	Supported
	RQ7: How does user Perceived Usability with the existing and the proposed system differs?	H ₁₄ : User Perceived Usability with existing and proposed systems is different across the three groups.	Supported
	RQ8: Is there a multivariate main effect of user's Flexible Traveling Behavior (High, Medium and Low) on	H_{15} : There are differences among effectiveness, efficiency and satisfaction caused by the users' Flexible Traveling Behavior.	Supported
	effectiveness, efficiency and satisfaction of the proposed system?	H_{16} : Effectiveness, efficiency and satisfaction in the proposed FOARS is highest for users with highest flexible behavior.	Supported

 Table 5.1: Summary of Research Questions, Hypotheses and Results (continue)

146

5.1 System's Flexibility

In systems engineering, non-functional characteristics are very important [130]. Normally, Functional Requirements (FRs) and Non-Functional Requirements (NFRs) are elicited separately but later they are merged together to assess the satisfaction level of FRs over NFRs [130]. Sometimes, NFRs are so much dependent on FRs that their enhancement is not possible. This is especially true in the case of Transaction Processing Systems where flexibility is sometimes reduced strongly to avoid non-standard operations. Nevertheless, the results of the study suggest that there is a significant relationship between NFRs and FRs and that NFRs (including flexibility) have a strong relationship to the usability of the reservation systems.

However, the statistical results indicate that there is a poor correlation between FRs and the usage of Online Airline Reservation Systems as shown in Chapter 4, Section 4.1.3. The rationale behind these could be that NFRs are always assessed against the availability of FRs. However, users are bound to assess the flexibility of existing online reservation systems on the basis of the FR that they are realistically exposed to. Therefore, when online users are inquired to respond to questions such as what additional FR features they like to see in online reservation systems, they may not conveniently know "what to ask for?" unless they are exposed to such facilities. On the other hand, NFRs such as the flexibility of the offered features are easier to assess. This indicates why customers wish to have a more flexible system rather than having more features.

Furthermore, the data shows that flexibility and other NFR are placed highest in terms of their Cronbach's α score as shown in Chapter 4, Section 4.1.1, which otherwise is low, if the data has multidimensional structures. This indicates that online users perceive the elements of flexibility and NFRs in online reservation systems at almost equal levels, whereas they perceive the elements of FRs less, these being the third-ranked item in terms of its Cronbach's α score.

The finding suggested that, first, the incorporation of flexibility has a significant impact on the usability of online reservation systems as shown in Chapter 4, Section 4.1.4 and, second, FRs and the usability of online systems are perceived to have a

meager influence on each other as discussed above. In contrast to the user's perception, however, one should not overlook, that flexibility is mirrored in functional requirements as well: Typically, one needs to—so to speak—add more buttons to make a system more flexible.

5.2 Users' Flexibility in terms of Compromising on SQAs of OARS

One of the major research questions of this study was to investigate Users' Flexibility towards Online Airline Reservation Systems. As mentioned earlier in Chapter 2, User's Flexibility is nothing but users' ability to rapidly change from one course of action to another and it is reflected in users' decision making behavior [53]. For understanding flexible human behavior, decision making is an important reflection of users thought process. In the context of User's Flexibility, different users may have different needs, interests and wishes to be served and system's effectiveness, efficiency and satisfaction may vary from one user to another based on their usability perception. For some users a system may be very effective but this may not be true for all.

The results of this study as shown in Chapter 4, Section 4.2.1 demonstrated that apparently there was enormous difference between the satisfaction levels of the three respondents groups, which may not be the actual case. Therefore as a follow-up and to backup this, the same data was interpreted using Error Bars with 95% Confidence Intervals (CI). CI of the groups is closely related to the results of the analysis of variance for these groups. The confidence interval for each graph showed a linear pattern of the sample distribution which otherwise appeared to be showing huge variations in the simple means plot as shown in Chapter 4. Section 4.2.1.2. In Error bars it was examined if mean of one group was included in the confidence interval of the other two groups. If yes, then it was interpreted that there was no difference among the groups. It is not relevant whether the error bars 'overlap' but whether the mean of one group 'overlaps' with the error bars of the other [185]. The confidence intervals can overlap by as much as 25 percent of their total length and still show a significant difference between the means for each group. Any more overlap and the results will not be significant [186].

T-Test (User's Flexible Behavior as outcome variable) showed significant results, for H₆ (t =36.760, df = 169, p = 0.000 < 0.001) and for H₇ (t =35.509, df = 169, p = 0.000 < 0.001). This indicated the three groups were significantly different in terms of their rated satisfaction in SBTs and OTAs.

For, H_6 and H_7 the difference among the mean satisfaction was found to be statistically significant in ANOVA as shown in Chapter 4, Section 4.2.2.1 and 4.2.4.1 respectively, but that does not indicate which means are actually making the difference. In other words, three groups of respondents are different, but how they are different was yet to be determined. Moreover, in error bar, the size of confidence intervals for the three groups also differed from one another. When F-test with a factor that consists of three or more means and additional exploration of the differences among means is needed to provide specific information on which means are significantly different from each other, Post hoc tests are performed. Therefore, ANOVA results of H_6 and H_7 wer e further investigated by performing Post-hoc Scheffe's tests. Scheffe's procedure is the most popular of the post hoc procedures, the most flexible, and the most conservative [187]. Scheffe's procedure corrects alpha for all pair-wise or simple comparisons of means, but also for all complex comparisons of means as well.

Post-hoc Scheffé tests as shown in Table 4.5 and Table 4.8 as shown in Chapter 4, Section 4.2.2.2 and 4.2.4.2, respectively demonstrated that in both cases (i.e. satisfaction level with existing SBTs and satisfaction level with existing OTAs) there is a significant difference between the pair of means of the respondents who reported their flexible attitude as "Cannot compromise" on service quality attributes of the airline with those who "Can compromise". The same group of respondents also differed significantly from the group of respondents who reported their flexible attitude as "May compromise" on service quality attributes.

Furthermore, the effect size show that 19% of the total variance in satisfaction with existing self-booking tools and 7.5% of the total variance in satisfaction with existing online travel agencies of the airline is accounted by the flexible attitude of the users in terms of compromising on service quality attributes of the airline. This is suggestive of the fact that there is some meaningful difference among the groups and

hence the level of satisfaction with existing SBTs and OTAs are considered different for respondents with different attitudes towards Users' Flexibility in compromising on service quality attributes of the airline.

5.3 Integration Assessment of OTAs Features into SBTs

Another important research question was related to the integration assessment of Online Travel Agencies (OTAs) features into Self-booking Tools (SBTs). The findings of this study as shown in Chapter 4, Section 4.3.1.1 showed that integration of OTA features would make SBTs more flexible Online Airline Reservation Systems. This was in accordance with PhoCusWright Report 2009 [9] that reported the overall share of online travel agencies is improving day-by-day, and was recorded at 13% in 2008 in US travel market alone and projected to touch 16% in 2011 as mentioned earlier in Chapter 2. This was further justified from Yahoo Travels' claim which says that 76% of all online travel purchases occur as a result of search function. The table presented in Chapter 2 (Section 2.1.2.2, Table 2.2) showed innovative attributes and function that have contributed immensely towards the popular acceptance of OTAs over Airlines' SBTs and are also widely common among travel companies in recent years [1], [16].

The findings of this study showed that respondents who considered integration of OTA features would make SBTs more flexible and also who stated otherwise, reported absolute lower level of satisfaction with hotel search facility as a recommended solution for SBTs as shown in Chapter 4, Section 4.3.1.2. The finding is very much self-explanatory because Hotel Search facility is a popular OTA integrated feature which displays information of multiple hotels, and promotional packages available in deals with specific airlines. However, in the context of SBTs, this feature may not be feasible for integration, since SBTs are self-booking tools offered by an airline, where multiple carrier and hotel reservation sources are not incorporated with the reservation planning.

When we look at the other two OTA features, i.e. Matrix Display and Alternate Airport Search the pattern is quite different and this is the part of interaction effect that the direction of difference is not in the same direction. Respondents who considered integration of OTA features into SBTs will not necessarily make them flexible, opted for Matrix Display and Alternate Airport Search as their recommended solution for the SBTs. The Matrix display and hotel search features are unique in sense of incorporating multiple sources reservation information. However from integrating into SBTs perspective, they are not feasible because they require merger of multiple information resources, which might not be an acceptable standard for an airline due to its privacy policy and other regulations. On the contrary, alternate airport search is related to providing additional information as well as the extent to which a traveler is willing to be flexible in identifying his/her destination sources. This feature seems practical and has implications for integration into SBTs.

Finally, unlike other OTA innovations, the opaque fare mechanism depends on hidden characteristics of the traveling plan, thus leveraging upon traveling behavior of leisure travelers, who are always up for grabs and less sensitive to traveling plans. This study showed that it is highly recommended by the experts in the airline industry for integration into SBTs. Researchers point out [117] that opaque products are flexible in characteristics; therefore, a seller is in a unique position to offer horizontally differentiated products to customers upon purchase due to the flexibility of assigning pre-determined products to the customer. Opaque mechanics gained popularity due to its very unique price discrimination mechanism [118] which could generate incremental revenue for the airline by deliberating upon price sensitive consumers [119]. In the very short time, opaque selling has attained the status of a competitive lever for the airline, signifying that an airline could suffer revenue loss to its competitors by not opting to offer opaque offers [120].

Airlines secure incremental revenue by way of disposing off their distressed inventory through last-minute sale discounts. The last minute sale discounts are offered at heavily discounted rates. This mechanism has nothing much of 'opaque' or 'hidden' in it. However, this type of direct selling at the last-minute is considered very risky for the airline, since potential travelers prefer to wait for last-minute sales and not purchase in anticipation of heavy discounts [40]. Such a condition may put an airline in a very risky position with potential possibility of revenue loss. That is why

5.4 Users' Perception on Factors Influencing Upon Flexible Traveling Behavior

Investigation of Users' Perception on factors influencing upon Users Flexible Traveling Behavior was also an important research questions. Results of this study as shown in Chapter 4, Section 4.4.1 indicated that users traveling behavior is molded by a number of important personality relevant determinants, both internal and external in characteristics. The findings was supported by other studies mentioning that flexibility is undeterminable without accounting for traveler's flexible traveling behavior in the case of Online Airline Reservation Systems [62].

The responses received through online discussion forums as shown in Chapter 4, Section 4.4.1 identified some very basic key elements that instigate flexibility in a travelers' traveling behavior, especially if he or she is traveling on a non-rigid schedule. For instance, a promotional package offered by a particular airline makes it a lucrative option for travelers who are ready to travel at any alternatively allocated itinerary. Similarly, if an airline does not offer services and comfort to its customers, they are likely to become inflexible for opting to fly with it, even if it offers lower fares. Likewise, travelers flying purpose and flying frequency determines the extent to which he/she is ready to become a flexible traveler. External influences of travelers' family and friends also play an important role in shaping up their flexible traveling behavior. As for the factors that influence upon perception of a flexible online reservation system, our analysis showed that a reservation system is perceived as being flexible if it offers easiness to users in terms of making reservations with minimal skills required, is not complicated in terms of operation and offers multiple features and options to users.

The responses received through in-depth interviews as shown in Chapter 4, Section 4.4.1 instigated flexibility in a travelers' behavior, factors such as airlines repute and travelers status consciousness contribute towards their flexible traveling behavior. Interestingly, customer loyalty, traveling mileage and patriotism also came into the limelight with our in-depth interviews with travelers. This indicates, if a traveler has a customer loyalty or earned traveling mileage from a particular airline it will influence upon his/her flexible traveling behavior, as well as, how patriotically associated the traveler feels with his/her country's airline. Finally, travelers who wish to live an eco-friendly life supported the idea of the green environment and they were of the view that if an airline supports eco-friendly policies, with lesser carbon emissions and lesser fuel consumption, it will influence upon their flexible traveling behavior as well as flying priorities. Likewise, it was revealed that travelers flexible traveling is also dependent upon their interest taking in recreational and leisure activities. Respondents who indicated that they were frequent flyers attributed flexible traveling behavior directly to their interest in recreational and leisure activities. Some travelers also highlighted that flexible traveling behavior of a single person or an individual is different from the one who is traveling with his/her family. As families with children would prefer to travel mostly during the school breaks or holiday sessions, therefore, their flexible traveling behavior will be accordingly influenced by this factor. Travelers during discussion also pointed out the high influence of travelers' occupation on his flexible traveling behavior, as travelers who are associated or are in a job that gives them high work load, lesser breaks and requires more on-job working and lesser mobility, would be less flexible in their traveling behavior. As for the factors that influence upon perception of a flexible online reservation system, our analysis showed that factors such as simplicity and easiness were again important determinants. From our in-depth interviews, it was found out that a flexible reservation system is considered to the one that can provide alternative dates for flying to the traveler and at the same time offers self-adjustment functionality, such as cancellation and changes, in the itinerary of the traveler. Some travelers, who were non-native English speakers, pointed that a system would be flexible to them if it supports multi-languages.

Finally, focus group participants had better and deeper understanding of travelers' flexible behavior as well as that of reservation system. In-depth interview results as shown in Chapter 4, Section 4.4.1 with focus groups on factors influencing users' flexible traveling behavior were very much the same as the one discussed earlier. However, additional factors that came into the scene included discounted airfares versus normal airfares and time frame, peak versus off travel season, chosen to fly also substantially influences upon flexible traveling behavior. Focus group provided a more critical feedback on factors influencing upon perceived flexibility of an online airline reservation system and we were able to extract a number of important additional factors in our analysis.

The data collected from three different sources depicted a similarity pattern, especially in case of data collected from online travel forums and in-depth interviews. Data analysis provided a more detailed perspective of travelers' flexible behavior by incorporating socio-economic factors and societal influences. After giving much thought process to results as shown in Chapter 4, Section 4.4.1 following 6 themes emerged under factors influencing upon Flexible Traveling Behavior and 3 themes emerged under factors influencing upon perceived flexibility of a reservation system.

Theme 1: Travelers' Flexible Behavior is molded by their Traveling Consciousness – Consciousness is a subjective experience of awareness. In context of flexible traveling behavior, our study showed that travelers have a certain form consciousness or awareness about their traveling behavior. This resulted in the emergence of first theme.

"I look for great deals through promotional schemes" (ID P2)

"Why I wouldn't travel with an airline that provides heavy discounts? I look for discounts all the time" (ID P6)

"I make cognizant traveling decisions. Do all background work first and then make a decision to purchase a ticket or not" (OTF P11)

"Airlines hefty discounts through their travel mileage schemes are very attractive and can make me to travel any time during the year" (FG P7)

Theme 2: Travelers' Flexible Behavior is molded by their Belief that they have the Required Digital Skills – Study showed that having the digital skills to search for the best traveling deals, can influence upon once flexible traveling behavior. Digital skills reflect upon travelers' self-efficacy and have resulted in evolution of our theme 2.

"I don't care traveling with any airline, as I can find best traveling online deals with much ease" (OTF P8)

"I don't waste time on googling what I am looking for, I know where to exactly go when I need to purchase an online ticket" (ID P26)

"It takes patience and perseverance to find out best traveling deals" (OTF P3)

Theme 3: Travelers' Flexible Behavior is molded by their Self-belief as Flexible Travelers – This theme was categorized on the basis of how respondents reflected upon their self-identity as travelers. Two sub-themes emerged.

Sub theme 1: Traveling Motivation (traveling purpose, flying frequency, interest in recreation).

Motivation to travel is an important factor to influence upon travelers flexible behavior. For instance;

"It very much depends upon my traveling purpose and motivation" (ID P6)

"I love to travel, this is my biggest motivation" (OTF P1)

"As a sales man, if one has to travel around the year, his traveling motivation will be lower" (FG P15)

Sub theme 2: Socio-economic Position

Socio-economic standing of the traveler also influence upon flexible traveling behavior.

"depends on the job you are in" (OTF P4)

"... with limited financial resources, one can only long to take leisure trips" (ID P2)

"I am more flexible in traveling when I am paying for my ticket and can also compromise on comfort. This is not the case when my office is paying for me" (FG P9)

Theme 4: Travelers' Flexible Behavior is molded by Societal Influences – This theme emerged due to the social influences that influence upon a traveler's flexible traveling behavior. Social influences occur when a traveler's flexible behavior is affected by other people.

"We travel with our two sons. So our preference is to travel during their school holidays" (ID P16)

"We look for cheap deals as we have a large family to support and travel together" (OTA P17)

"My father reserves ticket for me when I travel or I call my friend in a travel agency. They decide for me. Simple!" (ID P14)

Theme 5: Travelers' Flexible Behavior is molded by how they attribute A Cause to their Traveling Behavior – This theme emerged out a certain group of respondents who attributed a cause to their traveling behavior.

"We prefer Malaysian airlines over other airlines. It is our pride" (OTF P12)

"I prefer airlines with high repute, since they significantly enhance my traveling experience and pleasure" (ID P9)

"I am a supporter of eco-friendly living. I would be flexible to travel with an airline that supports this great cause" (ID P14)

"Airlines repute fascinates me a lot" (FG 8)

Theme 6: Travelers' Flexible Behavior is molded by their Prior Traveling Experiences – Surely, traveler's prior traveling experiences influence upon their future traveling decisions. This theme has much to do with traveling comfort that one experiences.

"I prefer airlines with better leg space area and comfortable seats" (ID P30)

"Sometimes customers compromise on jet ways availability" (FG P18)

"Staff attitude is a traveling motivator" (OTF P11)

"We prefer airlines that serve Muslim halal food" (OTF P21)

Theme 7: Systems Perceived Flexibility is influenced by its Perceived Usability – A System will never be questioned for flexibility, if it satisfies users' expectations. Therefore, systems perceived flexibility is always dependent on the usability of a system.

"Searching for good packages using existing OTAs and SBTs is quite simple and easy" (ID P8)

"It requires the fewest steps to accomplish what users want to do" (FG 15)

"Users would prefer to use our system because they can make changes in their itinerary very easily" (FG 9)

Theme 8: Systems Perceived Flexibility is influenced by End User Support – Our study showed that post-sale features are very essential in determining the perceived flexibility of a system. This theme has much to do with the user support.

"Every online reservation system has the capability to reserve a ticket..... for me a flexible system is the one which provides after sale service features" (ID 16)

"A flexible reservation system should support user prompting and guidelines throughout the reservation process" (FG P20)

Theme 9: Systems Perceived Flexibility is influenced by Comparison of Features on the Actual Level of Effect regarding to Complete the Reservation Process – This theme emerged due to the comparison features that influence upon
systems perceived flexibility. Provision of multiple features in search of a better flight fare on different dates enhances the systems perceived flexibility.

"Me as a user would like to compare flight fares on different dates" (ID P22)

" It may (system) indicate about multiple destination options to users to choose from and accordingly point out fare changes" (FG P17)

"A system providing multiple searching options, like by dates, fares, destinations" (ID P14)

"If required, changes in itinerary can be made or at least recommended by the customer to the system" (ID P21)

Our analysis showed that an Online Airline Reservation System is perceived as flexible if it provides fare quotes, post-sale features and generates receipt. Likewise, if the reservation system can sort airfares on different dates and destinations, it will be perceived as flexible by users. Moreover, if the user is prompted throughout reservation process for his/her guidance, is timely notified on low airfares though flexible and alternative flying dates, it will influence upon user's perception of a flexible reservation system. Also by incorporating multiple traveling components in a single search, such as hotel search facility, the perceived flexibility of the reservation system will also be influenced.

The findings of the study showed that users traveling behavior is molded by a number of important personality relevant determinants, both internal and external in characteristics. While, traveling consciousness, self-efficacy in digital skills and self-belief as flexible travelers are internal personality relevant determinants influencing directly upon travelers flexible behavior, societal influences, attribution and prior experiences are external personality relevant determinants that indirectly influence upon travelers flexible behavior. Moreover, external determinant may not necessarily always have the same influence every time, depending upon the situation a traveler is in.

5.5 Classification of Users on the Basis of Their Flexible Traveling Behavior

The most important research question of this study was the categorization of users on the basis of their flexible traveling behavior. Results as shown in Chapter 4, Section 4.5.1 revealed that users can be classified as High, Medium and Low flexible on the basis of Users' Flexibility score. The study results was supported by other researchers [61] since consumer behavior researchers have frequently employed schema theory as the theoretical underpinning of their investigations for classification of consumers as with a likelihood of high purchase power, medium purchase power and low purchase power.

The results further revealed that Users' Flexible Behavior is positively associated with Perceived Usability of Flexible Online Airline Reservation Systems as shown in Chapter 4, Section 4.5.2. A traveler is categorized as highly flexible, if he/she possesses a flexible nature of traveling [188]. This indicates that he/she has a non-rigid schedule for traveling. Flexible travelers can travel on alternatively allocated itineraries and their traveling satisfaction comes mainly from heavy discounted fares provided by any airline. On the contrary, a traveler is categorized as low flexible, if he/she possesses an inflexible nature of traveling. This indicates that he/she has a rigid schedule for traveling. Low flexible travelers only prefer to travel as per their indicated or preferred itineraries and their traveling satisfaction comes mainly from meeting their flying deadlines that are mostly based on reaching their destination as per their preferred itineraries. This indicates that as Users' Flexible Behavior improves (from Low to Medium / from Medium to High), their Perceived Usability of Flexible Online Airline Reservation Systems tends to also improve.

To understand causality of interrelationships between Users' Flexible Behavior, System's Flexibility, System's Adaptability, System's Adaptivity and System's Personalization, conclusions are drawn about variability by squaring the correlation coefficients. By squaring the correlation coefficient, a measure of how much of the variability in one variable is explained by the other can be derived.

Although, System's Flexibility share a non-significant correlation coefficient with Users' Flexible Behavior, two of its sub-variables, System's Adaptability and

System's Personalization happen to be sharing significant correlations. The value of r^2 for System's Adaptability is $(0.372)^2 = 0.138$, and for System's Personalization is $(0.314)^2 = 0.098$. This explains how much of the variability in Users' Flexible Behavior is accounted for by two of the sub-measuring variables of System's Flexibility. In percentage terms, System's Adaptability accounts for approximately 14%, while System's Personalization accounts for approximately 10% of the variability in Users' Flexible Behavior and together the two variables account for approximately 24% variability.

5.6 Role of SQAs and External Variables in Flexible Behavior of Travelers

The Pearson Correlation Analysis as shown in Chapter 4, Section 4.6.1.2 demonstrated that all correlations are statistically significant. This may help explain the inconclusive results found by other researchers [89] to determine the contextual interpretation of any event by contextual factors that reinforce viewers' schemas, formulate characteristics of the surrounding environment and ensure effective collaboration between the two.

The results on scatter plot examining service quality attributes of the airline against users' flexible personality as shown in Chapter 4, Section 4.6.1.1 are normally distributed along with linear regression line and have no obvious outliers. 5 represented being least important in terms of compromising on service quality attributes of the airline, while 5 also represented being highly flexible in nature. Therefore, with every point increase in compromising on service quality attributes of the airline, users' flexible behavior tends to increase linearly. The R square for the best fit line is 62.7% indicating a very strong positive relationship between the two variables. This further means that about 62.7% of the variability in Users' Flexibility is accounted for by the service quality attributes they are ready to compromise on or to forgo. Similarly, the results on scatter plot examining external variables against users' flexible personality as shown in Chapter 4, Section 4.6.1.1 are normally distributed along with linear regression line and have no obvious outliers. 5 represented strongly agreeing to the influence of a particular external variable, while 5 also represented being highly flexible in nature.

in strongly agreeing to the external variable influences, users' flexible behavior tends to increase linearly. The R square for the best fit line is 63%, which indicates a very strong positive relationship between the two variables. This further means that about 63% of the variability in Users' Flexibility is accounted for by the external variables and their influences upon their traveling behavior.

The value of the R^2 as shown in Chapter 4, Section 4.6.1 is a measure of how much of the variability in the outcome is accounted for by the predictors. Its value is 0.669, which means that the two variables approximately account for 67% of the variation in predicting users' flexible behavior. The adjusted R^2 gives some idea of how well this model generalizes and the closer its value is to R^2 , the better it is considered for model fitness. In this case, difference for the model is reasonable (0.669 - 0.667 = 0.002 or 0.2%). This shrinkage means that if the model was derived from the population rather than sample, it would account for approximately 0.2% less negligible variance in the outcome.

In Multiple Regression, the model takes a form of an equation, which has a coefficient (b values) for each predictor variable. The first part of the table as shown in Chapter 4, Section 4.6.1.5 estimates b values which indicate individual contribution of each predictor to the model. The *b* value signifies relationship of each predicting independent variable in Users' Flexibility. A positive coefficient indicates a positive relationship, while a negative coefficient indicates a negative relationship. The b*value* also tells us to what degree each predicting independent variable affects Users' Flexibility, if the effects of all other predictors are held constant. The highest positive standardized beta value is for the external variables, namely "Persuasion" (0.264) closely followed by "Engagement" (0.166). This indicates these two independent external variables have a strong impact on determining users' flexible behavior, with variable former having a slightly higher impact. This is followed by a Service Quality Attribute of the airline, "Last minute discounted airfares" (.128), which also tends to have a hefty influence upon users' flexible traveling behavior. External variable "Identity" also makes it into the top contributing factors with standardized beta coefficient (0.125). Finally, Service Quality Attributes of the airline, "Immediate Confirmation of Itinerary on purchase" and "Confirmation of Origin and Destination Airport" were found to have significant impact on determining Users' Flexibility with beta coefficients of (.119) and (.113), respectively.

After examining direct relationship of each predicting variable with Users' Flexibility, the *b value* of each predicting variable in association with its t-test statistic and significance level was interpreted. If the t-test associated with the *b value* is significant at either p<0.1, 0.5 or 0.001 level, then that predictor is making a significant contribution to the model. For this model, external variables Persuasion (t (249) = 1.335; p = .018 < .05), Engagement (t (249) = 2.548; p = .011 < .05), Identity (t (249) = 1.638; p = .037 < .05) and Service Quality Attributes of the airline, last minute discounted airfares (t (249) = 1.829; p = 0.43 < .05), confirmation of origin and destination airport (t (249) = 1.572; p = .017 < .05) and immediate confirmation of itinerary on purchase (t (249) = 1.542; p = .003 < .05) are significant predictors of Users' Flexibility.

5.7 User Perceived Usability with Existing and Proposed Systems

Researchers evaluate Perceived Usability of Online Airline Reservation Systems by doing content analysis on the basis of effectiveness, efficiency and satisfaction [101], [104]. It is also important to investigate web users' characteristics, preferences, expectations [108] and their behavior from online systems, and to compare with the Perceived Usability of the systems. The findings of the study as shown in Chapter 4, Section 4.7.1 demonstrate significant differences on effectiveness, efficiency and satisfaction among the classified users on the basis of their flexible traveling behavior.

The study revealed significant results F (2, 244) = 22.588, p < .001 for the respondents who indicated that there is a significant main effect of the users' classifications (High, Medium, and Low) on perceiving effectiveness of the proposed and existing systems. This effect indicates that overall, when the influence of existing and proposed system is ignored, the users' classifications on the basis of their flexible traveling behavior (High, Medium, and Low) influenced upon how they perceive effectiveness of a system. When the effect of existing and proposed systems was examined on perceived effectiveness, there was a significant main effect, F (1, 244) =

26.627, p < .001. This effect means that with respect to ignoring the classification of users on the basis of their flexible traveling behavior, the proposed and existing systems influenced upon users' perceived effectiveness. Furthermore, F-test results further reveal a significant interaction between the effect of existing and proposed online reservation systems and the user classifications on perceived effectiveness, F (2, 244) = 15.612, p < .001 as shown in Chapter 4, Section 4.7.1.2.

The second F-test for respondents, who indicated that there is a significant effect of user's classifications on the basis of flexible traveling behavior, there was a significant main effect, F (2, 244) = 6.065, p < .05. This effect indicates that overall, when the influence of existing and proposed system is ignored, the users' classifications on the basis of their flexible traveling behavior (High, Medium, and Low) influenced upon how they perceive efficiency of a system. When the effect of existing and proposed systems is examined on perceived efficiency, there was not a significant main effect, F (1, 244) = 0.545, p = .461. This effect means that if we ignore the classification of users on the basis of their flexible traveling behavior, the proposed and existing systems does not influenced upon users' perceived efficiency. Furthermore, F-test results further reveal a significant interaction between the effect of existing and proposed online reservation systems and the user classifications on perceived efficiency, F (2, 244) = 4.878, p < .05 as shown in Chapter 4, Section 4.7.1.6.

This was followed by the third F-test examining the respondents, who indicated there is a significant effect of user's classifications on the basis of flexible traveling behavior, there was a significant main effect, F (2, 244) = 3.476, p < .001. This effect indicates that overall, when the influence of existing and proposed system is ignored, the users' classifications on the basis of their flexible traveling behavior (High, Medium, and Low) influenced upon how they perceive satisfaction of a system. When the effect of existing and proposed systems is examined on perceived satisfaction, there was a significant main effect, F (1, 244) = 8.728, p < .05. Furthermore, F-test results further reveal a significant interaction between the effect of existing and proposed online reservation systems and the user classifications on perceived satisfaction, F (2, 244) = 7.372, p < .05 as shown in Chapter 4, Section 4.7.1.10.

5.8 Multivariate Main Effect of User's Flexible Traveling Behavior on Proposed System's Effectiveness, Efficiency and Satisfaction

One of the major research questions of this study was to investigate the effects of user's flexible traveling behavior on the proposed system's effectiveness, efficiency and satisfaction. The findings of the results as shown in Chapter 4, Section 4.8.1 suggested that users with highly flexible behavior differed significantly from users with medium and low flexible behavior in terms of rating effectiveness, efficiency and satisfaction of the proposed system. This may help explain the inconclusive results found by other researchers [101] saying that Usability of a website cannot be improved without considering consumer intend and behavior.

Researchers [34]–[36] argued that clear understanding of consumer intent and behavior in the case of online airline ticket shopping and elsewhere cannot be achieved without considering the factors that affect purchase decisions. Results as shown in Chapter 4, Section 4.8.1.2 prove that the effectiveness of the proposed system differed significantly from users with highly flexible behavior with medium and least flexible behavior. Similarly, in the case of efficiency of the proposed system, users with highly flexible behavior differed significantly from users with least flexible behavior. Furthermore, satisfaction of the proposed system also differed significantly from users with highly flexible behavior.

5.9 Recommendations for Flexible Online Airline Reservation Systems

Research findings show that system's perceived flexibility is reflected in its Perceived Usability. Perceived Usability is a combination of system's effectiveness, efficiency and satisfaction. End user support or user prompting is considered to be a supporting characteristic that substantially augments efficiency and effectiveness of the system, while multiple options directly influence upon user satisfaction. If a system provides ranges of dates as flying and source destination options at different fares, flexibility of the system is enhanced in its Perceived Usability in the eyes of the users. This is because if a user chooses a flying option 'A' from a given one or two options, he has not made a flexible decision. But if he chooses the same flying option 'A' from a

variety of given flying options, he is likely to enjoy extra satisfaction that he will get from the flexibility of the system and also in his purchase making decision. Likewise, if a system offers multiple flying options, they will also influence upon users' decision and make them change their mind to opt to fly from option 'B' instead of 'A'. This will again have positive influence upon user's satisfaction from the system – Perceived Usability.

This thesis suggests a new approach of reservations while introducing high, medium and low flexible travelers. Traditionally, flights schedule is made available for the travelers after finalizing the resources such as, planes, airports and dates. Travelers make reservations on their favorite dates using the offered flight schedule as shown in the Existing Approach of Figure 5.1. Current approach of airline reservation systems is adequate in terms of making reservations. However, traditional approach does not provide an opportunity to travelers of being flexible and also it does not provide any support to improve load factors in the flights, and therefore, airlines either cancel the flight [189] at 11th hour if they receive less booking in any particular day or they send flight with minimal profit margin.

In this study travelers are categorized on the basis of their flexible traveling behavior as shown in the Proposed Approach of Figure 5.1. In the proposed framework, least flexible customers may choose confirmed schedule offered by the airline with normal fare. Least flexible customers will get their confirmed tickets with seats allocated and itinerary finalized at the time of booking. Since flexible travelers are flexible enough and have not indicated their preferred dates to fly, therefore, they can be requested to provide their preferred time-frames (e.g. Departure: any flight between 10th June to 20th June from Kuala Lumpur to Frankfurt, Arrival: any flight between 10th July to 20th July from Frankfurt to Kuala Lumpur) in which they would like to travel. Tickets to highly flexible customers will be issued at booking time but the seats and flight day will be notified only a week before their actual departure day. Thus, it becomes the responsibility of the arbitrator systems to manage and finalize their traveling itineraries as per their indicated time-frames. Once done, the flexible customers can be allocated seats a few days before their actual departure by the airline.



Figure 5.1: Framework for Flexible Online Airline Reservation Systems

5.9.1 Graphical User Interface for FOARS

The recommendations for the design of Flexible Online Airline Reservation helped in developing the Graphical User Interfaces as shown in the following sections.

5.9.1.1 Home Page of FOARS

Figure 5.2 shows interface for the Home Page of Flexible Online Airline Reservation System. The description of *Flexible* and *Inflexible* passengers as shown in Figure 5.2 will help users to understand the difference between the two terminologies used for booking purpose. Passengers who are flexible with regards to flying dates will be treated as *Flexible Passengers* and those who wish to travel on specified dates will be treated as *Inflexible Passengers*.



Figure 5.2: Home Page for Flexible Online Airline Reservation System

5.9.1.2 Booking Interface of FOARS

The users of FOARS will be provided two options for *Flight Type* i.e. Inflexible and Flexible as shown in Figure 5.3. If users opt for *Flexible Flight*, they will be requested to provide the number of days in which they will be flexible as shown in Figure 5.3

.71			
raTanana and and and and and and and and an			
R	ooking		FlavT Gall
9		Online Booking	Tuxi Gam
Us			And I
25	From :	Kusata Lumpur 👻	-
4 He	To :	Dubai -	
a os	Date :	09-11-2010	
rtisement	Flight Type :	C Inflexible 🕷 Flexible	and an and the
	Flexible Days :	6 📼 days discounts	A service and
10 10	Adult :	3 (2) ++ (44/5)	-
the extra mile	Children :	9 12 (2 = 12 (min)	-
	Infant :	0 • E to 1 year)	0
PLAL PERDIT	Class Type :	Economy *	trav

Figure 5.3: Booking Window for Flexible Online Airline Reservation System

5.9.1.3 Flight Search Window for Flexible Travelers

Flight search with *Flexible Flight Type* will let the users to the searched window with flights available within the flexible dates as shown in Figure 5.4.



Figure 5.4: Flight Search Window for Flexible Travelers

5.9.1.4 Travelers' Details Window for Flexible Travelers

Traveler will be requested to fill in the required information for the booking as shown in Figure 5.5.

Woul Us O You are a flexibl	e passen	ger				
ervices	Flight					
From	То	Flexible Days	Discount	Class		
dvertisement Kuala Lumpur	Dubai	6	20%	Economy		
The second second		Lead Traveller	実品の実施			
Full Name :	Mr +				1 Sector	
no go the extra trile Date of Birth :	1 + 1	anuery - 1957 -			1 - House and the state	
Sex :	Maie	•			O.	
Nationality :						
Passport Number :						
		Contact Details		1.6 3.5		
Heip the environment by attacting your carbon anisalona					Bent You Month Blue	
Home Telephone :		t and farming the second second			Follows	
3 4 Mobile Telephone :						
Email Address :						

Figure 5.5: Travelers' Details Window for Flexible Travelers

5.9.1.5 Payment Window for Flexible Travelers

Users will be requested to make payment with discounted air fare using the possible options as shown in Figure 5.6.

Booking > Flight Re Ø You are a flex	ouirement >	<u>Travellers Details</u> > ger	Payment Meth	od
FREE		Flight		
From	To	Flexible Days	Discount	Class
Kuala Lumpur	Dubai	6	20%	Economy
Dell'Estate States		Payment Details	the second	
Card Type :		Master Card +		
Card Number :				
Card Holder :				
Card Holder Tele	phone :			

Figure 5.6: Payment Method for Flexible Travelers

5.9.1.6 Immature Ticket Window for Flexible Travelers

Users will be issued a receipt of payment as shown in Figure 5.7. However, the allocation of flight out of the given flights will be done 10 days before the departure date.

AVER AND	T	raveilers E	etalls (Flexib	le)		100 - 2
ervices	Booking 1D :		A1001			
ontact Us	Passport Nu	mber :	12345678910	0		
Advertisement	Name :		Miss Gehan 1	rahya Alsala		
The area	From :		Kuala Lumpu	ır		Nama-Tank
Same Con	To :		Dubai			Sugara and
Most the team	Class Type :		Economy			1
who go the extra nille	Flexible Day	51	6			0
AMPRICAL EXPORTS	Discount :		20%			trav
And and a state of the state of	Total to pay	:	MYR1725.60	0		1
	Po	ssible Dat	es for your Fli	ght		Beat
y offseting No.	From	То	Date	Time	Status	Monteles
missions FLY	Kuala Lumpur	Dubai	12/11/10	00:30	Available	Billing
2	Kuala Lumpur	Dubai	14/11/10	13:00	Available	Follows us
and the second se			() = 10 C 10 C	1.000		

Figure 5.7: Immature Ticket for Flexible Travelers

5.9.1.7 Flight Search Window for Inflexible Travelers

Flight search with *Inflexible Flight Type* as discussed in Section 5.9.1.2 will let the users to the searched window with available flights as shown in Figure 5.8.



Figure 5.8: Flight Search Window for Inflexible Travelers

5.9.1.8 Travelers' Details Window for Inflexible Travelers

Traveler will be requested to fill in the required information for the booking as shown in Figure 5.9.

bout Us	• You are an	inflexible	e passer	nger				L'ACH A
CEV PLASES				Flight				ALC: N
ontact us	Flight ID	Fro	m	То	Date	Time	Class	s
dvertisement	FA100	Kuala L	umpur	Dubal	12/11/10	00:30	Economy	Contraction of the local division of the
1 . A. A. A.	Total to pay :					MYR2157	.00	Sec. 1
200		an barren	-D	ead Trave	ller			S. Lord Landson
Meet the Learn-	Full Name:		Mr 🕳	I				10 million and and and and and and and and and an
ANCOME NEPOPO	Date of Bir	th	1 -	January	• 1957 •			travel
"Yanda	Gender :	Gender :		*				
- LUSA MERCE	Nationality	<i>i</i> :						
	Passport N	umber :		11111				Beat
offsatting		Contact Details						Uour
wour carbon PLY	Γ		1					Bines
. ×	Address :							Follows us or
1 合用用	Home Teleph	none :						
	Mobile phone	e:						
	Email Addres	is :	-					

Figure 5.9: Travelers' Details Window for Inflexible Travelers

5.9.1.9 Payment Window for Inflexible Travelers

Users will be requested to make payment with actual fare using the possible options as shown in Figure 5.10.

FiggT Arrive Getaway			Ċ				1
ne king nut Us	Booking > Flight Method Oyou are an	<u>Requirement</u> > <u>F</u>	light Price >	Travellers (<u>Details</u> > Pa	ayment	FlaxT Gaile
vices			Flight				A STATE
tact Us	Flight ID	From	То	Date	Time	Class	- 5
	FA100	Kuala Lumpur	Dubai	12/11/10	00:30	Economy	
vertisement	Total to pay :				MYR2157	.00	Sime
1		p	ayment De	etails			Distantion of the local distance
Ment the team	Card Type :		Master Card	•			1 100000
An una erita usie	Card Number	1					0
version headers	Card Holder :						trav
Le Con	Card Holder Telephone :						
the environment	Please tic	k this box to ac	cept the	Back		Book	Beat

Figure 5.10: Payment Method for Inflexible Travelers

5.9.1.10 Ticket Window for Inflexible Travelers

Users will be issued a ticket with the allocation of flight, seat and other relevant information as shown in Figure 5.11.

Your Airline Getaways		and an and the second	
ome Bo ooking Ma bout Us	oking > Flight Requirement ethod > Travellers Print Deta	> <u>Flight Price</u> > <u>Travellers Details</u> > <u>Pavment</u> is	FlaxT Galler
ervices	Trave	llers Details (Inflexible)	200
ontact Us	Booking ID :	A1001	
COLORIS DISCOUT	Seat No :	FAA10	States and a state of the state
Advertisement	Passport Number :	12345	5
	Name :	Miss Dayang Rohaya Binti Awang Rambli	Contraction of the local division of the loc
Mean the teams	Flight ID :	FA100	1 States
who go the extra mile	From :	Kuala Lumpur	
ANULINI REPLIKE	то:	Dubal	trave.
	Date :	12/11/10	1
Level 1	Time :	00:30	
- Charles States	Class Type :	Economy	
100 million (100 m	and the second	100000000	BERL

Figure 5.11: Confirmed Ticket for Inflexible Travelers

5.10 Chapter Summary

Results obtained during the study were discussed in this chapter. Results supported most of the hypothesis achieving all the three research objectives. In the first place System's Flexibility and Users' Flexibility were discussed to investigate the user needs associated with FOARS. This was followed by the discussion on the classification of users on the basis of their Flexible Traveling Behavior. Later, the results obtained from a case study to test the validity of the proposed framework for designing more FOARS was discussed. This study concludes that although System's Flexibility and Users' Flexibility are unequivocally independent of another, in terms of designing flexible Online Airline Reservation Systems, they are part of single quantum. While User's Flexibility is related to their purchase making decisions, Systems Flexibility, although is reflected in its Perceived Usability, can influence upon user's traveling behavior by enhancing their satisfaction. Finally, the recommendations for the designing of a more flexible Online Airline Reservation Systems were discussed.

CHAPTER 6

CONCLUSIONS

6.0 Chapter Overview

This chapter presents the summary of the thesis with major findings. Section 6.1 presents the dissertation summary while summarizing the overall thesis. Section 6.2 is dedicated for the major research findings and Section 6.3 is devoted for the future work.

6.1 Dissertation Summary

Behavioral characteristics in terms of making travelers flexible are appealing and an important area of research. This research provides a comprehensive study focused on what makes a traveler flexible on the basis of his/her behavioral characteristics, hence, giving a crucial insight for designing of flexible Online Airline Reservation Systems. This research therefore, contributes to the small but growing literature of designing flexible Online Airline Reservation Systems and provides a general framework by molding upon flexible behavior of travelers.

In this research a framework for flexible Online Airline Reservation Systems has been proposed, based on integrating opaque characteristics into the SBTs of airlines, to address the research gap as stated in Chapter 1. The framework is based on categorizing travelers on the basis of their flexible traveling behavior. Travelers with least flexible behaviour may choose confirmed schedule offered by the airline with normal fare. Least flexible customers will get their confirmed tickets with seats allocated and itinerary finalized at the time of booking. Since flexible travelers are flexible enough and have not indicated their preferred dates to fly, therefore, they can be requested to provide their preferred time-frames in which they would like to travel instead of providing fixed schedule. Tickets to highly flexible customers will be issued at booking time but the seats and flight day will be notified only a week before their actual departure day. Thus, it becomes the responsibility of the arbitrator systems to manage and finalize their traveling itineraries as per their indicated time-frames. Once done, the flexible customers can be allocated seats a few days before their actual departure by the airline.

To develop the framework, existing Online Airline Reservation Systems were examined to assess the flexibility and usability of the systems. The overall research was divided into three phases and each phase contains one core research objective which was achieved through quantitative and qualitative techniques to assess System's Flexibility, Users' Flexibility and Usability of the systems. A redesign solution for enhanced usability was developed based on HCI guidelines and the flexibility tactics used in online travel agencies. A new flexible Online Airline Reservation System design was applied, which led to a proposed interface with the integration of opaque mechanism. The two interfaces were used in the case study. Participants were requested to complete the evaluation of the existing and proposed interfaces.

The benefit of framework given in this research can be used as a baseline for further studies and to design more flexible Online Airline Reservation Systems. Moreover, the Users' Flexibility measuring scale established and tested in this research is also a significant research contribution for future studies to measure flexibility. This research builds on previous and ongoing work within the disciplines of Human Computer Interaction by introducing psychometric scales to measure Users' Flexibility in terms of compromising on service quality attributes of an airline. Furthermore, this research introduces a new approach of reservations to increase the passenger load factor by leveraging upon travelers' flexibility. The Information System (IS) theory given through qualitative reporting is based on empirical findings, which is novel in its own right and testable in different environments.

6.2 Research Findings and Contributions

This study concludes that although System's Flexibility and Users' Flexibility are unequivocally independent of another, in terms of designing flexible Online Airline Reservation Systems, they are part of single quantum. While User's Flexibility is related to their purchase making decisions, Systems Flexibility, is reflected in its adaptability, adaptivity and personalisation. When considered together, the two can immensely enhance systems' perceived usefulness, as reflected in its effectiveness, efficiency and satisfaction.

The overall aim of this research was to investigate the associations between System's Flexibility, Users' Flexibility and Perceived Usability of Online Airline Reservation Systems and to propose a framework for flexible Online Airline Reservation Systems. This research has been conducted to answer the research questions and achieve the desired research objectives. The objectives were used as guidelines for the development and analysis of proposed framework. The findings of this research based on the objectives are summarized below:

6.2.1 Objective 1

To assess user needs (System's Flexibility and Users' Flexibility) associated with Online Airline Reservation Systems.

Findings – The literature was reviewed that revealed the facts that OTA's success immensely depends upon hidden or opaque characteristics. It was further revealed that an opaque characteristic depends on flexible behavior of travelers. Hence, it is important for the airlines to design their self-booking tools in view of customers' preferences, expectations and online usage behavior to run successful business. Therefore, the same concept of opaque mechanism was adopted for integration into SBTs of airlines and considered a potential research gap. The framework proposed in this study moulds upon flexible traveling behaviour of users' too and integrated opaque characteristics in designing of flexible Online Airline Reservation Systems.

Since flexibility is referred to its ability to respond to internal or external changes. Change can be defined as the transition over time which requires change a gent. Researchers argued that "if the change agent is external to the system, then the change under consideration is a *flexible*-type change" [12]. Therefore, in case of SBTs incorporated with opaque fares would serve the role of external change agent by way of providing flexibility in users' decision making. Similarly, "if the change agent is internal to the system, then the change under consideration is an *adaptable*-type change" [12]. Thus the provision of opaque fares into SBTs also serves the role of internal change agent by way of providing the capability of accepting changed decisions. If no change agent exists, then the system is rigid (no change can occur). Since provision of opaque fares could make users flexible and also increases the adaptability of the system, it is expected that the usability of the system would be enhanced.

The opaque fare mechanism depends on hidden characteristics of the traveling plan, thus leveraging upon traveling behavior of leisure travelers, who are always up for grabs and less sensitive to traveling plans. The findings showed that it is highly recommended by the experts in the airline industry for integration into SBTs.

6.2.2 Objective 2

To propose a framework for designing more flexible Online Airline Reservation Systems while classifying users on the basis of their Flexible Traveling Behavior.

Findings – The research revealed a framework for flexible Online Airline Reservation System based on the research scope as presented in Chapter 1. The framework is a general framework that can be applied to different reservation systems particularly, the Airline Reservation Systems due to their immense profit making potential and research gap as identified in Chapter 2 and framework methodology in Chapter 3. The framework is based upon investigating three research variables namely Users' Flexibility, System's Flexibility and Perceived Usability of Online Airline Reservation System. User's Flexibility is based upon the extent to which they can forgo or compromise on service quality attribute of the airline. By doing this users'

classification based on their mental models was done for randomized experimentation of the existing and proposed interfaces. As for System's Flexibility the proposed and existing interfaces used in experimentation differed by way of leveraging upon opaque characteristics as offered by OTAs. The opaque characteristics in the proposed flexible systems will positively enhance perceived usefulness of the systems.

6.2.3 Objective 3

To study the interrelationship between System's Flexibility, Users' Flexibility and Perceived Usability of Online Airline Reservation Systems and to determine the Perceived Usability of the existing and proposed systems.

Findings – The research findings on the interrelationship between Users' Flexible Behavior and Perceived Usability showed positive relationship. This means that as one variable increases in value, the second variable also increases in value. This is called a positive correlation. The significance value indicates that the relationship between the two variables is genuine, hence both variable share positive relationship between each other. The overall relationship between User's Flexible Behavior and System's Flexibility showed non-significant values. However, there were significant relationships between User's Flexible Behavior and the two components of the System's Flexibility i.e. System's Adaptability and System's Personalization. Therefore, both variables i.e. System's Flexibility and Users' Flexibility are partially associated.

Furthermore, findings proved that System's Flexibility predicts Perceived Usability of the Online Airline Reservation Systems. What was more interesting is that when the effect of System's Flexibility was added, the effect of Users' Flexible Behavior remained significant towards predicting Perceived Usability of flexible Online Airline Reservation System.

Moreover, the findings of the study showed significant differences on effectiveness, efficiency and satisfaction with existing and proposed systems among the classified users on the basis of their Flexible Traveling Behavior. When the influence of existing and proposed system was examined on perceived effectiveness, efficiency and satisfaction there was a significant main effect with effectiveness and satisfaction. This effect means that if we ignore the classification of users on the basis of their Flexible Traveling Behavior, the proposed and existing systems influenced upon users' perceived effectiveness and satisfaction. However, there was no significant effect of efficiency showing that if we ignore the classification of users on the basis of their flexible traveling behavior, the proposed and existing systems does not influenced upon users' perceived efficiency.

6.3 Future Work

The computerized translation of customers' flexible and inflexible traveling behavior into tailored and meaningful actionable results is currently not supported by any of the reservation systems in place. Moreover, a lot of the process involved in preparing tailor-made arrangements to cater for flexible and inflexible behavior of travelers can be automated – or at the very least significantly be optimized through technology. The current limitation of the research work is, a very basic Graphical User Interface design for the proposed framework has been developed and implemented. However, for giving flexible travel options early in the Online Airline Reservation Systems through formal opaque quoting mechanism, a very fluid interface based on arbitrator system between human input and database populated emails and documents is a potential research area and it can further be standardised in a way which would work for lots of traveling agencies and companies operating with different kinds of reservation systems.

To proceed further, one can create a layer on top of the standard system – everybody is booked onto a specific flight, but some days before the departure date the additional layer looks for better options and reshuffles the passengers in the legacy system below.

REFERENCES

- [1] A. Schulz, "The Role of Global Computer Reservation Systems in the Travel Industry Today and in the Future," *Tourism*, vol. 6, no. 2, pp. 17–20, 1996.
- [2] D. Quinby, "Travel Agency Distribution Landscape: Key Terms and Definitions," PhoCusWright Inc., Sherman, CT 06784-1430 USA, Apr. 2008.
- [3] B. Braham, "Computer Systems in the Hotel and Catering Industry," London: Cassell, 1988.
- [4] P. O'Connor, "Tourism and Hospitality Electronic Distribution and Information Technology," Oxford: CAB International, 1999.
- [5] P. Sheldon, "Tourism Information Technology," Oxford: CAB International, 1977.
- [6] The Transnational. (2008). Lufthansa, Swiss Add GDS Fees [Online].
 Available: <u>http://www.thetransnational.travel/news.php?cid=Lufthansa-</u> Swiss.Jan-08.24
- B. C. Smith, R. Darrow, J. Elieson, D. Gunther, B. V. Rao and F. Zouaoui,
 "Travelocity Becomes a Travel Retailer," *Interfaces*, vol. 37, no. 1, pp. 68–81, 2007.
- [8] Retail in Asia. (2010). Why Malaysia Tourism is Thriving [Online]. Available: <u>http://www.retailinasia.com/article/markets/malaysia/2010/10/why-</u>malaysia-tourism-thriving
- [9] D. Quinby and R. Merten, "The Role and Value of the Global Distribution Systems in Travel Distribution," PhoCusWright Inc., Sherman, CT 06784-1430 USA, Nov. 2009.
- [10] K. Jerath, S. Netessine and S. K. Veeraraghavan, "Revenue Management with Strategic Customers: Last-Minute Selling and Opaque Selling," *Management Science*, vol. 56, no. 3, pp. 430–448, Mar. 2010.
- [11] J. Sviokla, "Value Poaching," Across the Board (March/April 2003): 11–12.

- [12] A. M. Ross, D. H. Rhodes, and D. E. Hastings, "Defining System Changeability: Reconciling Flexibility, Adaptability, Scalability, and Robustness for Maintaining System Lifecycle Value," 17th INCOSEInternational Symposium, San Diego CA, Jun. 2007.
- K. Jerath, S. Netessine and S. Veeraraghavan, "Selling to Strategic Customers: Opaque Selling Strategies", *Tepper School of Business*, Paper 215, 2009. <u>http://repository.cmu.edu/tepper/215</u>
- [14] K. Talluri and G. J. V. Ryzin, *The Theory and Practice of Revenue Management*, Kluwer Academic Publishers, NY, 2004.
- [15] M. Lambert. (2006). Last-Minute Gamble: Take a Chance, But Be Ready for Anything [Online]. Available: <u>http://www.accessmylibrary.com/article-1G1-154985891/last-minutegamble-take.html</u>
- [16] J. Harrison, *Traveling Blind*, Ottawa Citizen: L2, 2006.
- [17] N. Granados, K. Han and D. Zhang, "Demand and Revenue Impacts of the Opaque Channel: Empirical Evidence from the Airline Industry," *Practice Article*, Dec. 3, 2010.
- [18] S. Fay and J. Xie, "Probabilistic Goods: A Creative Way of Selling Products and Services," *Marketing Science*, vol. 27, no. 4, pp. 674–690, 2008.
- [19] D. Post, "Variable Opaque Products in the Airline Industry: A Tool to Fill the Gaps and Increase Revenues," *Journal of Revenue and Pricing Management*, Practice Article, 2010. <u>www.palgrave-journals.com/rpm/</u>
- [20] K. A. Olsen and A. Malizia, "Flexible User Interfaces for B2C Systems, Using Booking as a Case Study," in *Proceedings of the HCI 2005*, Las Vegas, USA, 2007.
- [21] A. Malizia, "Adding Flexibility to B2C Booking Systems Using a Virtual Intermediate Travel Agent," in Proceedings of the IEEE Symposium on Visual Languages and Human-Centric Computing, pp. 337–338, 2005.
- [22] K. Charmaz, "Grounded Theory in the 21st Century," In *Holstein & Gubrium*.Social Action Thousand Oaks, Cal: Sage, 2005.
- [23] R. Miller, B. Johnson and W. W. Woolfolk, "Designing Flexible Information Systems Saves Time and Money, Now and in the Future," *Educause Quarterly*, no. 3, pp. 44–51, 2002.

- [24] N. C. Goodwin, "Functionality and Usability," *Communications of the ACM*, vol. 30, no. 3, 229–233, 1987.
- [25] D. A. Norman, *The Invisible Computer*, Cambridge MA: MIT Press, 1998.
- [26] P. Kalczynski, S. Senecal and J. Nantel, "Predicting Online Task Completion with Clickstream Complexity Measures: A Graph-Based Approach," *International Journal of Electronic Commerce*, vol. 10, no. 3, pp. 123–143, 2006.
- [27] E. Boling, "Usability Testing for Web Sites Learning for the Global Community," in *Seventh Annual Hypermedia Conference*, Bloomington, 1995.
- [28] M. Levi and F. Conrad, "A Heuristic Evaluation of a World Wide Web Prototype," *Interactions*, vol. 3, no. 4, pp. 1–12, 1996.
- [29] J. Nantel and S. Senecal, "The Effect of Counterproductive Time on Online Task Completion," in M. Craig-Lees, G. Gregory and T. Davis (eds.). Advances in Consumer Research Asia Pacific, 2007.
- [30] J. W. Palmer, "Web Site Usability, Design, and Performance Metrics," *Information Systems Research*, vol. 13, no. 2, pp. 151–167, 2002.
- [31] J. Pitkow and C. Kehoe, "Emerging Trends in the WWW User Population," *Communications ACM*, vol. 39, no. 6, pp. 106–108, 1996.
- [32] J. Nielsen, *Designing Web Usability*, Indianapolis: New Riders Publishing, 2000.
- [33] G. Saks. (2006). Travel: The Emergence of Meta Search [Online]. Available: http://blog.compete.com/2006/11/14/meta-search-kayak-sidestep-farechasemobissimo-pinpoint-travel/
- [34] S. Devaraj, M. Fan and R. Kohli, "Antecedents of B2C Channel Satisfaction and Preference: Validating e-Commerce Metrics," *Information Systems Research*, vol. 13, no. 3, pp. 316–331, 2003.
- [35] D. McCloskey, "Evaluating Electronic Commerce Acceptance with the Technology Acceptance Model," *Journal of Computer Information Systems*, vol. 44, no. 2, pp. 49–57, 2004.
- [36] T. P. Monsuwe, B. G. C. Dellaert and K. Ruyter, "What Drives Consumers to Shop Online? A Literature Review," *International Journal of Service Industry Management*, vol. 15, no. 1, pp. 102–121, 2004.

- [37] M. Limayem, M. Khalifa and A. Frini, "What Makes Consumers Buy from Internet? A Longitudinal Study of Online Shopping," *IEEE*, vol. 30, no. 4, 2000.
- [38] N. M. Shariff and M. Z. Shah, "Factors Influencing Travel Behavior and their Potential Solution: A Review of Current Literatures", *Jurnal Alam Bina*, vol. 11, no.2, pp. 19–28, 2008.
- [39] M. Manzari, "Intention to Use Internet Reservation Systems by Iranian Airline Passengers," Master's Thesis, Lulea University of Technology, 2008.
- [40] K. Jerath, S. Netessine and S. Veeraraghavan, "Selling to Strategic Customers: Opaque Selling Strategies," in *Consumer-Driven Demand and Operations Management Models*, Springer, pp. 253–300, 2009.
- [41] K. Jerath, S. Netessine and S. Veeraraghavan, "Selling to Strategic Customers: Opaque Selling Strategies," in *Consumer-Driven Demand and Operations Management Models* (eds. S. Netessine and C.S. Tang), pp. 253-300, 2009.
- [42] J. Law, A Dictionary of Business and Management, Oxford University Press, 2009. <u>http://www.oup.com/uk/isbn/9780199234899/</u>.
- [43] J. Browne. (1984). Classification of Flexible Manufacturing Systems. The FMS Magazine. vol. 2, no. 2, pp. 114–117.
- [44] A. K. Sethi and S. P. Sethi, "Flexibility in Manufacturing: A Survey," The International Journal of Flexible Manufacturing Systems, vol. 2, pp. 289–328, 1990.
- [45] R. E. Giachetti, L. D. Martinez, O. A. Saenz and C. S. Chen, "Analysis of the Structural Measures of Flexibility using a Measurement Theoretical Framework," *International Journal of Production Economics*, vol. 86, pp. 47– 62, 2003.
- [46] J. H. Saleh, D. E. Hastings and D. J. Newman, "Flexibility in System Design and Implications for Aerospace Systems," *Acta Astronautica*, vol. 53, no. 12, pp. 927–944, 2003.
- [47] A. M. King and S. Sivaloganathan, "Flexible Design: A Strategy for Design Reuse," in *Proceedings of Engineering Design Conference '98 on Design Reuse*, (Eds S. Sivaloganathan and T.M.M. Shahin), Professional Engineering Publishing Limited, London.

- [48] P. K. P. Rajan, M. Van Wie, M. I. Campbell, K. L. Wood and K. N. Otto, "An Empirical Foundation for Product Flexibility," *Design Studies*, vol. 26, no. 4, pp. 405-438, 2005.
- [49] J. Preece, Y. Rogers, H. Sharp, D. Benyon, S. Holland and T. Carey, *Human-computer Interaction*, Workingham, England: Addison-Wesley, 1994.
- [50] R. Nilchiani, "Measuring Space Systems Flexibility: A Comprehensive Sixelement Framework," PhD. Dissertation, Massachusetts Institute of Technology: Department of Aeronautics and Astronautics, Cambridge, MA, 2005.
- [51] A. M. Ross, "Managing Unarticulated Value: Changeability in Multi-Attribute Tradespace Exploration," PhD. Dissertation, Massachusetts Institute of Technology: Engineering Systems Division, Cambridge, MA, 2006.
- [52] J. H. Saleh, "Weaving Time into System Architecture: New Perspectives on Flexibility, Spacecraft Design Lifetime, and On-orbit Servicing," PhD. Dissertation, Massachusetts Institute of Technology: Department of Aeronautics and Astronautics, Cambridge, MA, 2002.
- [53] R. L. Bryck, "Flexible Behavior Under Control? Neural and Behavioral Evidence in Favor of a Two-component Model of Task-switching," PhD. dissertation, Department of Psychology and the Graduate School, University of Oregon, Eugene. 2008.
- [54] P. Barnard, "The Contributions of Applied Cognitive Psychology Toward the Study of Human-computer Interaction," 1995. In R. M. Baecker, J. Grudin, W.A.S. Buxton and S. Greenberg (eds.), Readings in Human-Computer Interaction: Toward the Year 2000. Second Edition (pp. 640-658). San Francisco, CA:Morgan Kaufman Publishers.
- [55] S. K. Card, T. P. Moran and A. Newell, *The Psychology of Human-Computer Interaction*, Hillsday, J: Lawrence Erlbaum, 1983.
- [56] T. R. G. Green, S. P. Davies and D. J. Gilmore, "Delivering Cognitive Psychology to HCI: The Problems of Common Language and of Knowledge Transfer," *Interacting with Computers*, vol. 8, pp. 89–111, 1996.
- [57] T. K. Landauer, "Let's Get Real: A Position Paper on the Role of Cognitive Psychology in the Design of Humanly Usefid and Usable Systems," 1995. In R.M. Baecker, J. Grudin, W.A.S. Buxton and S. Greenberg (eds.), Readings in

Human-Computer Interaction: Toward the Year 2000. Second Edition (pp. 659–665). San Francisco, CA: Morgan Kaufman Publishers.

- [58] E. Pollitzer and E. Edmonds, "Editorial: The Evolving Partnership between Cognitive Science and HCI," *International Journal of Human-Computer Studies*, vol. 44, pp.731–741, 1996.
- [59] R. Guindon, Cognitive Science and its Applications for Human-Computer Interaction, Hillsdale, NJ: Lawrence Erlbaum, 1988.
- [60] M. M. Gardiner and B. Christie, *Applying Cognitive Psychology to User-Interjiace Design*, Chichester, UK: John Wiley & Sons, 1997.
- [61] K. E. Flaherty and J. C. Mowen. (2010). The Role of Schemas in Consumer Behavior Research [Online]. Available: <u>http://onlinelibrary.wiley.com/doi/10.1002/9781444316568.wiem03055/abstra</u> <u>ct</u>
- [62] A. D. Palma and D. Rochat, "Understanding Individual Travel Decisions: Results from a Commuters Survey in Geneva," *Transportation*, vol. 26, pp. 263–281, 1999.
- [63] J. E. Mills and R. Law, "Handbook of Consumer Behavior, Tourism, and the Internet," *Journal of Travel and Tourism Marketing*, vol. 17, no. 2/3, 2004.
- [64] M. Sigala, A. Lockwood and P. Jones, "Strategic Implementation and IT: Gaining Competitive Advantage from the Hotel Reservation Process," *International Journal of Contemporary Hospitality Management*, vol. 17, no. 3, pp. 364–371, 2001.
- [65] W. D. Walls, "Competition in the Market for Travel Information Services," Proceedings of the Eastern Asia Society for Transportation Studies, vol. 5, pp. 533–544, 2005.
- [66] Orbitz. (2010). Flights [Online]. Available: http://www.orbitz.com/App/PrepareFlightsTab
- [67] I. H. Hann and C. Terwiesch, "Measuring the Frictional Costs of Online Transactions: The Case of a Name-Your-Own-Price Channel," *Management Science*, vol. 49, no. 11, pp. 1563–1579, 2003.
- [68] C. Terwiesch, S. Savin and I. H. Hann, "Online Haggling at a Name-Your-Own-Price Retailer: Theory and Application," *Management Science*, vol. 51, no. 3, pp. 339–351, 2005.

- [69] Priceline. (2010). Name Your Own Price [Online]. Available: <u>http://www.priceline.com/</u>
- [70] Hotwire. (2010). Welcome to Flexible Date Search [Online]. Available: <u>http://www.hotwire.com/us/index.jsp</u>
- [71] A. Hill, "Top 5 Reasons Your Customers Abandon Their Shopping Carts," *Ziff Davis Smart Business for the New Economy*, vol. 14, no. 3, pp. 80–85, 2001.
- [72] L. R. Klein, "Evaluating the Potential of Interactive Media through a New Lens: Search vs. Experience Goods," *Journal of Business Research*, vol. 41, pp. 195–203, 1998.
- [73] W. J. Doll, A. Hendrickson and X. Deng, "Using Davis's Perceived Usefulness and Easeof-use Instruments for Decision Making: A Confirmatory and Multigroup Invariance Analysis," *Decision Sciences*, vol. 29, no. 4, pp. 839–869, 1998.
- [74] J. Nielsen, *Designing Web Usability: The Practice of Simplicity*, Indianapolis: New Riders Publishing, 2000.
- [75] J. Nielsen, Homepage Usability: 50 Websites Deconstructed, Indianapolis: New Riders, 2002.
- [76] A. J. Head, Design Wise: A Guide for Evaluating the Interface Design of Information Resources, New Jersey: CyberAge books, 2000.
- [77] S. Krug, *Don't Make me Think!*, Indianapolis: New Riders Publishing, 2000.
- [78] R. Williams and J. Tollett, *The Non-designers Web Book. Berkeley*, CA: Peachpit Press, 2000.
- [79] F. D. Davis, "A Technology Acceptance Model for Empirically Testing New End-User Information Systems: Theory and Results," PhD. dissertation, MIT Sloan School of Management, Cambridge, MA, 1986.
- [80] N. E. Miller, and J. Dollard, Social Learning and Imitation, New Haven: Yale University Press, 1941.
- [81] E. M. Rogers, *Diffusion of Innovations*, 5th ed., New York, NY: Free Press, 2003.
- [82] M. Fishbein and I. Ajzen, *Belief, Attitude, Intention and Behavior: An Introduction to Theory and Research*, Reading, MA: Addison-Wesley, 1975.

- [83] I. Ajzen and M. Fishbein, "Attitude-Behavior Relations: A Theoretical Analysis and Review of Empirical Research," *Psychological Bulletin*, vol. 84, pp. 888–918, 1977.
- [84] I. Ajzen and M. Fishbein, Understanding Attitudes and Predicting Social Behavior, Englewood Cliffs, NJ: Prentice Hall, 1980.
- [85] H. C. Triandis, *Interpersonal Behavior*, Monterey, Calif.: Brooks/Cole Pub. Co, 1977.
- [86] P. Legris, J. Ingham and P. Collerette, "Why Do People Use Information Technology? A Critical Review of the Technology Acceptance Model," *Information & Management*, vol. 40, pp.191–204, 2003.
- [87] R. O. Briggs, M. Adkins, D. Mittleman, J. Kruse, S. Miller and J. F. Nunamaker, "A Technology Transition Model Derived From Field Investigation of GSS Use Abroad the U.S.S. CORONADO," *Journal of Management Information Systems*, vol. 15, no. 3, pp. 151–195, 1999.
- [88] D. S. Roberston, "Social Determinants of Information System Use," Journal of Management Information Systems, vol. 5, no. 4, pp. 55–71, 1989.
- [89] D. Graber, Processing the News: How People Tame the Information Tide, 2nd ed., New York: Longman, pp. 119 & 133, 1988.
- [90] Y. Lee, K. A. Kozar and K. R. T. Larsen, "The Technology Acceptance Model: Past, Present, and Future," *Communications of the Association for Information Systems*, vol. 12, no. 50, pp. 752–780, 2003.
- [91] H. C. Lucas and V. K. Spitler, "Technology Use and Performance: A Field Study of Broker Workstations," *Decision Sciences*, vol. 30, no. 2, pp. 291– 311, 1999.
- [92] V. Venkatesh and F. Davis, "A Theoretical Extension of the Technology Acceptance Model: Four Longitudinal Field Studies," *Management Science*, vol. 46, no. 2, pp. 186–204, 2000.
- [93] D. Robey, "Research Commentary: Diversity in Information Systems Research: Threat, Promise, and Responsibility," *Information Systems Research*, vol. 7, pp. 400–408, 1996.
- [94] Y. Malhotra and D. F. Galletta, "Extending the Technology Acceptance Model to Account for Social Influence: Theoretical Bases and Empirical Validation,"

in Proceedings of the 32nd Hawaii International Conference on System Sciences, Hawaii, 1999.

- [95] M. Y. Chuttur, "Overview of the Technology Acceptance Model: Origins, Developments and Future Directions," Indiana University, USA. Sprouts: Working Papers on Information Systems, vol. 9, no. 37, 2009. <u>http://sprouts.aisnet.org/9-37</u>
- [96] V. Venkatesh, "Determinants of Perceived Ease of Use Integrating Control, Intrinsic Motivation, and Emotion into the Technology Acceptance Model," *Information Systems Research*, vol. 11, no. 4, pp.342–365, 2000.
- [97] P. Benckendorff, "An Exploratory Analysis of Traveler Preferences for Airline Website Content," *Information Technology & Tourism*, vol. 8, pp. 149–159, 2006.
- [98] S. Balgolu and Y. A. Pekcan, "The Website Design and Internet Site Marketing Practices of Upscale and Luxury Hotels in Turkey," *Tourism Management*, vol. 27, no. 1, pp. 171–176, 2006.
- [99] P. J. Benckendorff and N. L. Black, "Destination Marketing on the Internet: A Case Study of Australian Regional Tourism Authorities," *Journal of Tourism Studies*, vol. 11, no. 1, pp. 11–21, 2000.
- [100] L. Cai, J. A. Card and S. T. Cole, "Content Delivery Performance of World Wide Web Site of US Tour Operators Focusing on Destination in China," *Tourism Management*, vol. 25, pp. 219–227, 2004.
- [101] R. Law and R. Leung, "A Study of Airlines' Online Reservation Services on the Internet," *Journal of Travel Research*, vol. 39, pp. 202–211, 2000.
- [102] J. Murphy, E. J. Forrest, C. E. Wotring and R. A. Brymer, "Hotel Management and Marketing on the Internet," *Cornell Hotel and Restaurant Administration Quarterly*, vol. 37, no. 1, pp. 61–71, 1996.
- [103] C. Wan, "The Web Sites of International Tourist Hotels and Tour Wholesalers in Taiwan," *Tourism Management*, vol. 23, pp. 155–160, 2002.
- [104] A. M. Morrison, J. S. Taylor and A. Douglas, "Website Evaluation in Tourism and Hospitality: The Art is Not Yet Stated," *Journal of Travel and Tourism Marketing*, vol. 17, no. 2/3, pp. 233–251, 2004.

- [105] A. Scharl, K. W. Wober and C. Bauer, "An Integrated Approach to Measure Web Site Effectiveness in the European Hotel Industry," *Information Technology & Tourism*, vol. 6, pp. 257–271, 2004.
- [106] J. A. Siguaw, C. A. Enz and K. Namasivayam, "Adoption of Information Technology in US Hotels: Strategically Driven Objectives," *Journal of Travel Research*, vol. 39, pp. 192–201, 2000.
- [107] Y. Yuan, U. Gretzel and D. R. Fesenmaier, "Internet Technology Use by American Convention and Visitors Bureaus," *Journal of Travel Research*, vol. 41, pp. 240–255, 2003.
- [108] H. Nysveen, L. B. Methlie and P. E. Pedersen, "Tourism Websites and Value-Added Services: The Gap between Customer Preferences and Web Sites' Offerings," *Information Technology & Tourism*, vol. 5, pp. 165–174, 2003.
- [109] J. A. Card, C. Chen and S. T. Cole, "Online Travel Products Shopping: Differences between Shoppers and Non-shoppers," *Journal of Travel Research*, vol. 42, pp. 133–139. 2003.
- [110] A. M. Susskind, M. A. Bonn and C. S. Dev, "To Look or Book: An Examination of Consumers' Apprehensiveness toward Internet Use," *Journal* of Travel Research, vol. 41, pp. 256–264, 2003.
- [111] S. Jang, "The Past, Present and Future Research of Online Information Search," *Journal of Travel and Tourism Marketing*, vol. 17, no. 2/3, pp. 41– 47, 2004.
- [112] M. Luo, R. Feng and C. Liping, "Information Search Behavior and Tourist Characteristics: The Internet vis-a-vis other Information Sources," *Journal of Travel and Tourism Marketing*, vol. 17, no. 2/3, pp. 15–25, 2004.
- [113] B. Bai, C. Hu, J. Elsworth and C. Countryman, "Online Travel Planning and College Students: The Spring Break Experience," *Journal of Travel and Tourism Marketing*, vol. 17, no. 2/3, pp. 79–91, 2004.
- [114] A. Rudstrom and P. Fagerberg, "Socially Enhanced Travel Booking: A Case Study," *Information Technology & Tourism*, vol. 6, pp. 211–221, 2004.
- [115] S. Klein, F. Kohne and A. Oorni, "Barriers to Online Booking of Schedule Airline Tickets," *Journal of Travel and Tourism Marketing*, vol. 17, no. 2/3, pp. 27–39, 2004.

- [116] C. H. Marcussen, "The Effects of Internet Distribution of Travel and Tourism Services on the Marketing Mix: No-frills, Fair Fares, and Fare Wars in the Air," *Information Technology & Tourism*, vol. 2, pp. 197–212, 1999.
- [117] G. Gallego and R. Phillips, "Revenue Management of Flexible Products," Manufacturing and Service Operations Management, vol. 6, no. 4, pp. 321– 337, 2004.
- [118] Y. Jiang, "Price Discrimination with Opaque Products," *Journal of Revenue and Pricing Management*, vol. 6, no. 2, pp. 118–134, 2007.
- [119] R. J. Deneckere and R. P. McAfee, "Damaged Goods," Journal of Economics and Management Strategy, vol. 5, pp. 149–174, 1996.
- [120] X. Huang and G. Sošić, "Name-Your-Own-Price as a Competitive Secondary Channel in the Presence of Posted Prices," *Working Paper*, Concordia University, Montreal, Canada, 2010.
- [121] S. Fay, "Selling an Opaque Product through an Intermediary: The Case of Disguising One's Product," *Journal of Retailing*, vol. 84, no. 1, pp. 59–75, 2008.
- [122] N. Granados, A. Gupta and R. J. Kauffman, "Designing Online Selling Mechanisms: Transparency Levels and Prices," *Decision Support Systems*, vol. 45, no. 4, pp. 729–745, 2008.
- [123] S. L. Puller, A. Sengupta and S. N. Wiggins, "Testing Theories of Scarcity Pricing and Price Dispersion in the Airline Industry," *Working Paper*, Texas A&M University, 2008.
- [124] N. Granados, A. Gupta and R. J. Kauffman, "Information Transparency in Business-to-Consumer Markets: Concepts, Framework, and Research Agenda," *Information Systems Research*, vol. 21, no. 2, pp. 207–226, 2010.
- [125] D. Shapiro and X. Shi, "Market Segmentation: The Role of Opaque Travel Agencies," *Journal of Economics & Management Strategy*, vol. 17, no. 4, pp. 803–837, 2008.
- [126] G. Ng-Kruelle, P. Swatman and O. Kruelle, "e-Ticketing Strategy and Implementation in an Open Access System: The case of Deutsche Bahn," 2006. http://www.insyl.unisa.edu.au/publications/working-papers/2006-08.pdf
- [127] J. P. Schwieterman, "Fare is Fair in Airline Deregulation: The Decline of Price Discrimination," *Regulation*, vol. 9, no. 4, pp. 32–38, 1985.

- [128] L. A. Garrow, "Online Travel Data: A Goldmine of New Opportunities," Journal of Revenue and Pricing Management, vol. 8, no. 2, pp. 247–254, 2009.
- [129] W. J. Carroll, R. J. Kwortnik and N. L. Rose, "Travel packaging: An Internet Frontier," *Cornell Hospitality Report*, vol. 7, no. 17, pp. 4–16, 2007.
- [130] J. Doerr, D. Kerkow, T. Koenig, T. Olsson and T. Suzuki, "Non-functional Requirements in Industry – Three Case Studies Adopting an Experience-based NFR Method," in *Proceedings of the 2005 13th IEEE International Conference on Requirements Engineering (RE'05)*, Washington, DC, USA, pp. 373–384.
- [131] Y. Y. Thaw, A. K. Mahmood and P. D. D. Dominic, "A Study on the Factors that Influence the Consumers' Trust on E-Commerce Adoption," *International Journal of Computer Science and Information Security (IJCSIS)*, vol. 4, no. 1 and 2, pp. 153–159, 2009.
- [132] J. P. Chin, V. A. Diehl and K. L. Norman, "Development of an Instrument Measuring User Satisfaction of the Human-computer Interface," in Proceedings of the SIGCHI Conference on Human Factors in Computing Systems CHI '88, ACM, New York, NY, USA, 1988, pp. 213–218.
- [133] H. X. Lin, Y. Y. Choong and G. Salvendy, "A Proposed Index of Usability: A Method for Comparing the Relative Usability of Different Software Systems," *Journal of Behaviour and Information Technology*, vol. 16, no. 4/5, pp. 267– 278, 1997.
- [134] J. R. Lewis, "IBM Computer Usability Satisfaction Questionnaires: Psychometric Evaluation and Instructions for Use," *International Journal of Human-Computer Interaction*, vol. 7, no. 1, pp. 57–78, 1995.
- [135] N. Burns, S. Grove, "Introduction to Quantitative Research," in *The Practice of Nursing Research: Conduct, Critique, and Utilization*, 2nd ed. Philadelphia: WB Saunders Company, 1993.
- [136] C. F. Waltz, R. B. Bausell, "Quantitative Research," in Nursing Research: Design, Statistics and Computer Analysis, 2nd ed. Philadelphia: FA Davis Company; 1983.
- [137] J. Aldrich, "Correlations Genuine and Spurious in Pearson and Yule," Statistical Science, vol. 10, no. 4, pp. 364–376, 1995.

- [138] J. P. Chin, V. A. Diehl and K. L. Norman, "Development of an Instrument Measuring User Satisfaction of the Human-Computer Interface," in CHI '88 Proceedings of the SIGCHI conference on Human factors in computing systems, pp. 213–218.
- [139] H. X. Lin, Y. Y. Choong and G. Salvendy, "A Proposed Index of Usability: A Method for Comparing the Relative Usability of Different Software Systems," *Behavior and Information Technology*, vol. 16, no. 4/5, pp. 267–278, 1997.
- [140] P. Donald, "Behavioral Criterion," in *Essential Personality*. Hodder Arnold, 2003.
- [141] S. A. McIntire and L. A. Miller, *Foundations of Psychological Testing*, 2nd ed. Thousand Oaks, CA: Sage, 2005.
- [142] O. Johnson, "About the Information Theory and the Central Limit Theorem," in *Information Theory and the Central Limit Theorem*. Imperial College Press, 2004, p. 88. ISBN 1860944736.
- [143] S. Wright, "Effective Size of Population," in Evolution and the Genetics of Populations, 2nd ed. Chicago: University of Chicago Press, 1984. ISBN 0-226-91039-3.
- [144] M. S. Rahmat. (2000). Grounded Theory Methodology as the Research Strategy for a Developing Country [Online]. Available: <u>http://nbn-resolving.de/urn:nbn:de:0114-fqs0001198</u>
- [145] A. Strauss and J. Corbin, Basics of Qualitative Research: Grounded Theory Procedures and Techniques. Newbury Park, CA: Sage, 1990.
- [146] B. G. Glaser and A. L. Strauss, "Grounded Theory," in *The Discovery of Grounded Theory: Strategies for Qualitative Research*. New York: Aldine Pub. Co, 1967, p.271.
- [147] R. W. Riley and L. L. Love, "The State of Qualitative Tourism Research," Annals of Tourism Research, vol. 27, no. 1, pp. 164–187, 2000.
- [148] K. Charmaz, "Grounded Theory in the 21st Century," In *Holstein & Gubrium*.Social Action Thousand Oaks, Cal: Sage, 2005.
- [149] Marshall, Catherine and G. B. Rossman, *Designing Qualitative Research*, 3rd Ed. London: Sage Publications, pp. 115–1999.

- [150] A. M. Schonland and P. W. Williams, "Using the Internet for Travel and Tourism Survey Research: Experiences from the Net Traveller Survey," *Journal of Travel Research*, vol. 35, no. 2, 1996.
- [151] A. Decrop, "Triangulation in Qualitative Tourism Research," Tourism Management, vol. 20, no. 1, pp. 157–161, 1999.
- [152] D. K. Padgett, Qualitative Methods in Social Work Research: Challenges and Rewards. Thousand Oaks, CA: Sage, 1998.
- [153] M. Q. Patton, *Qualitative Evaluation Methods*. Beverly Hills, CA: Sage, 1980.
- [154] N. Kowtha and Choon, "Determinants of Website Development: A Study of Electronic Commerce in Singapore," *Information & Management*, vol. 39, no. 3, pp. 227–242, 2000.
- [155] B. D. Wright, "IRT in the 1990s: Which Models Work Best?" Rasch Measurement Transactions, vol. 6, no. 1, pp. 196–200, 1992.
- [156] A. Field. (2005). Project 3: Design a Questionnaire [Online]. Available: http://coe.georgiasouthern.edu/foundations/bwgriffin/edur9131/Andy_Field_re search_project.pdf
- [157] J. P. Ferrando, "Difficulty, Discrimination and Information Indices in the Linear Factor-Analytic Model for Continuous Responses, Applied Psychological Measurement, vol. 33, pp. 9–24, 2009.
- [158] Kelley
- [159] HHS, & GSA. (2006). Research-based Web design and usability guidelines[Online]. Available: <u>http://usability.gov/pdfs/guidelines.html</u>
- [160] J. Nielsen and R. L. Mack, Usability Inspection Methods (Eds.). New York, NY: John Wiley & Sons, 1994.
- [161] N. Brace, R. Kemp and R. Snelgar, "Multiple Regression," in SPSS for Psychologists, 3rd ed. Mahwah, NJ: Lawrence Erblaum, 2006.
- [162] Laerd Statistics. (2011). Statistics [Online]. Available: <u>http://statistics.laerd.com/statistical-guides/descriptive-inferential-statistics.php</u>
- [163] J. L. Rodgers and W. A. Nicewander, "Thirteen Ways to Look at the Correlation Coefficient," *The American Statistician*, vol. 42, no. 1, pp. 59–66, Feb. 1988.

- [164] M. S. Stigler, "Francis Galton's Account of the Invention of Correlation," *Statistical Science*, vol. 4, no. 2, pp. 73–79, 1989.
- [165] W. Q. Meeker and L. A. Escobar, "Reliability Concepts and Reliability Data," in *Statistical Methods for Reliability Data*, Hoboken, New Jersey: Wiley, 1998.
- [166] J. M. Cortina, "What Is Coefficient Alpha? An Examination of Theory and Applications," *Journal of Applied Psychology*, vol. 78, no. 1, pp. 98–104, 1993.
- [167] StatSoft. (2011). Reliability and Item Analysis [Online]. Available: http://www.statsoft.com/textbook/reliability-and-item-analysis/
- [168] StatSoft. (2011). ANOVA/MANOVA [Online]. Available: http://www.statsoft.com/textbook/anova-manova/
- [169] A. Agresti. (2007). Chapter 12 Comparing Groups: Analysis of Variance (ANOVA) Methods [Online]. Available: <u>http://www.stat.ufl.edu/~aa/sta6127/ch12.pdf</u>
- [170] MEDCALC. (2011). Analysis of Covariance [Online]. Available: <u>http://www.medcalc.org/manual/analysis-of-covariance.php</u>
- [171] R. Lowry. (2011). Chapter 17: One-Way Analysis of Covariance for Independent Samples – Part 2 [Online]. Available: <u>http://faculty.vassar.edu/lowry/ch17pt2.html</u>
- [172] Statistically Significant. (2011). MANOVA [Online]. Available: http://www.statisticallysignificantconsulting.com/Manova-Statistic-SPSS.htm
- [173] G. Carey. (1998). Multivariate Analysis of Variance (MANOVA): I. Theory
 [Online]. Available: http://ibgwww.colorado.edu/~carey/p7291dir/handouts/manova1.pdf
- [174] G. G. David. (2011). Multivariate GLM, MANOVA, and MANCOVA[Online]. Available: <u>http://faculty.chass.ncsu.edu/garson/PA765/manova.htm</u>
- [175] UCLA: Academic Technology Services. (2011). Stata Annotated Output -MANOVA. Available: <u>http://www.ats.ucla.edu/stat/stata/output/Stata_MANOVA.htm</u>
- [176] J. P. Stevens, "Multivariate Analysis of Variance," in Applied Multivariate Statistics for the Social Sciences, 4th ed. Mahwah, NJ: Lawrence Erblaum, 2002.
- [177] F. Galton, "Kinship and Correlation," in *Statistical Science*, vol. 4, no. 2, pp. 80–86, 1989. <u>http://www.jstor.org/stable/2245330</u>
- [178] G. U. Yule, "On the Theory of Correlation," *Journal of the Royal Statistical Society*, vol. 60, no. 4, pp. 812–54, 1987.
- [179] K. Pearson, G. U. Yule, N. Blanchard and A. Lee, "The Law of Ancestral Heredity," *Biometrika*, vol. 2, no. 2, pp. 211–236, 1903.
- [180] MEDCALC. (2011). Multiple Regression [Online]. Available: <u>http://www.medcalc.org/manual/multiple_regression.php</u>
- [181] Stevens. (1999). Post Hoc Tests in ANOVA [Online]. Available: <u>http://pages.uoregon.edu/stevensj/posthoc.pdf</u>
- [182] N. Crichton, "Methodological Issues in Clinical Research," Journal of Clinical Nursing, vol. 10, pp. 707–715, 2001.
- [183] UNESCO. (2011). Non-parametric Measures of Bivariate Relationships: Kendall's Tau [Online]. Available: http://www.unesco.org/webworld/idams/advguide/Chapt4_2.htm
- [185] Skeeterssays. (2008). On the Use of Error Bars [Online]. Available: http://skeetersays.blogspot.com/2008/07/on-use-of-error-bars.html
- [186] JMP. (2008). What Good are Error Bars? [Online]. Available: <u>http://blogs.sas.com/jmp/index.php?/archives/127-What-Good-Are-Error-Bars.html</u>
- [187] J. Pitocchelli. (2001). Multiple Comparisons Tests [Online]. Available: <u>http://www.anselm.edu/homepage/jpitocch/anova/multcomp.html</u>
- [188] Airport International. (2011). How Important is Ticket Flexibility Really Necessary for Business Travellers? [Online]. Available: <u>http://www.airportint.com/article/how-important-is-ticket-flexibility-really-necessary-forbusiness-travellers.html</u>
- [189] G. Pell. (2010). Top 5 Reasons Flights Get Canceled [Online]. Available: http://www.splendora.com/five_things_on_my_radar_by_splendora_founder_ gina_pell/5_top_5_reasons_flights_get_canceled

LIST OF PUBLICATIONS

Journal Publications

- A. Mushtaq, S. Sulaiman, P. D. D. Dominic, S. Riaz and M. A. T. Khalil, "A Study to Examine If Integration of OTAs Features can make SBTs more Flexible Online Airline Reservation Systems," *Information Technology Journal*, vol. 10, no. 7, pp. 1315–1324, 2011.
- A. Mushtaq, S. Sulaiman and S. Riaz. "Flexibility Concept: Facilitating Post-Sale Changes in Online Reservation Systems," *International Journal on Computer Science and Engineering (IJCSE)*, vol. 1, no. 3, pp. 156–162, 2009.

Conference Punlications

- A. Mushtaq, S. Sulaiman, P. D. D. Dominic and S. Riaz, "A Qualitative Enquiry to Explore the Concept of Users' Perception on Factors Influencing Flexible Traveling Behavior and Flexible Online Airline Reservation Systems," in *Proceedings of the Digital Enterprise and Information Systems*, E. Ariwa and E. El-Qawasmeh (Eds.): DEIS 2011, CCIS 194, pp. 739–753, Springer-Verlag Berlin Heidelberg 2011.
- A. Mushtaq, S. Sulaiman and J. Loviscach, "Issues with Flexibility: A Reason for not Using Online Reservation Systems?" in *Proceedings of the IEEE International Symposium on Information Technology*, vol. 3, pp. 1076–1081, 2010.
- A. Mushtaq, S. Sulaiman, P. D. D. Dominic and S. Riaz, "Green Environment with Flexible Airline Reservation Systems," in *Proceedings of the IEEE International Conference on Intelligence and Information Technology*, Lahore, vol. 1, pp. 488–492, 2010.

APPENDIX A

SURVEY ON EVALUATION OF ONLINE AIRLINE RESERVATION SYSTEMS

SURVEY ON EVALUATION OF ONLINE AIRLINE RESERVATION SYSTEMS

Objective – The objective of this survey is to find out physiological, cognitive social behavior and the experience of the selected group of people who used airline online reservation systems.

The feedback of people may help to propose a better framework which could be used to design more flexible online reservation systems.

Last date of survey form submission - 15 October 2009

Note – Please return the form to the person place from where it was collected before or by the deadline. In case of queries, please do not hesitate to contact the undersigned either by call or email.

Arif Mushtaq PhD Scholar Department of Computer and Information Sciences Universiti Teknologi PETRONAS Bandar Seni Iskandar, 31750 Tronoh, Perak, Malaysia +60195969598 coolafee@yahoo.com

	SECTION A – YOUR DETAILS (Please place a tick (/) at the appropriate box)								
1.	Age	1 16-25 2 26-35	3 36-45 5 56-65 4 46-55 6 Above 65						
2.	Gender	1 Male	2 Female						
3.	Nationality	1 Malaysian 2 EU National	3 Pakistani 4 Others (specify)						
4.	Educational background	1 Post-graduate 2 Master	3 Under-graduate 4 Secondary						
5.	How frequently you shop online?	1 Mostly 2 Occasionally	3 Rarely 4 Never						
6.	Have you ever booked an online airline ticket?	1 Yes (if "Yes") 2 No (if "No" pl	please attempt SECTION B only) lease attempt SECTION C only)						

SECTION B – USERS WITH ONLINE EXPERIENCES (Please place a tick (/) at the appropriate box)

1. Which of the following airline's web-site you have used to book ticket?

		TRAVELLED					
	Name of the Airline	Last six months	Last one year	More than one years			
1.	Malaysia Airline						
2.	Emirates Air						

Э.	British Airways			
4.	Thai Airways			
5.	Lufthansa Airline			
б.	Pakistan International Airline			
- T.	Air Asia			
<u> </u>	Other (please specify name)			

Flexibility of Existing Systems

- 2. Have you everthed to make changes in your traveling dates online?
- 3. Have you everthed to cancel your ticket online?
- 4. Have you everthed to transfer your ticket to someone else online?
- 5. Have you ever thed to correct error (such as typos) in you name or in your address online?

б.	Have you everthed to reserve a ticket for few days with the
	intention to buy it later?

Successful Never do so available

Option not

Unable to

7. Rank the following features from 1 to 5 (1most important -5 least important) you would like to use in online system.

	Rank		Rank
Online changes in traveling schedule		Online correction of typo errors	
Online cancellation of ticket		Online reservation of ticket	
Online transfer of ticket			

Please indicate (tick) your level of agreement to the statement below

Strongly Disagree (SDA)	Disagree (DA)	Neutral (N)	Agree (A)	Strongly Agree (SA)
11	2	3	4	5

Evaluation of Existing Online Systems

- 8. Reading characters on the screen of online reservation web site was hard.
- 9. Organization of information was confusing.
- 10. Sequence of screens was confusing.
- 11. Use of terms throughout system was inconsistent.
- 12. Position of messages on screen was inconsistent.
- 13. Learning to operate the system was difficult.
- 14. Provides effective linkage with other travel-related partners (e.g. link to other airline reservation system in case of connected flights).
- 15. The design for data entry was flexible.
- 16. The web site provides flexible user guidance.

SDA	DA	N	A	5A
1	2	7	4	5
1	2	1	4	3
•••	2	41	4	1.24
1	1	**	4	17.4
l	2	1.44	4	1.1
1	2	3	4	<u>-</u>
1	. 2	4	4	5
I	2	-	4	3
1	2	3	4	

- 17. Completion of processing was clearly indicated.
- 18. It requires the fewest steps possible to accomplish what I want to do with it.
- 19. I can use it without reading instructions.
- 20. Both occasional and regular users would like it.
- 21. I canrecover from mistakes quickly and easily.
- 22. I would prefer to use online system for future booking.
- 23. I would recommend it to a friend.
- 24. Overall, I am satisfied with the support information (online-line help, messages, documentation) when completing the tasks.
- 25. Overall, I am satisfied with the ease of completing the tasks in this scenario.
- 26. Overall, I am satisfied with the amount of time it took to complete the tasks in this scenario.

:	2	3	4	5
1	2	3	4	5
1	2	3	4	3
L	2	2	4	Ť.
1	-	3	4	904 1
1	2	3	4	1
I	2	3	4	11. L
ĩ	3	3	4	ŗ,
1	2	3	4	5
1	2	3	4	Ś

SECTION C - USERS WITHOUT ONLINE	EXPERIENCES
Please indicate (tick) your level of agreement to the	statement below

Strongly Disagree (SDA)	Disagree (DA)	Neurral (N)	Agree (A)	Strongly Agree (SA)
		3	4	5

	Reason for not Using Online Reservation Systems	SDA	DA	N	A	SA
1.	I do not have access to internet.	1	2	3	4	5
2.	I do not know how to book an online ticket.	1	2	3	4	5
3.	I am not comfortable shopping online.	1	2	3	4	5
4.	I do not have credit debit card to make an online booking.	1	2	3	4	5
5.	I do not want to try because chances of frauds are there.	1	2	3	4	5
б.	Online reservation systems are very complicated to use.	1	2	3	4	5
7.	I do not use online systems because if some mistake will occur, how would I correct it?	!	2	3	4	5
S.	I prefer to go to travel agent because they can find a better package for me.	1	2	3	4	5
9.	I prefer to go to travel agent because they are more reliable.	1	2	3	4	5
10.	I prefer to go to travel agent because I can request them to make changes in my flight schedule.	I	2	3	4	5
11.	I prefer to go to travel agent because I can reserve tickets (for few days) without even paying a penny.	1	2	3	4	5

12. Rank the 3 most important reasons (from the above statements 1 to 11) for not using online systems 1.____ 2.____ 3.____

C Thank you very much for completing this questionnaire

APPENDIX B

USERS' FLEXIBLE BEHAVIOR IN TERMS OF COMPROMISING ON SERVICE

QUALITY ATTRIBUTES

Users' Flexible Behavior in term of Compromising on Service Quality Attributes

Objective - The objective of this questionnaire is to find out the percentage of flexible users that will support the flexibility of reservation systems. The feedback received may help in designing more flexible reservation systems.

		Ge (Please place :	neral Informati a tick (-) at the app	on ropriste box)					
1.	Age	1 Below 20 4 41 to 50 ye	years 2 21 ears 5 51	to 30 years to 60 years	3 31 t 6 Abo	io 40 yr ovie 60	ears vears		
2.	Gender	1 Male	2 Fe	male					
3.	Educational Background	1 High Scho 4 Post-gradu	ool 2 Di 1ate	ploma	3 Gra	duate			
4.	Occupation	1Student4Retired	Pro	ofessional	3 Self	fEmpl	oyed		
ā.	Use of Online Tool	1 SBT	20	ГА					
	Please indicate whi	ch of the followir	ng statements de	scribes you in th	e best po	ossible	way?		
	Strongly Disagree	Disagree	Neutral	Agree	Stroz	gly Age	ŧ+		
				· _ ·	_ <u>_</u>]		
~		Identity				_	0	+	++
δ.	l'amnot a frequent flyer, l	fly only once in a	year or so.						
	I normally fly to spend holi	days with family	friends.						
8.	I amnot a status conscious	person: I can fly 1	ising any class.						
	n an an an an An Anna An Anna Anna Anna	Self-efficacy					0	+	++
9.	I am excellent in computer	skills; Icanbook	my itinerary in a	very short time	<u> </u>			L	
10.	I am a regular news reader;	I alwayskeep my	selfupdated on t	ourism.	Ł				
11.	I am excellent in managing	trips; I can plan n	ny trip in a very s	short time notice					
	Attribution (paramet	ers that contribu	ites in your rese	rvations)		-	Q.	+	++
12.	Airline's repute does not fa reservations.	scinate me a lot: ii	t plays a minor re	ole in my					
13.	I do not always prefer my c	syn country airline	e while reserving	my tickets.					
14.	I do not always prefer to fly	during occasion:	3.						
15.	. I do not always prefer to fly from my own town; I can take a flight from my neighbor town.								
16.	I do not always prefer airlin	es with better leg	space area and o	comfortabl e seat	s }			1	
17	I do not alwaysprefer airlin climb the gangway and wai	es which provide t for busses.	jet ways, so that	I don't have to					
e	na 1824 per companya ang tanàng ang taona 2010. Ikang kanang ang taona 2010 ang taon		aan ing ay ay sa	er to a su a s		···			

Engagement		-	0	+	++
18. The availability of rewards from customer loyalty programs motivates me to					
use the same airline for my next Journey. 19. Staff attitude not always motivates me to use the same airline for my next					┞──-
journey.	 	[
 I prefer airlines which contributes their efforts towards carbon free environment. 	ļ				ĺ
Persuasion (criteria for choosing a flight)		-	0	+	++
21. I do not always fly <i>duri</i> ng weekends.					
22. I do not always use the same airline my ancestors uses used.					
23. I easily get persuaded by cheap tickets; cheap tickets are at highest priority.					
24. I do not always get persuaded by more baggage weight allowed.					
25. I do not always prefer airlines with best food services; I do not eat much during journey.					
26. I do not always fly during day time.					
27. I do not always prefer airlines with excellent complaint handling services.					
Qualitative Perception of Flexibility			0	+	++
28. For me a system is flexible if I can do changes in my itinerary.			1		1
29. For me a system is flexible if I can complete my work.					
30. For me a system is flexible if I can recover from mistakes quickly and easily.				- <u>-</u>	
31. For me a system is flexible if it offers all features lintend to use.				' ı	
Satisfaction			0	+	 ++

32. Overall, I am not satisfied with how easy it is to use this website.

- 33. It was not simple to use this website.
- 34. I cannot effectively complete my work using this website.
- 35. I amnot able to complete my work quickly using this website.
- 36. I amnot able to efficiently complete my work using this website.
- 37. I do not feel comfortable using this website.
- 38. It was not easy to learn to use this website.
- 39. I believe I became unproductive quickly using this website.
- The website does not give error messages that clearly tell me how to fix problems.
- 41. Whenever, I make a mistake using the website, I cannot recover easily and quickly.
- 42. The information (such as online help, on-page messages, and other documentation) provided with this website is not clear.
- 43. It is not easy to find the information I need.
- 44. The information provided by the website is not easy to understand.

).		
		'		
			·	

an Ghegar sann eo achdalain gart seis gear an san san she an an san san san san sa

- 45. The information is not effective in helping me complete the tasks and scenarios.
- 46. The organization of information on the website pages is not clear.
- 47. The interface of this website is not pleasant.
- 48. I do not like using the interface of this website.
- 49. This website does not have all the functions and capabilities I expect it to have.
- 50. Overall, I am not satisfied with this website.

Cheap Fare vs. Your Preferences

- 51. If you are offered a cheap flight fare, up to what extent you may compromise on the following?
 - a. I may fly from neighbor town rather flying from my own town.
 - b. I may compromise on seat comforts and leg space area.
 - c. I may choose a flight without jet ways.
 - d. I may compromise on staff attitude.
 - e. I may fly during weekdays.
 - f. Imay compromise on ancestry relationships.
 - g. I may compromise on baggage weight allowed.
 - h. I may compromise on food and services.
 - i. I may fly during night time.
 - j. I may compromise on complaint handling services.
 - $\mathbf{k} = I \max$ compromise on airline's repute.

© Thank you very much for completing this questionnaire ©

	ł		[ţ,	
				<u> </u>	
				i	
	i				
				l I	ł
	Ì				
i	<u> </u>				
	1				
1					
				i '	Ľ
	F .				ł
		-	0	[+]	++
i					l
1	ŀ). ·]	l
				<u> </u>	
	<u> </u>				
1	i '				
				<u> </u>	
1					
			í l		
	L				
1	}				
	<u> </u>			<u> </u>	<u> </u>
			-		
	<u> </u>				
			}) '	
	┝┈ <u>──</u> ─		~		<u>├</u> ───
	[,			(I	
		L			

7 as e 3

анаа соораанын жааны ооланын тоолооны улаанын тоораанын оодоо тоогын тоорааны оодаан оораан араан араан араан а

APPENDIX C

USERS' FLEXIBILITY TOWARDS FLEXIBLE ONLINE AIRLINE

RESERVATION SYSTEMS

Users' Flexibility towards Flexible Online Airline Reservation Systems (FOARS)

Objective – The objective of this questionnaire is to find out the degree of flexibility of airline travelers. The feedback received may help in designing more flexible reservation systems.

	General Information (Please place a tick (/) at the appropriate box)										
1.	Age	1Below 20 years441 to 50 years	2 21 to 30 years 5 51 to 60 years	3 31 to 40 years6 Above 60 years							
2.	Gender	1 Male	2 Female								
3.	Educational Background	1 High School 4 Post-graduate	2 Diploma	3 Graduate							
4.	Occupation	1 Student 4 Retired	2 Professional	3 Self Employed							
5.	Ethnicity	1 EU 4 Westem	2 Asian 5 Far East	3 African 6 Others							

Rate your priorities for the following service quality attributes of an airline in terms of their importance.	Highest	High	Neutral	Low	Least
	1	1	3	4	5
6 Fiving date confirmation					
7. Flying carrier confirmation					
Flying time confirmation		,			
9. Number of stop-over					
10. Number of connected flights					
11. Ticket class (economy business)					[
12. Seat specifications					
13. Discounted airfares					
14. Origin Destination airports					
15. Immediate confirmation of itinerary on purchase of ticket					L

Please indicate which of the following statements describes you in the best possible way?

	Strongly Disagree	Disagree	Neutral	Agrae	Strong	y Agre	æ		
		2 3		4		-			
						1	2	3	5
16.	I rarely prefer my own cou	intrv's airline.						-	-
17.	I rarely prefer airlines that and wait for busses	provide jet ways,	so that I don't h	ave to climb the g	angway				
18,	I rarely prefer to fly during	g occasions.							Į
19.	Peak time travel does not :	fascinate always.							
20.	Off time travel does not fa	iscinate always.							
21.	I would prefer to use syste environment.	ms that contribute	e their efforts tow	ards carbon free					

23ga 1

Quarter basy Phot	2	±r€.	eís:	÷.
-------------------	---	------	------	----

202

- 22. Staff attitude never motivates me to use the same airline
- 23. Trarely prefer airlines with best food services; I do not eat much during journey
- 24. Trarely shop, eat or use internet at the airport.
- 25. Cheap tickets are at highest priority in my itinerary planning.
- 16. I prefer online systems because I can find a better package.
- 27. I rarely get persuaded by a higher baggage weight being allowed.
- 28. I rarely prefer airlines with excellent complaint handling services.
- 29. Iprefer manual check-in over online check-in.
- 30. I amnot a frequent flyer; I fly only once in a year or so.
- 31. I normally fly to spend holidays with family friends.
- 32. I prefer to fly during my children holidays.
- 33. Inormally fly with my family or friends.
- 34. I normally fly to expanding business or to meeting professional objectives
- 35. How do you rate your overall personality in terms of flexibility?
- 36. I am a regular news reader; I always keep myself updated on tourism.
- 37. I am excellent in managing trips; I can plan my trip on a very short notice.
- 38. I am excellent in computer skills; I can book my itinerary in a very short time.
- 39. Online systems are not very complicated for me to use.
- 40. Online systems provide support to help recover from mistakes.

Perceived Usefulness

- 1. This website has all the functions and capabilities I expect it to have.
- 2. I will be more productive using this website to book my itinerary.
- 3. I can effectively make reservations using this website.
- 4. I am able to book my itinerary more quickly using this website.
- 5. It is easy to make reservations using this website.
- 6. I am able to efficiently complete my work using this website.
- 7. It is easy to learn to use this website.
- 8. I feel comfortable using this website.
- 9. It is simple to use this website.
- 10. The website is flexible to interact with.
- 11. The information is effective in helping me complete the tasks and scenarios.
- 12. I find this website easy to use.
- 13. Overall, I am satisfied with the functionality it offers.
- 14. Overall, I am satisfied with this website.

\otimes Thank you very much for completing this questionnaire \otimes

, the constraint of the test of the second sec	er e	 gen e en server e e generer		i i an a' statu s	22 . C	· · ·		
jaarin din 6 Felidi Sakis	. • :) nga	2

	_	

1	3	3	4	5

APPENDIX D

۹

TRANSFORMATION OF EVALUATORS USERS' FLEXIBILITY SCORING

Evaluators	Date Confirmation	Carrier Confirmation	Time Confirmation	No. of Stop- over	No. of Connected Flights	Ticket Class	Seat Specification	Discounted Airfares	Immediate Confirmation of Itinerary	Destination Source Airport	Users' Flexibility Score
User 1	2	4	+	4	4	4	1	4	4	3	7
User 3	4	4	4	1	1	3	4	5	1	2	8
User 3	4	1	4	3	3	4	3	5	3	+	
User 4	1	4	5	4	4	4	\$	3	3	4	11
User 5	5	5	5	4	4	4	4	3	3	1	S
User 6	5	4	5	4	4	3	5	3	-1	1	8
User 7	4	5	4		4	5	4	2	3	4	10
User S	4	1		<u> </u>	4	+				4	11
User 9	3		5			<u> </u>	3	4		3	19
User 10	5	4	3		2	2		4	4	1	11
User 11	4	4	+		4	-	3	<u>د</u>		د .	<u> </u>
User 1.		4	3		3	+	4	*	2		11
User 13	3		+				+			*	14
User 14	2	4				2		· · · · · · · · · · · · · · · · · · ·			12
User 16	- 	- 4		÷	-	4	4		5		17
User 17	Ś	- L		Ĩ		1		- <u> </u>			- 1 - 0
Elser 18		-				1	3		· · · · · · · · · · · · · · · · · · ·	2	8
User 10		1	5	1	4	1 1		3	3	1	9
User 20			5	5	1	5	5	1	\$	1	18
User 21	3	4		4				2	4	3	10
User 22	4	5	4		4	1 3	4	3	3	2	
User 23		·	5	5		3	5	3	5		18
User 24	3	3	4	4	4	i i	4	1	4	4	12
User 25	5	4	5	3	4	3	3	3	3	3	6
User 16	5	5	5	5	5	3	5	1	5	3	19
User 27	2	2	2	2	4	4	4	4	4	4	2
User 28	-	4	2	2	4	4		4	4	4	- 1
User 29	5	5	5	5	5	5	5	1 1	5	4	18
User 30	5	5	5	5	5	5	5	4	1	4	17
User 31	5	4	5	5	5	5	4	4	4	4	15
User 32	5	4	5	5	5	5	4	4	4	4	15
User 33	5	4	4	4	3	4	5	3	3	3	
						.			,		
User 34	5	4	5	4	4	4	4	3	3	3	9
User 35	5	5	5	5	4	4	4	4	4	4	14
User 36	5	\$	4	5	5	4	4	4	4	2	12
User 37		3	5	4	4		<u> </u>	3	3	1	8
User 38		4	4	3	4	ز ا	3	3	3		3
User 39	4	2	4	-+	4	4	1	4	ذ	,	9
User 40			+	+	+	4		2	3		
User 41		4	3		-	+				. 3	
User 42		L L		5	5	5	1	4	4	3	13
User 11	<u>;</u>	<u> </u>	4	1	3	1		4	3	- <u>ī</u>	10
User 45	5	1	5		4	4	4	4	2	2	9
User 46	5	5	4	5	5	4	4	4	4	2	12
User 47	5	5	4	4	4	4	1	4	3	3	10
User 48	5	5	4	ż	5	4	4	3	2	1	8
User 49	Š	5	5	5	4	-4	3	3	3	2	9
User 50	5	5	4	3	4	4	4	1	3	2	10
User 51	5	5	4	5	5	5	4	4	2	2	9
User 52	4	3	+	4	4	4	3	3	3	3	-
User 53	3	1	4	4	4	4	4	3	4	2	9
User 54	5	3	5	5	5	4	3	3	3	2	S
User 55	5	4	4	5	4	4	4	3	3	3	9
User 56	3	4	+	5	+	4	4	3	3	3	9
User 57	5	4	5	4	5	1 1	3	3	3	3	8
User 38	5	4	5	4	4	15	3	<u> </u>	3	11	<u> </u>
User 59	4	1	4	3	↓ ↓	1.3	4	5	3	3	<u> </u>
User 60	5	3	3	4	+	4	4	4	2	2	9
User öl	3	4		4	<u> </u>	3	1	3	4	1	<u> </u>
i User 62	15	4) 5	4	1 3	1 4	1 4	1 3	1 3	1 2	1 7

Transformation of Evaluators Users' Flexibility Scoring

APPENDIX E

CLASSIFICATION OF TRAVELERS' ON THE BASIS OF THEIR FLEXIBLE TRAVELING BEHAVIOR

Classification of Travelers' on the basis of their Flexible Traveling Behavior

Evaluators	Date Confirmation	Canier Confirmation	Time Confirmation	No. of Stop- over	No of Connected Flights	Ticket Class	Seat Specification	Discounted Airfares	Immediate Confirmation of Itmerary	Destination Source Airport	Users' Flexibility Score	Classify
User 1	2	4	4	4	4	4	4	4	1	3	-	1
User 2	4	4	1	1	4	3	4	2	4	5	S	2
User 3	+	4	4	3	3	4	3	Ś	3	1		!
User 4		+	5	4	4	1	3	3	3	4	11	
User 5		5	5	4	4	1	4	3	3	1	8	2
User 6	3	4	5	4	4	3	5	3	1	1	S	
User	1		4	5		5	, i		3	<u> </u>	10	L
User S	4		4	<u>}</u>	<u> </u>	4	,		· · · · · · · · · · · · · · · · · · ·		11	- ÷
User 9	2	2	2	>	2	2	2	1	3	2	19	
User 10		4					-4	*	*	1		<u></u>
User 11	÷	+						······	······			
User 13	-								3	-	11	
User 14	- -		5					· · · · ·	4		15	3
User 15	ŝ	ŝ	5	5	5		<u> </u>		3		12	2
User 16	\$	5		-		5			-		17	3
User 1	3	4	4	4	4	4	3	4	3	3	9	<u> </u>
User 13	5	4	3	4	4	4	3	}	2	3	\$	
User 19	5	4	3	4	4	4	3	3	3	4	9	1 2
User 20	3	-	5	5	5	5	5	1	3	4	18	3
User 21	:	4	ź	L I	5	÷	4	2	4	3	10	2
User 11	+	5	4	Š	4	3	4	3	3	3	<u> </u>	1
User 23	< .	5	5	5	3	5	3	3	\$	5	18	3
User 24	5	5	4	+	4	4	4	4	4	4	12	2
User 25	3	4	5	3	4	3	3	3	3	3	6	3
User 16	5	5	5	5	5	5	5	4	3	5	19	3
User 2	2	2	2	2	4	÷	4		1	4	2	1
User 28	2	4	2	2	4	4	4	4	-	1	4	1
User 29	5	3	3	5	5	Ś	5	4	5	1	1\$	3
User 30	5	5	1	5	5	3	5	4	4	1	17	3
User 31	5	4	3	5	5	5	4	4	4	4	15	3
User 32	5	4	5	5	5	5	4	4	1	4	15	3
User 33	2	4	4	4	3	4	5	3	3	3	7	1
TT 7.1												
User 34			2	+			- +		,	,	<u>ÿ</u>	2
They 36		2						*	+		14	<u> </u>
User 30			4	2	د	+					14	
User 38								3	3		4	1
User 39	<u>i</u>	4	1				4	1	;		9	÷
User 40		· · · · · · · · · · · · · · · · · · ·	4	1		1	4	-	3			
User 41		1	1	1			1	3	1		Ă	<u>-</u>
User 12		5	3	4	4	4	4	1	3		- ŭ	
User 43	3	4	5	5	5	3	4	4	4	3	13	- 2
User 44	÷	3	1	3	5	4	4	4	3	1	10	2
User 45	4	4		5	4	4	t.	4			9	- 1
User 40	3	<	4	5	5	4	4		4		12	2
User 4	5	5	4	4	4	4	4	4	3	3	10	2
User 48	5	5	4	<u> </u>	5	4	4	3	2	1	8	5
User 19	*	5	1	5	4	÷	3	3	ž	2	9	2
User 50	\$	5	4	5	4	4	1	4	3	2	10	2
User 51	5	5	4	5	- 5	5	1	4	2		9	2
User 52	4	Ś	4	+	4	4	3	3	3	3	-	1
User 23						4	4		1		- 9	
User 24		2			3	4	ś		j		8	
User 22	2						4		<u> </u>	<u> </u>	<u> </u>	
User 20 User 30								<u>;</u>			9	
User :		+	2	+	2	+	د .		5		<u>8</u>	
Tiser 30		*	2	4	4	3	5		3		3 9	
User 60						4			,	·	a Q	
User 61						•		2	-			l i
User 6			<	-		<u>د</u>		2	•	1		
0.00		4	3	-	3	-4	+	2	3	-		L

Classification: 1 – Low Flexible 2 – Medium Flexible

³ - High Flexible

APPENDIX F

PERMISSION TO COLLECT DATA FROM TRAVEL AGENTS FOR RESEARCH

PURPOSES



s^h March 20*1

To Whom It May Concern

Ref: Permission to Collect Data from Travel Agents for Research Purposes

The arrive matter is solutional.

Mr.Ant Mushtab is one of my Politistudence of Universit Tokinglogi PETRONAS whose research maximum of the fight reservation systems. He is consolity of the data gathering stage where ris needs to memory and conduct a survey to have approximation potential customers and conduct a survey to have approximation of strength and potential.

With this report. Each service for your which approval and experience to each. Min Ant to concrete this study on your company. Each averagease ance that all and collected will be set is confidential and strictly for research purpose unity. Asso, the coupy for to conducted will the activit be an ordinated of the event.

- would like to the skype. Tenyour keed concension and support in this matter. Placese car bot heselate to contain the figure need any further information or car figuresh regarding this reduced.

Brank your

You's any even of the many
De Subgradification
Sector of Researcher
Description of Computer of the many is assessed to a sector of the many labor of the many is assessed to a sector of the many labor of the many is a sector of the many labor of t

1-66.05-3687415 Hax, 25-3856386 Ethail Staasl@greatories.com my

الایک ۲۰۵۵ (۲۰۵۵ (۲۹۷۵) ۲۰۱۰ (۲۹۹۵) ۲۰۱۰ (۲۹۹۹) ۲۰۱۰ ۲۰۱۰ (۲۹۹۹) ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ ۲۰۱۰ ۲۰۱۰ ۲۰۰۰ - ۲۰۰۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹۹۰ - ۲۹

APPENDIX G

PERMISSION TO COLLECT DATA FROM TRAVEL AGENTS AT MATTA

FAIR 2011



Ms Hong Obs alice: Manager MATTA Har 2011 Killa a Lombur

311 Balbuary 2001

Sta Maria

Ref. Permission to Collect Data from Travel Agents at MATTA Fair 2011 for Research Purposes

This above matter is referred.

 $M_{\rm e}$ (of Mushtaq is one of may 97D students of university (s-notog, 92TRONAS whose responsible nucleus during high inservation systems. He is currently in the bata gathering states where the therein to interview and conduct a survey to movel agents and contactly current to interview and conduct a survey to movel agents and contactly currents movel agents and contactly survey to movel agents and contactly currents.

 $\mathcal{M}(t)$ this regard i am seeking for your kind accross and oblige above to allow Mr Anfro connections study during the forthooming MeV [A Fair 2011 in Klubb , or put i amig ving destance that as data collected with be kept confidential and strictly for research parameter may blact the study to be the confidential without discretion the sought control the point.

I would like to them you for your citol cooperation and would on including the set of or antication regardless to contact the if you need any further information or clandication regardless.

inan-ye.

Suis Spokely
 Structs
 Busis Spokely
 Busis States Researcher
 Bepartment of Computer and Internation Sciences
 Struct Technology PETRONAS
 Struct Technology PETRONAS
 Struct Technology PETRONAS

1-61-05-0007415 Filo - 25-3036189 Botwel Blucks Sylpestonias Commy

(A. 10.4) In the probability of the probability

APPENDIX H

PERMISSION TO COLLECT DATA FROM TRAVEL AGENTS AT MATTA

FAIR 2011

Usability Evaluation Checklist

	Guidelines	Rating	Comments
Chapter 2	: Optimizing the User Experience		
2.1	Do Not Display Unsolicited Windows or Graphics		
2.2	Increase Web Site Credibility		
23	Standardize Task Sequences		
2.4	Reduce the User's Workload	1	
2.5	Display Information in a Directly Usable Format		
3.6	Do Not Require Users to Multitask While Reading		
	Use Users' Terminology in Help Documentation	<u> </u>	
7.8	Provide Assistance to Uterr	<u> </u>	
Depres 5. The Homeoage			
51	Create a Positive First Impression of Your Site		· · · · · · · · · · · · · · · · · · ·
	Communicate the Web Site & Value and Purnose		
* 3	I mit Prote Text on the Homenage		
5.1	Limit Homena as Learth		· · · · ·
Chamten	Ender Formepage Length		I
Casper	Auril Channel Dimini	r	
6.1	Avoid Cluttered Displays		
0.2	Place Important items Consistently		
ن_0	Place Important Items at Top Center		
6.4	Structure for Easy Companison		
6.5	Use Moderate White Space		
6.6	Choose Appropriate Line Lengths		
Chapter 8: Scrolling and Paging			
\$ 1	Eliminate Horizontal Scrolling	L	
8.2	Facilitate Rapid Scrolling While Reading	<u> </u>	
Chapter !	: Headings, Titles, and Labels		
9.1	Use Clear Category Labels		
9.2	Use Unique and Descriptive Headings		
9.3	Highlight Critical Data		
94	Provide Users with Good Ways to Reduce Options		
Chapter	I: Text Appearance	•	
11.1 Use Black Text on Plain, High Contrast Backgrounds			
11.2	Format Common Items Consistently		
11.3	Ensure Visual Consistency		
11.4	Use Bold Text Spannely	1	· · · · · · · · · · · · · · · · · · ·
11.5	Use Attention-Attracting Features when Appropriate	1	
11.6	Use Familiar Fonts		
11.7	Use at Least 12-Point Font	t	
11.8	Color-Coding and Instructions	+	
11.9	Emphasize Importance		
Chapter	15: Writing Web Content	·•	
15.1	Make Action Sequences Clear	1	
15.2	Use Familiar Words		
15.3	Define Acronyms and Abbreviations		
15.4	Use Mixed Case with Prose		
15.5	Limit the Number of Words and Sentences		
156	Use Active Voice		
Chapter	16: Content Organization		· · · · · · · · · · · · · · · · · · ·
16 1	Organize Information Clearly		
16.2	Ensure that Necessary Information is Displayed		
16.3	Group Related Elements		
16.4	Minimize the Number of Clicks or Pages		
16.5	Design Quantitative Content for Quick Understanding		
16.6	Display Only Necessary Information	T	
16 7	Format Information for Multiple Audiences		
16.8	Use Color for Grouping		

Note: Assign each guideline a problem severity rating as follows

- Not Applicable
 I don't agree that this is a usability problem at all
 Connetic problem only: need not be fixed unless extra time is a vallable on project
 Muror usability, problem: fixing this should be given low prionty
 Major usability, problem: important to fix, so about de given high prionty
 Usability catastrophe: imperative to fix this before product can be released

N/A = Not Applicable 0 = I don't agree that 1 = Cosmetic problem 2 = Minor usability p 3 = Major usability p 4 = Usability catastro